Designing Attentive Information Dashboards with Eye Tracking Technology

Zur Erlangung des akademischen Grades eines Doktors der Wirtschaftswissenschaften

(Dr. rer. pol.)

von der KIT-Fakultät für Wirtschaftswissenschaften des Karlsruher Instituts für Technologie (KIT)

genehmigte

DISSERTATION

von

Peyman Toreini, M.Sc.

Tag der mündlichen Prüfung:	02.09.2020
Referent:	Prof. Dr. Alexander Mädche
Korreferent:	Prof. Dr. Andreas Oberweis

Karlsruhe

2020

Acknowledgment

First and foremost, I want to express my sincere gratitude to Prof. Dr. Alexander Mädche for the continuous support of my PhD study and for helping me to overcome the challenges I faced through this journey. He inspired my interest in developing innovative systems and gave me the freedom to do whatever I wanted while contributing valuable feedback, helpful advice, practical suggestions, and encouragement. In addition to my PhD study, I greatly value what I learned under his supervision in projects, education, and organizational tasks. I want to thank him for assisting my career and allowing me to grow in diverse dimensions. I was fortunate to have such a great mentor through this phase of my life. I would also like to thank Prof. Dr. Andreas Oberweis, Prof. Dr. Petra Nieken, and Prof. Dr. Rudi Studer for agreeing to serve on my PhD committee. I truly appreciate all of their time to review my thesis and insightful comments and discussions before and during the dissertation defense.

Thanks should also go to all the people whose assistance was precious in completing my dissertation. I would particularly like to acknowledge my colleagues at the Institute of Information Systems (IISM) at Karlsruhe Institute of Technology (KIT) for their wonderful collaboration during the last years. I thank them for the stimulating discussions in research meetings, for the sleepless nights we were working together before deadlines, and for all the pleasant moments we have had in the last years. It is their companionship that has made my study and life in Karlsruhe an amazing time. Besides, I would like to thank my co-authors, whose cooperation was a turning point in shifting my ideas into reality. I would particularly like to single out Moritz Langner for his great assistance during my PhD and for helping me in several studies.

I would also like to thank my friends, who make my life delightful. I am beyond lucky to have such friends that bring their support in all conditions. I could not have completed this dissertation without them providing me happy distractions in our gatherings in Karlsruhe or other cities. Also, thanks for their concern and suggestions on how to move forward in my life and career. In particular, I would like to thank Soheil Aghaei and Dr. Behdad Ariatabar for providing me with the tools that I needed to choose the right direction in different phases of my PhD life.

Above all, my sincerest gratitude goes to my family: my parents, sister, and brother. I dedicate this dissertation to two heroes and the biggest influences on my life who have always been my motivator; my father, Moslem Tourini, and my mother, Roya Behboodi. Thanks for profound belief in my abilities and for giving me your endless and unconditional

love even from thousands of kilometers away. You proved that borders and distances could not stop spreading love and support, and this work would not have been possible without that.

Finally, very special thanks to my partner, Annet Kartouzian, for walking shoulder to shoulder with me throughout every step of my dissertation. I could not finish this journey without your love, guidance, patience, and care. Thank you for always being there whenever I needed a close friend and pushing me towards my goals. I am immeasurably grateful for you in my life.

Peyman Toreini Karlsruhe, Germany October 2020

Abstract

With the use of Business Intelligence and Analytics (BI&A) systems, companies collect and analyze data from various sources to support data-driven decision making. In order to access BI&A systems, decision-makers leverage information dashboards, visual displays that arrange the most important information on a single screen. Despite the fact that information dashboards come with strong potentials to support better decision making, but they challenge their users' limited attentional resources by presenting huge amounts of information on one screen. Providing solutions addressing users' difficulties in managing limited attentional resources while working with information dashboards is an important research gap. Existing research in the Human-Computer Interaction (HCI) field highlights the importance of designing attentive user interfaces to assist users' attentional processes using eye tracking technology. This thesis builds upon the fields of HCI, Information Systems (IS) and psychology and follows the Design Science Research (DSR) methodology in order to answer the following research question: "How to design attentive information dashboards for BI&A systems that enhance users' ability to manage attentional resources?".

Overall, five studies are performed in this thesis. In Study I, a systematic literature review on previous research focusing on eve-based interactive intelligent systems is conducted. As part of this study, a conceptual framework is developed, and future research directions are identified. Building on these results, three design cycles for designing attentive information dashboards are performed. The first design cycle includes two studies: Study II investigates attention problems of information dashboard users under consideration of their individual Working Memory Capacity (WMC). The results show that users with high and low WMC have different difficulties in managing their attentional resources. Study III evaluates different potential solutions for supporting data exploration tasks using eye movement data. The results show that providing feedback integrating real-time eye movement data supports users in managing their attentional resources better than general feedback with integrated off-line recordings of eye movement data. In the second design cycle, an attentive information dashboard for the data exploration task is designed and evaluated. In Study IV, theoretically grounded design principles are articulated, instantiated as a software artifact, and evaluated in a large-scale laboratory experiment. The results from analyzing the users' eye movement data reveal that the suggested design principles positively affect users' ability to manage limited attentional resources during data exploration tasks. In the third design cycle, attentive information dashboards with task resumption support are investigated. Study V instantiates three software artifacts using different gaze-based highlighting methods and evaluates them in a large-scale laboratory

experiment by considering short-term IT-mediated interruptions and the role WMC. The results demonstrate the need for personalization of such support under consideration of users' WMC.

This thesis contributes to the IS and HCI field by providing prescriptive knowledge in the form of nascent design theory for designing attentive information dashboards using eye tracking technology to enhance users' ability to manage limited attentional resources. The proposed attentive information dashboards are the first BI&A systems presented in the IS field that use real-time eye movement data for designing advanced built-in attention support functions. Also, practitioners can leverage the findings from this thesis for integrating attention support functionality into existing BI&A system supporting users in managing limited attentional resources.

Contents

Lis	st of	Figures		xiii
Lis	st of	Tables		xiv
Lis	st of	Abbrev	iations	xiv
1.	Intro	oductio	n	1
	1.1.	Motiva	ation	. 1
	1.2.	Resear	ch Gaps and Associated Research Questions	. 7
	1.3.	Thesis	Structure	. 12
2.	Con	ceptual	Foundations	15
	2.1.	Overv	iew	. 15
	2.2.	Theore	etical Foundations	. 16
		2.2.1.	Attention	. 16
		2.2.2.	Working Memory	. 18
		2.2.3.	Human Information Processing	. 20
	2.3.	Relate	d Work	. 22
		2.3.1.	Eye Tracking Technology	. 22
		2.3.2.	Attentive User Interfaces	. 25
		2.3.3.	Business Intelligence and Analytics Systems	. 27
		2.3.4.	Information Dashboards	. 28
3.	Stat	e-of-th	e-art and Conceptual Framework	30
	3.1.	Study	I: Overview	. 30
	3.2.	Conce	ptual Foundations	. 31
		3.2.1.	Eye-based Interactive Intelligent Systems	. 31
		3.2.2.	Integrated Conceptual Framework	. 32
	3.3.	Metho	dology	. 35
		3.3.1.	Planing the Review	. 35
		3.3.2.	Conducting the Review	. 37
	3.4.	Findin	gs	. 38
		3.4.1.	Influencing Factors	. 38
		3.4.2.	Eye-based IIS Properties	. 44
		3.4.3.	Outcome	. 48

	3.5.	Discus	sion and Future Research Directions	51
		3.5.1.	Influential Factors	53
		3.5.2.	Eye-based IIS Properties	55
		3.5.3.	Outcomes	56
	3.6.	Summa	ary	57
4.	Rese	earch M	lethodology	59
	4.1.	Design	Science Research	59
		4.1.1.	Design Cycle 1	60
		4.1.2.	Design Cycle 2	61
		4.1.3.	Design Cycle 3	61
Б	Doc	ian Cyc	lo 1. Attention Management Problems and Possible Solutions	63
5.	5 1	Study	II: Overview	63
	0.1.	5.1.1	Laboratory Experiment	65
		0.1.1.	5 1 1 1 Experiment Design	65
			5.1.1.2. Experiment Participants	67
			5.1.1.3. Measurements	68
		5.1.2.	Data Analysis and Results	69
			5.1.2.1. First Visit Phase	69
			5.1.2.2. Revisit Phase	71
			5.1.2.3. End of the Task	72
		5.1.3.	Derived Meta-requirements	75
		5.1.4.	Summary	77
	5.2.	Study	III: Overview	79
		5.2.1.	Meta-requirements	80
		5.2.2.	Instantiation of VAF Types	81
		5.2.3.	Eye Tracking Pilot Study	84
			5.2.3.1. Experiment Design	84
			5.2.3.2. Experiment Participants	85
			5.2.3.3. Measurements	85
		5.2.4.	Data Analysis and Results	86
		5.2.5.	Discussion	87
		5.2.6.	Summary	88
6.	Desi	ign Cyc	le 2: Attentive Information Dashboards for Data Exploration	89
	6.1.	Study	IV: Overview	89
	6.2.	Meta-r	equirements and Design Principles	91
	6.3.	Develo	pment	93
	6.4.	Hypot	heses	95
	6.5.	Labora	atory Experiment	97
		6.5.1.	Experimental Software and the Apparatus	97
		6.5.2.	Experimental Design	99
		6.5.3.	Participants	100

		6.5.4. Measurements	01
	6.6.	Data Analysis and Results	03
		6.6.1. Manipulation and Control Checks	03
		6.6.2. Attentional Resource Allocation	03
		6.6.3. Attention Shift Rate	06
		6.6.4. Attentional Resource Management	08
	6.7.	Discussion \ldots \ldots \ldots \ldots \ldots \ldots 10	09
	6.8.	Summary	10
7.	Desi	ign Cycle 3: Attentive Information Dashboards with Task Resumption	
	Sup	port 1	12
	7.1.	Study V: Overview	12
	7.2.	Background and Conceptualization	14
	7.3.	Meta-requirements and Design Principles	18
	7.4.	Development	21
	7.5.	Laboratory Experiment	24
		7.5.1. Experimental Software and the Apparatus	24
		7.5.2. Experimental Design	26
		7.5.3. Participants	27
		7.5.4. Measurements	28
	7.6.	Data Analysis and Results	31
		7.6.1. Task Resumption Performance	32
		7.6.2. Task Performance	34
		7.6.3. Gaze-based TRS Relevance	35
	7.7.	Discussion	37
	7.8.	Summary	39
Q	Dice	ursion 1	۸۸
0.	8 1	Theoretical Contributions	40
	0.1. 8 9	Practical Contributions	±0 ΛΛ
	0.2. 8 3	Naccont Design Theory	14 16
	0. J .	DSP Knowledge Contribution	±0
	8.5	Limitations and Future Research	57
	0.0.		94
9.	Con	clusion 1	60
٨	nond	liv 1	62
~	λ	Study I 1	62 62
	л.	A 1 List of Papers for Literature Review	02 69
		A 2 Coding Tables for Literature Review	62
	р	Study II	co co
	ы.	P1 Experimental Software	00 60
		D.1. Experimental Software	Uð GO
		D.2. Further Analysis	09

eferen	ices		186
	ш.э.		104
	E 3	Further Analysis	18/
	E.2.	Survey Items	183
	E.1.	Experimental Software	178
Е.	Study	V	178
	D.3.	Further Analysis	175
	D.2.	Survey Items	175
	D.1.	Experimental Software	173
D.	Study	IV	173
	C.2.	Further Analysis	171
	C.1.	Experimental Software	170
С.	Study	III	170

References

X

List of Figures

1.1.	Value chain of BI&A systems and the role of computers and humans	2
1.2.	Major components of attentive information dashboards (Figure is adapted	
	from Riedl and Léger (2016)). \ldots \ldots \ldots \ldots \ldots \ldots \ldots	5
1.3.	Overview of research questions addressed in this thesis.	8
1.4.	Structure of the thesis	14
2.1.	Overview of research streams, theoretical foundations and research gaps	15
2.2.	An example of goal-directed and stimuli-driven attention. \ldots \ldots \ldots	17
2.3.	An example of overt and covert attention	18
2.4.	Multi-store model proposed by Atkinson and Shiffrin (1968). \ldots	18
2.5.	Updated version of Baddeley's model of working memory by Baddeley (2000).	19
2.6.	A model of HIP stages adapted from Wickens et al. (2016)	22
2.7.	Tobii 4C eye tracker with the sampling rate of 90 Hz	24
2.8.	Categories of eye tracking applications	25
3.1.	Integrated conceptual framework for eye-based IIS	32
3.2.	Steps of the SLR study following Kitchenham and Charters (2007)	36
3.3.	Search string with four main categories to conduct the SLR study	37
3.4.	Distribution of collected papers based on years and research fields	38
3.5.	Descriptive statistics: user context	39
3.6.	Descriptive statistics: physical context	40
3.7.	Descriptive statistics: tasks	41
3.8.	Descriptive statistics: eye tracking devices.	41
3.9.	Descriptive statistics: eye tracking hardware providers	42
3.10.	Descriptive statistics: eye tracking technology prices	43
3.11.	Descriptive statistics: experimental design.	44
3.12.	Descriptive statistics: number of participants	44
3.13.	Descriptive statistics: eye tracking measures	47
3.14.	Descriptive statistics: modeling user states	47
3.15.	Descriptive statistics: focus of eye-based IIS	48
3.16.	Descriptive statistics: user perceived outcomes	49
3.17.	Descriptive statistics: behavior outcomes	50
3.18.	Descriptive statistics: user performance outcomes	51
3.19.	Descriptive statistics: system performance outcomes	51

4.1.	Design cycles integrated in this thesis.	59
5.1.	The focus of Study II in this DSR project.	64
5.2.	Stages of the designed exploratory experiment.	65
5.3.	The six pre-defined AOIs based on their position.	66
5.4.	Measures used in the exploratory study.	69
5.5.	Distribution of dwell-times on six AOIs after the first phase of the experiment.	70
5.6.	The ARM of participants after the first phase of the task	70
5.7.	Total dwell-times on previously low and high visited AOI in the revisit phase	
	for users with high WMC (left). Total dwell-times on previously low and	
	high visited AOIs in the revisit phase for users with low WMC (right)	71
5.8.	Comparing last selected AOI in the first phase with first selected AOI in	
	revisit phase of the data exploration task.	72
5.9.	Distribution of dwell-times on six AOIs at the end of the task.	73
5.10.	The ARM of participants at the end of the task.	75
5.11.	The focus of Study III in this DSR project.	80
5.12.	The information dashboard layout used in two rounds of the experiment.	82
5.13.	Three VAF types that are suggested and evaluated in Study III	83
5.14.	Stages of the designed eye tracking pilot study	84
5.15.	Measures used to compare three suggested VAF types	85
5.16.	ARA during the revisit phases of experimental rounds	86
5.17.	ASR during the revisit phases of experimental rounds	87
6.1.	The focus of Study IV in this DSR project.	91
6.2.	System architecture for designing attentive information dashboards that	
	support data exploration tasks.	94
6.3.	Instantiation of DP2	95
6.4.	Research model to investigate the effect of DP1&2	97
6.5.	The designed information dashboard to control for stimulus-driven attention.	99
6.6.	The experiment's procedure used to evaluate the effects of DPs	100
6.7.	Heatmaps of both groups in the first and revisit phases	104
6.8.	The interaction between and after VAF types for both groups	105
6.9.	Transition matrix of the users in both groups	106
6.10.	ASR of users before and after receiving VAF types	107
6.11.	Interaction effect of groups and phases on ARM performance	108
7.1.	The focus of Study V in this DSR project	114
7.2.	Conceptualization of gaze-based TRS.	116
7.3.	Example scenario to represent steps for providing TRS	118
7.4.	System architecture for designing attentive information dashboards that	
	support resuming interrupted tasks.	122
7.5.	Three suggestions for DP3 as gaze-based TRS	123
7.6.	Example of the information dashboard used in Study V	124
7.7.	Stages of interruption and resumption adapted from Trafton et al. (2003) .	125

7.8. The main steps of the experiment.	. 126
7.9. ARA of users in the first visit phase	. 131
7.10. RSR of users in the revisit phase	. 132
7.11. ARA of users on previously high-visited AOIs.	. 134
7.12. Users' opinions about Gaze-based TRS	. 136
8.1. DSR knowledge contribution based on Gregor and Hevner (2013). \ldots	. 154
B.1. The information dashboard layout used in Study II	. 168
B.2. The break page as an interruption used in Study II	. 168
B.3. The heatmaps of users with high and low WMC in Study II	. 169
C.1. The information dashboard layout used in the first round of Study III	. 170
C.2. The information dashboard layout used in the second round in Study III.	. 170
C.3. Example of VAF design used in the second round in Study III	. 171
C.4. The heatmaps of users in first round of the experiment in Study III. $\ . \ .$. 171
C.5. The heatmaps of users in second round of experiment in Study III	. 172
D.1. Simplified information dashboard used as the learning phase in Study IV.	. 173
D.2. The information dashboard layout used in Study IV	. 173
D.3. An example of individualized VAF used in Study IV	. 174
D.4. The general VAF as a text-based explanation used in Study IV	. 174
D.5. Distribution of fixation duration values before and after VAF types	. 176
D.6. Distribution of number of fixations before and after VAF types	. 176
D.7. ARM of the users in the first and end of the data exploration tasks based	
on fixation duration values.	. 177
D.8. ARM of the users in the first and end of the data exploration tasks based	
on number of fixation values.	. 177
E.1. First information dashboard used in Study V	. 178
E.2. Second information dashboard used in Study V	. 178
E.3. Third information dashboard used in Study V.	. 179
E.4. Fourth information dashboard used in Study V	. 179
E.5. First e-mail used as an IT-mediated Interruption in Study V	. 180
E.6. Second e-mail used as an IT-mediated Interruption in Study V	. 180
E.7. Third e-mail used as an IT-mediated Interruption in Study V	. 181
E.8. Fourth e-mail used as an IT-mediated Interruption in Study V	. 181
E.9. The fixation cross position before first visit in Study V	. 182
E.10. The fixation cross position after gaze-based TRS in Study V	. 182
E.11. The control condition in Study V.	. 183
E.12. The heatmaps of users with low WMC in first visit and mandatory revisit	
phases of the experiment in Study V	. 184
E.13. The heatmaps of users with high WMC in first visit and mandatory revisit	
phases of the experiment in Study V	. 185

List of Tables

3.1.	Summary of future research directions based on identified perspectives 52 $$
5.1. 5.2.	The results from Turkey test by comparing the dwell-times of participants on six AOIs after finishing the data exploration task
6.1.6.2.	MRs and DPs of designing attentive information dashboards that provide individualized VAF for data exploration tasks
6.3.	Comparing ARM performance of the users in both groups
7.1.	Summary of MRs and DPs for designing attentive information dashboards that support users' task resumption
7.2.	Summary of measurements used in Study V
7.3.	Summary of findings by comparing gaze-based TRS and the role of WMC 138
 8.1. 8.2. 8.3. 8.4. 8.5. 	Theoretical contributions according to theory types by Gregor (2006) 143 Nascent design theory: purpose and justificatory knowledge
A.1. A.2.	Coding table of eye-based IIS literature review: context and task
A.3.	Coding table of eye-based IIS literature review: sense-reasoning and focus. 165
A.4.	Coding table of eye-based IIS literature review: perception and behavior 166
A.5.	Coding table of eye-based IIS literature review: performance
D.1.	Control variables and items used for Study IV
D.2.	Comparing the control variables – wilcoxon signed-rank test
E.1.	Items used in the final survey of Study V

List of Abbreviations

ANOVA	Analysis of Variance
AOI	Area Of Interest
AR	Augmented Reality
ARA	Attentional Resource Allocation
ARM	Attentional Resource Management
ASR	Attention Shift Rate
AUI	Attentive User Interface
BI	Business Intelligence
BI&A	Business Intelligence and Analytics
DP	Design Principle
DSR	Design Science Research
DSS	Decision Support Systems
EEG	Electroencephalography
EIS	Executive Information Systems
fMRI	Functional Magnetic Resonance Imaging
HCI	Human-Computer Interaction
HIP	Human Information Processing
IS	Information Systems
IIS	Interactive Intelligent Systems
MIS	Management Information Systems
MR	Meta-requirement
RQ	Research Question
RSR	Resumption Success Rate
SDK	Software Development Kit

- SLRSystematic Literature ReviewTRSTask Resumption SupportUIUser InterfaceVAFVisual Attention FeedbackVRVirtual Reality
- WMC Working Memory Capacity

1. Introduction ¹

1.1. Motivation

Already in 1971, Herbert Simon pointed out that "in an information-rich world, the wealth of information means a dearth of something else: a scarcity of whatever it is that information consumes ... it consumes the attention of its recipients". (Simon, 1971, pp. 40-41). Following this idea, Goldhaber (1997) and Davenport and Beck (2001) have articulated the concept of "attention economy" emphasizing that human attention should be considered as a scarce commodity and should be treated as a new currency of business. When humans receive intensive information, they process it through a selective filter (Broadbent, 1958) which is considered as paying attention (Driver, 2001). Human attention is the core for perceptual and cognitive operations (Chun et al., 2011) and the reason for the selective processing of information is humans' limitation in attentional resources (Broadbent, 1958; Kahneman, 1973).

These days, the possibility to have instant access to huge amounts of information besides humans' limited attentional resources result in the battle of capturing users' attention by companies (Ahn et al., 2018; Hong et al., 2004b; Networks and PWC, 2016; Shen et al., 2015). According to the CEO of Microsoft, Satya Nadella: "we are moving from a world where computing power was scarce to a place where it now is almost limitless, and where the true scarce commodity is increasingly human attention" (Gausby, 2015, pp. 4). Recent studies show that the amount of time concentrating on a task before becoming distracted, decreased massively during the last years (Gausby, 2015; Statistics Brain, 2015). This means that these days users allocate their attention to tasks for only a short time and shift their attention rather fast. However, having proper attention allocation plays an important role in information processing and makes it possible to focus on the important information to pursue goals (Atkinson and Shiffrin, 1968; Wickens et al., 2016). Given this situation, supporting users in managing their limited attentional resources is one of the most pressing and difficult challenges in practice and research in today's information-rich world (Anderson et al., 2018; Bulling, 2016; Davern et al., 2012; Lerch and Harter, 2001).

With the use of big data technologies, organizations collect and analyze data from various resources to assist users in making better decisions (Günther et al., 2017). However, the path from data to decision is typically complex (Keim et al., 2008). Collecting data, extracting insights, and creating value is a challenging endeavor for companies and includes many activities. One essential activity is enabling decision-makers seamless access to data (Delen and Ram, 2018). A well-known class of Information Systems (IS) that supports such data-driven decisions are Business Intelligence and Analytics (BI&A) systems (Chen et al., 2012). As Figure 1.1 shows, the value chain of BI&A systems in companies can be organized in two stages. In the first stage, the collected raw data is transferred to insights by applying

¹This Chapter is based on the following studies which are published or in work: Toreini and Morana (2017), Toreini et al. (2018c), Toreini et al. (2018b), Toreini and Langner (2019), Toreini et al. (2020c), Toreini et al. (2020c), Toreini and Maedche (2020), Toreini et al. (2020a)

different analytical techniques leveraging comprehensive BI&A infrastructures. This stage is critical since having high-quality insights is a foundation to make better decisions (Fink et al., 2017). Because of this capability, BI&A systems became a ubiquitous subject for companies' business performance and competitiveness in last years (Elbashir et al., 2008; Peters et al., 2016).

The second stage is extracting value from the provided insights by making proper decisions. The successful implementation of the underlying processes in this stage is the responsibility of human decision-makers. There is a need to empower them to leverage the extracted insights effectively with advanced interaction technologies. Human-Computer Interaction (HCI) field deals with developing interaction technologies and making the usage of systems effective (Preece et al., 2015). For BI&A systems, one prominent interaction technology is so-called information dashboards, which are increasingly popular forms of visualizing information (Behrisch et al., 2018; Pauwels et al., 2009; Preece et al., 2015; Yigitbasioglu and Velcu, 2012). Few (2006) has described information dashboards as *"visual displays of the most important information needed to achieve one or more objectives; consolidated and arranged on a single screen so the information can be monitored at a glance"*. Information dashboards are known as one of the most effective BI&A tools (Negash and Gray, 2008). They should be designed to present insights in a comprehensive way and be effective for decision-makers (Bačić and Fadlalla, 2016; Pauwels et al., 2009; Phillips-Wren et al., 2015; Yigitbasioglu and Velcu, 2012).



Figure 1.1.: Value chain of BI&A systems and the role of computers and humans.

Many BI&A systems fail to provide benefits to organizations because of improper design and usage of interaction technologies, including information dashboards (Deng and Chi, 2012; Schwarz et al., 2014; Trieu, 2017). In fact, the challenge for organizations is not to collect more information and derive insights as the first stage, but to use the information in an effective way as the second stage (Lerch and Harter, 2001). In this stage, users' cognition plays an important role while making business decision (Chen and Lee, 2003; Niu et al., 2013). However, humans have limited cognitive abilities (e.g., attention, working memory, etc.) that affect their performance while working with information dashboards (Davern et al., 2012; Lerch and Harter, 2001; Yigitbasioglu and Velcu, 2012). Extensive research on information visualization have shown that different types of charts could help to overcome human cognitive limitations (Dilla et al., 2010; Healey and Enns, 2012; Kelton et al., 2010), such as the limited capacity of attention and working memory (Haroz and Whitney, 2012). Both attention and working memory are known as a limited resource of humans, and they play an important role in information processing (Atkinson and Shiffrin, 1968; Wickens et al., 2016), constructing decisions (Orquin and Mueller Loose, 2013) and complex cognitive tasks, such as comprehension, reasoning, and problem-solving (Engle, 2002). Although providing visualized information is a practical approach for addressing these limitations (Borkin et al., 2016,1; Healey and Enns, 2012; Somervell et al., 2002; Ward et al., 2010), presenting several charts on one screen in the form of an information dashboard can rechallenge the boundaries of cognitive resources such as allocation of attentional resources (Toreini and Langner, 2019). The allocation of attentional resources is the set of processes enabling and guiding the selection of incoming perceptual information (Eriksen and Yeh, 1985). Understanding different information is strongly limited to the selection of attended locations (Itti and Koch, 2001; March and Shapira, 1987). Therefore, proper allocation of attentional resources is necessary to analyze business insights while processing information dashboards (Lerch and Harter, 2001; Singh, 1998). Thus, information dashboards should consider users' cognitive limitations in their design and functional features. However, existing research on BI&A systems is limited to their business significance and widespread use and providing solutions regarding corresponding users' cognitive challenges while working with BI&A systems is a research gap (Browne and Parsons, 2012; Chen and Lee, 2003; Davern et al., 2012; Niu et al., 2013). Several researchers have suggested a synergistic collaboration of BI&A and HCI to find efficient solutions for handling the huge amount of collected data (Chen et al., 2012; Holzinger, 2013; Keim et al., 2008).

Users' eye movements have been shown to provide insights into users' cognitive process (Hayhoe and Ballard, 2005; Liversedge and Findlay, 2000; Rayner, 1998). More recently, eye tracking technology usage increased considerably in different research areas primarily because of the availability of cheaper, faster, more accurate, and easier to use eye trackers (Duchowski, 2017). In general, Duchowski (2002) has broadly categorized eve tracking applications into the classes of diagnostic and interactive. Researchers use eye tracking technology to provide quantitative evidence of the user's cognitive processes in diagnostic applications. Diagnostic eye tracking applications collect user's eye movement data while doing a task, and later, researchers use these off-line records for the evaluation. The second category comprises interactive eye tracking applications that interact with the user based on observed eye movement data. In this case, the system uses the user's eye movement data in real-time and enables eye-based interactions. The off-line mode used for diagnostic purposes has been the most well-known usage scenario of eye trackers and researchers have intensively used it in various fields (Jacob and Karn, 2003; Lai et al., 2013; Sharafi et al., 2015; Tien et al., 2014; Wedel and Pieters, 2008). This usage is popular since it provides objective data about users, which is complementary to subjective self-reported data (Dimoka et al., 2012). Also, eye tracking has been extended in IS research with the primary focus on understanding users visual behaviour and the diagnostic usage (Dimoka et al., 2012; Riedl et al., 2017).

Furthermore, with the advancement of eye tracking technology during the last years, researchers have increasingly leveraged the possibility of collecting and processing user eye movement data in real-time. The real-time mode has created new opportunities for re-

4

searchers beyond the off-line mode by actively involving the user in an interactive system, deriving cognitive dimensions and building intelligent systems that are sensitive to the cognition of the user (Bulling and Gellersen, 2010; Bulling et al., 2011). Jameson and Riedl (2011) introduce Interactive Intelligent Systems (IIS) as systems that focus on intelligent technology and user interactions. I consider systems that integrate users' eye movement data in real-time to design IIS as eye-based IIS. The integration of real-time eye movement data as input for designing eye-based IIS has grown during the last years (Chuang et al., 2019; Nakano et al., 2016). The primary usage of the eve movement data in real-time is to track users' attention (Carrasco, 2011; Kowler, 2011). In the HCI field, an eye-based IIS that is sensitive to users' attention and assist them in attentional processes is called Attentive User Interface (AUI) (Bulling, 2016; Roda and Thomas, 2006; Vertegaal, 2003). The research on AUIs arose from the idea that processing huge amounts of information surrounding users is difficult since their attention is a limited resource (Anderson et al., 2018; Bulling, 2016). The need for designing AUIs is emphasized by HCI researchers during last years (Anderson et al., 2018; Bailey and Konstan, 2006; Bulling, 2016; Roda, 2011; Roda and Thomas, 2006). Furthermore, the users' eye movement data is considered as the primary data source for designing AUIs (Bulling, 2016; Majaranta and Bulling, 2014; Roda and Thomas, 2006).

IS researchers also have suggested using eye movement data to design innovative IS applications that enhance users' capability (Davis et al., 2014; Dimoka et al., 2012; Maglio et al., 2000; Riedl and Léger, 2016; vom Brocke et al., 2013). In BI&A field, the usage of eye tracking technology was limited to diagnostic purposes so far (Kurzhals et al., 2016). However, researchers have called for integrating eye tracking technology to BI&A systems and design innovative features that support decision-makers while using these systems based on real-time eye movement data (Silva et al., 2019). Therefore, in this thesis, I focus on closing this gap by designing innovative information dashboards using eye tracking technology as **Attentive Information Dashboards**. I define the term attentive information dashboard as follows:

"An information dashboard that is sensitive to the decision-makers' attention and supports managing limited attentional resources".

Based on this definition and depicted in Figure 1.2, an attentive information dashboard has two major components that include sensory and attention support. First, the sensory component focuses on making the dashboard sensitive to the users' attention. To reach this goal, I developed a system to track users' eye movement data by integrating a low-cost desktop-mounted eye tracker (Farnsworth, 2019), Tobii 4C eye tracker. Collecting, analyzing, and activating attention support features are the three key steps in this component.



Figure 1.2.: Major components of attentive information dashboards (Figure is adapted from Riedl and Léger (2016)).

The attention support component focuses on assisting information dashboard users in managing their limited attentional resources in different situations. Such assistants can be done by adapting the system automatically (e.g., (Buscher et al., 2012; Ishii et al., 2013; Maglio et al., 2000)), or increasing the users' self-awareness about attentional challenges (e.g., (D'Angelo and Gergle, 2018; Kern et al., 2010; Mariakakis et al., 2015)). Users need to scan complex displays systematically to extract the relevant information (Proctor and Vu, 2006). However, users of decision support systems have difficulty remembering activities that have been accomplished and need cognitive aids to remind them (Singh, 1998). Also, general research shows that users have difficulty to recall what they saw, where they looked and in what order (Dimoka et al., 2012). This situation can results in missing information on information dashboards during data exploration tasks. Providing users with feedback during interacting with UIs is known as one of the most basic and important usability principles (Nielsen, 1993). Preece et al. (2015, p.26) have described feedback as "sending back information about what action has been done and what has been accomplished while allowing the person to continue with the activity". Various feedback types are available in digital environments for assisting users in accomplishing their tasks (Morana et al., 2017). One of them is cognitive feedback that presents information about user's cognitive strategy (Lim et al., 2005; Nah and Benbasat, 2004). Prior studies have shown the benefit of using cognitive feedback for users to do their tasks (Balzer et al., 1989; Nah and Benbasat, 2004; Sengupta and Te'eni, 1993). Also, providing feedback about users' attention using eye tracking technology has shown positive effects in information processing (e.g., (D'Mello et al., 2012; Qvarfordt et al., 2010; Sharma et al., 2016)). Therefore, as the attention

support component, I propose providing feedback on how users visually explore information dashboards. I name this type of feedback as individualized **Visual Attention Feedback** (VAF) and define it as:

"A type of feedback that leverages users' real-time eye movement data to increase their awareness about the previous attentional resource allocation".

This type of feedback is a self-tracking feature and has the aim to increase users' awareness. Self-tracking techniques support gaining self-knowledge about own behaviors and habits and are becoming an emerging trend (Choe et al., 2014; Lupton, 2016; Rivera-Pelayo et al., 2017). However, there is a lack of evidence on how such techniques can influence the users' performance in workplaces (Rivera-Pelayo et al., 2017). Furthermore, there is limited research on using eye movement data as self-tracking feedback for IS applications (Lux et al., 2018). In this thesis, I focus mainly on providing individualized VAF to increase users' self-awareness for two situations that users have difficulty in managing their limited attentional resources: data exploration and resuming interrupted tasks.

First, the individualized VAF focuses on supporting data exploration tasks and avoiding missing important information. The users' tasks with information dashboards can be distinguished into search tasks and data exploration tasks (Vandenbosch and Huff, 1997). In a search task, the user seeks answers to specific questions. In data exploration tasks, the user browses the dashboard generally to get a comprehensive understanding of the current status. In this case, the user examines data without having prior understanding of what information it might contain (Baker et al., 2009). Gartner-Magic Quadrant for BI&A has emphasized that well-designed information dashboards that enable the exploration of data and support proper decision making as a critical capability of BI&A systems (Sallam et al., 2017). However, even with a well-designed dashboard, exploring the compressed amount of visualized information is challenging for users (Baskett et al., 2008; Figl and Laue, 2011; Haroz and Whitney, 2012; Healey and Enns, 2012; Sedig and Pasob, 2013). Particularly, information dashboards have the potential to create difficulties in managing attentional resources since users can only focus on a limited set of information and can miss other parts while exploring the information dashboard (Alberts, 2017; Dilla et al., 2010; Lurie and Mason, 2007). Such a task competes for the decision-maker's attentional resources (Lerch and Harter, 2001).

Second, the VAF focuses on supporting the users' difficulty in resuming an interrupted task. The interest in BI&A systems has increased in recent years because of the opportunities associated with data and analysis (Chen et al., 2012) and the increased usage of data-driven decision making. Meanwhile, these days the workplace is accompanied by frequent interruptions of task execution (Czerwinski et al., 2004; Mark et al., 2008). For example, operational dashboards are used to monitor the process execution in manufacturing in real-time, while interruptions can negatively affect their performance. Interruptions are known to have various negative impacts, such as higher task completion time, the number of errors, or anxiety in multitasking situations (Addas and Pinsonneault, 2015; Borst et al., 2015; Czerwinski et al., 2004; Gillie and Broadbent, 1989). Also, in the workplaces,

the increasing number of interruptions affect decision-makers' primary task performance (Galluch et al., 2015; Gupta et al., 2013; Ou and Davison, 2011). Employees have difficulty resuming their primary task after an interruption and having delay getting back on the task (Hemp, 2009; Mark et al., 2005). Even though task disruptions can sometimes be unavoidable, e.g., due to the importance of the secondary task or the specific context of work (Dostal et al., 2013), it is necessary to design systems that provide advanced support in better interruption management (Anderson et al., 2018; Bailey and Konstan, 2006). Leveraging such systems to detect an interruption and provide Task Resumption Support (TRS) by highlighting previously attended areas have seen to be supportive for users in task resumption (Göbel and Kiefer, 2019; Jo et al., 2015; Kern et al., 2010; Mariakakis et al., 2015). A promising approach to provide TRS is using eye tracking technology. This technology tracks the user's eye movements to understand the moment of task switching and subsequently visualize the user's eye movement data as an indicator of the most recent area of attention (Dostal et al., 2013; Jo et al., 2015; Kern et al., 2010; Mariakakis et al., 2015). Gaze-based TRS makes previous cognitive processes more explicit by providing memory aids. It gives hints to the users to remember what they were thinking or what might have been their intention before shifting to the interruption task (Majaranta and Bulling, 2014).

In summary, in this thesis, I deliver an innovative solution (attentive information dashboards with individualized VAF) for a real-world problem (managing limited attentional resources) following the Design Science Research (DSR) methodology. DSR has the aim to solve significant social or organizational problems by designing artifacts (Hevner et al., 2004). The challenges and strategies of designing attentive information dashboards and individualized VAF are described as Research Questions (RQs) in the subsequent section.

1.2. Research Gaps and Associated Research Questions

With this thesis, I explore designing attentive information dashboards and support users in managing their limited attentional resources in different situations. Therefore, I focus on answering the following main RQ in this thesis:

Main RQ: How to design attentive information dashboards for BI&A systems that enhance users' ability to manage attentional resources?

To answer the main RQ step-by-step, I defined five sub-RQs can be seen in Figure 1.3. Next, I designed and executed five studies to address each sub-RQs. In the following, I present the research gaps and the assigned sub-RQs that are covered in this thesis.

Main RQ How to design attentive information dashboards for BI&A systems the enhance users' ability to manage attentional resources?	at
---	----



Figure 1.3.: Overview of research questions addressed in this thesis.

The first **RQ** deals with state-of-the-art and possible future research directions in using eye tracking technology for designing IIS in general. Although the number of research studies focusing on eye-based IIS increased during the last years (Chuang et al., 2019; Nakano et al., 2016), there is a lack of common understanding of this class of systems. The current eve tracking literature reviews focus mainly on summarizing the usage of eve trackers in specific fields concentrate on diagnostic or interactive purposes (Duchowski, 2002; Jacob and Karn, 2003) rather than integrating real-time eve movement data for designing IIS. For example, Velloso and Carter (2016) presented the review of gazebased interaction systems in gaming, Sharafi et al. (2015) discussed the advantages of eye tracking for software engineering, Tien et al. (2014) discussed the role of eye tracking technology in medical research, Lai et al. (2013) presented how eye tracking technology can be used to describe the process of learning, Rosch and Vogel-Walcutt (2013) presented the implementation of eye tracking into training environments, Wedel and Pieters (2008) discussed the usage of eye tracking technology in marketing, and Vasseur et al. (2019) presented the review of using eye tracking in IS field. To the best of my knowledge, summarizing existing research in using real-time eye movement data for designing eyebased IIS (e.g. AUIs) is a research gap. Therefore, as the first step, this thesis focuses on closing this gap by answering the following RQ in **Study I**:

RQ1: What is the state-of-the-art and potential future research directions for designing eye-based IIS?

I addressed this RQ by conducting a Systematic Literature Review (SLR) study and identifying the state-of-the-art in eye-based IIS field. Based on this analysis, I provide a list of future research directions.

The second RQ deals with identifying attentional problems when users explore information dashboards and derive Meta-requirements (MRs) for designing innovative information dashboards. Visualized information supports users to process huge amounts of information by overcoming their limitations in attentional resources and working memory (Borkin et al., 2013; Healey and Enns, 2012; Somervell et al., 2002; Ward et al., 2010), but it is not clear how they play a role while exploring information dashboards. Few studies have addressed users' attentional challenges while working with information dashboards (Alberts, 2017). The primary focus is to evaluate the design features (e.g., presentation formats, colors, size, etc.) and their impact on users' attention (Bera, 2014,1; Burch et al., 2011; Nadj et al., 2020) rather than users difficulties in managing limited attentional resources. It is important since to analyze business insights; it is necessary to have proper allocation of attentional resources (Lerch and Harter, 2001; Singh, 1998). Understanding different information objects is strongly limited to the selection of attended locations (Itti and Koch, 2001; March and Shapira, 1987) and attention processes play an active role in constructing decisions (Einhorn and Hogarth, 1981; Orquin and Mueller Loose, 2013). Besides, it has been studied how individual user characteristics such as working memory impact the effectiveness of visualization techniques (Steichen et al., 2013; Toker et al., 2013), but again this individual characteristic is not investigated during information dashboards exploration. Designing effective supports requires a detailed understanding of the underlying cognitive processes (Lerch and Harter, 2001). Cognitive limitations and related errors are among under-researched topics in the IS field, and there is a general need for more research on that (Browne and Parsons, 2012). Also, few researchers have examined BI&A systems and users' cognitive limitations (Davern et al., 2012; Niu et al., 2013). Particularly, researchers have emphasized the need to study individual cognitive limitations on the effectiveness of information dashboards (Pauwels et al., 2009; Yigitbasioglu and Velcu, 2012). Using eye tracking technology to assess the impact of systems on users' information processing capacity is considered as a research opportunity by IS researchers (Dimoka et al., 2012). Therefore, I focused on closing this research gap by comparing how individuals with different Working Memory Capacity (WMC) allocate their attentional resources during data exploration tasks. With this focus, I aim to find MRs for designing attentive information dashboards by addressing the following RQ in **Study II**:

RQ2: What are meta-requirements for designing dashboards that consider users' cognitive limitations of attentional resources and working memory?

I addressed this RQ by conducting an exploratory study using eye tracking technology for diagnostic purposes to explore the impact of attention and working memory limitations on the effectiveness of information dashboards. The results from RQ2 highlight the user challenges in managing limited attentional resources while exploring information dashboards. Therefore, the third RQ deals with examining different solutions to support managing attentional resources with the usage of eye tracking technology. The results from RQ1 show the user adaptation approach's effectiveness by increasing users' awareness of previous visual behavior. Furthermore, the results show a trend in using this approach by applying low-cost eye tracking technology. Maglio et al. (2000) have suggested observing user behavior through multiple resources, including eye tracking technology and adapting IS applications by displaying relevant information to the user. However, the usage of eye movement data for designing feedback and increase users-awareness is a research gap (Lux et al., 2018). Such awareness can be either delivered by the usage of off-line records of eye movement data from the users that accomplished the same task previously (e.g., (Sridharan et al., 2012)) or by using users' real-time eye movement data (e.g., (Qvarfordt et al., 2010)). Although eye tracking is the dominant tool for IS studies that integrate neuroscience tools (Riedl et al., 2017), the results from RQ1 show a lack of research on integrating real-time data for designing innovative IS applications. Furthermore, off-line recordings are mainly used to support designers in understanding users' behavior and improving the design rather than supporting users by providing them as VAF. Both using eye movements from previous users and using real-time eye movement data for designing support features in BI&A systems are suggested by researchers in the intersection between BI&A and eye tracking field (Silva et al., 2019). Also, IS researchers have called for the usage of eye tracking technology to design innovative IS applications that enhance users' capability (Davis et al., 2014; Dimoka et al., 2012; Riedl and Léger, 2016; vom Brocke et al., 2013). However, best of my knowledge, none of these techniques is investigated so far. Therefore, I focus on closing this research gap by addressing the following RQ in **Study III**:

RQ3: What type of VAF enhances users' ability to manage attentional resources while exploring information dashboards?

I addressed this RQ by conducting an eye tracking pilot study for comparing three different types of VAF (one real-time usage and two off-line usages of eye movement data) and their impact on managing limited attentional resources during data exploration tasks.

The **fourth RQ** deals with designing attentive information dashboards for data exploration tasks. Such a task competes for the decision-maker's attentional resources (Lerch and Harter, 2001). Also, findings from RQ2 highlights users' challenges in managing limited attentional resources of users when performing the data exploration task. By answering RQ3, the initial findings indicated that using real-time eye movement data for individualized VAF works better than off-line eye movement data as general VAF types. Therefore, in a subsequent study, I investigate the design of individualized VAF and its influence on data exploration tasks in more detail. Previous studies in other fields have shown the positive impacts of individualized VAF as a self-tracking feature on managing attentional resources (Deza et al., 2017; D'Mello et al., 2012; Göbel and Kiefer, 2019; Qvarfordt et al., 2010; Sharma et al., 2016; van Gog et al., 2009). Although using eye tracking technology is popular among IS researchers (Riedl et al., 2017), no study has examined using eye movement data for designing individualized VAF (Lux et al., 2018). Furthermore, researchers have suggested to integrate eye trackers for designing supportive features for BI&A users (Silva et al., 2019). Therefore, there is a lack of knowledge on how to design individualized VAF and what is its effect on users of information dashboards. To close this gap, I addressed the following RQ in **Study IV**:

RQ4: How to design attentive dashboards providing individualized VAF to enhance users' ability to manage attentional resources for data exploration tasks?

I addressed this RQ by comparing the effects of individualized VAF with the condition that users did receive a general text-based explanation as VAF in a large-scale laboratory eye tracking study.

The fifth RQ deals with providing attentive information dashboards with TRS. The results from RQ2 highlighted user's difficulty in resuming an interrupted task. Previous studies also show that the workplace is also accompanied by frequent interruptions of task execution (Czerwinski et al., 2004; Mark et al., 2008) and employees have have a long delay in getting back to the primary task after interruptions (Hemp, 2009; Mark et al., 2005). The results from RQ1 reveal that existing TRS features integrated with eye trackers are not investigated in workplaces. Also, Anderson et al. (2018) have noted the necessity to design systems that provide advanced support in better managing interruptions. Therefore, as another type of individualized VAF. I decided to leverage eve tracking to provide TRS for information dashboard users. Previous studies have used eye tracking technology to collect and then visualize the user's eve movement data as an indicator of the most recent area of attention after resuming an interrupted task (Dostal et al., 2013; Jo et al., 2015; Kern et al., 2010; Mariakakis et al., 2015). As the results from RQ1 shows, existing gaze-based TRS systems mainly focused on reading tasks. No research has investigated providing such support for performing data exploration tasks on information dashboards. Therefore, there is a lack of knowledge on how to design it and its effect on users of information dashboards. To close this gap, I addressed the following RQ in Study V:

RQ5a: How to design attentive information dashboards providing individualized VAF to enhance users' ability to manage attentional resources in resuming interrupted tasks?

Besides, previous studies have shown that user characteristics such as WMC play an essential role in managing interruptions (Cane et al., 2012; Foroughi et al., 2016; Mark et al., 2008; Ratwani and Trafton, 2008; Werner et al., 2011). Also, theoretical frameworks about interruption emphasize the role of working memory in handling interruptions (Altmann and Trafton, 2002; Borst et al., 2015). Users with higher WMC can better remember their previous visual behavior and thus better deal with task resumptions. Moreover, it is considered as an essential individual characteristic while users are working with visualized information such as on information dashboards (Haroz and Whitney, 2012; Healey and Enns, 2012; Toker et al., 2013). Also, the findings from RQ2 suggest designing VAF types that fit users WMC. However, reviewing gaze-based TRS studies collected in RQ1, highlights that the impact of gaze-based TRS under consideration of WMC is a research gap. There is a need for further research on how to present TRS to the user based on their individual characteristics (Anderson et al., 2018). To close this gap, I addressed the following RQ in **Study V** as well:

RQ5b: What is the role of working memory capacity in effectiveness of gazebased task resumption supports?

Therefore, in this study, I focus on designing and understanding the impact of different highlighting methods for gaze-based TRS (last point, heatmap, scanpath) under consideration of a specific task and a specific user characteristic.

1.3. Thesis Structure

Figure 1.4 presents an overview of this thesis with 9 chapters. Chapter 1 motivates the topic and provides an overview of the entire thesis. In Chapter 2, I first present the conceptual foundations relevant to this thesis including attention, working memory, and human information processing theory. Additionally, I present related work studies relevant to this thesis including eye tracking technology, AUI, BI&A systems, and information dashboards. Subsequently, I address the first RQ in Chapter 3 as Study I. In this chapter, I first present a conceptual framework for the systems that integrate real-time usage of eye movement data to develop innovative interaction, considered as eye-based IIS. Later, I summarize existing knowledge in this field by conducting a SLR study. Moreover, I provide a list of future research directions for eye-based IIS as an outcome of this analysis. Later in Chapter 4, I introduce the research methodology of the entire DSR project with three design cycles that were investigated in each chapter of this thesis.

Chapter 5 presents the results from the first design cycle. This design cycle includes two studies addressing RQ2 (Study II) and RQ3 (Study III). In Study II of this thesis, I addressed RQ2 by conducting an exploratory study using eye tracking technology and investigated the impact of attention and working memory limitations on the effectiveness of information dashboards. This study's results highlight the need to design attentive information dashboards to support users in managing their limited attentional resources during data exploration and task resumption after an interruption. Therefore, I articulated initial MRs for designing attentive information dashboards. In the Study III of this thesis, I addressed RQ3 by designing three types of VAF (two based on off-line eye movement data and one based on real-time eye movement data) and conducted an eye tracking pilot study to evaluate their effectiveness. The results highlight that an individualized VAF using gaze data in real-time supports users in a better way compared to other VAF types.

Chapter 6 presents the results from the second design cycle, which addresses RQ4 (Study IV). In this design cycle, I articulate theoretically grounded Design Principles (DPs) and instantiate a software artifact leveraging users' eye movement data in real-time to provide attentive information dashboards that support data exploration with individualized VAF. Later, I evaluated the instantiated software artifact in a large-scale laboratory experiment

with 92 participants as a suggested method to evaluate DSR projects (Pries-Heje et al., 2008; Venable et al., 2012). Analysis of users' eye movements reveals that the suggested DPs have a positive effect on enhancing users' ability to manage attentional resources.

Chapter 7 present the results of the third design cycle, which addresses RQ5 (Study V). In this design cycle, I propose attentive information dashboards that support users in resuming an interrupted task. Eye trackers are a promising technology to provide TRS using gaze-based highlighting methods. Consequently, with gaze-based TRS as individualized VAF, users are reminded regarding their previous visual behavior and assisted in resuming the primary task more efficiently in case of interruptions. In this design cycle, I first present dimensions that impact designing effective gaze-based TRS by analyzing previous research in the field of interruption and gaze-based TRS. Next, I articulated theoretically grounded DPs and instantiate a software artifact leveraging users' eye movement data in real-time to provide attentive information dashboards with three types of gaze-based TRS. Later, I evaluated three gaze-based highlighting methods (last point, heatmap, scanpath) after a short-term IT-mediated interruption (Addas and Pinsonneault, 2015) in an eve tracking study (N=48) and compared their effectiveness based on users' WMC. The results suggest that the need for gaze-based TRS types is different for users with high and low WMC. Notably, the heatmap highlighting method is supportive for low WMC users, while users with high WMC may not need gaze-based TRS for short-term IT-mediated interruptions.

Chapter 8 presents the theoretical and practical contributions of this thesis. Moreover, I present the prescriptive knowledge created in this DSR project in the form of a nascent design theory. I also discuss the knowledge contribution of this DSR project according to the DSR knowledge contribution framework by Gregor and Hevner (2013). Furthermore, I present the limitation of this thesis and provide future research suggestions. Finally, in **Chapter 9**, I conclude the thesis.

	Chapter 1. Introduction	
Motivation	Research Questions	Thesis Structure
Chap	ter 2. Conceptual Founda	tions
2.	1: Theoretical Foundation	s
Attention	Working Memory	Human Information Processing
	2.2: Related Work	
Eye Tracking TechnologyAttentInter	ive User rfaces BI&A System	ns Information Dashboards
Chapter 3. Stat <u>Study I (RQ1)</u> : Eye-base	te-of-the-Art and Conceptue ed IIS: State-of-the-Art and Fu	ual Framework ture Research Direction
Chap	pter 4. Research Methodol	ogy
Design Cycle 1	Design Cycle 2	Design Cycle 3
Chapter 5	Chapter 6	Chapter 7
Chapter 5 Study II (RQ2) Exploring attention management problems on information dashboards Study III (RQ3) Exploring possible solutions: real-time Vs. off-line records of eye movement data as VAF	Chapter 6 <u>Study IV (RQ4)</u> Instantiating and evaluating attentive information dashboards with individualized VAF for data exploration tasks	Chapter 7 <u>Study V (RQ5)</u> (a) Instantiating attentive information dashboards that support resuming interrupted tasks (b) Evaluating the effect of working memory capacity on different gaze- based task resumption supports
Chapter 5 <u>Study II (RQ2)</u> Exploring attention management problems on information dashboards <u>Study III (RQ3)</u> Exploring possible solutions: real-time Vs. off-line records of eye movement data as VAF	Chapter 6 <u>Study IV (RQ4)</u> Instantiating and evaluating attentive information dashboards with individualized VAF for data exploration tasks Chapter 8. Discussion	Chapter 7 <u>Study V (RQ5)</u> (a) Instantiating attentive information dashboards that support resuming interrupted tasks (b) Evaluating the effect of working memory capacity on different gaze- based task resumption supports
Chapter 5 <u>Study II (RQ2)</u> Exploring attention management problems on information dashboards <u>Study III (RQ3)</u> Exploring possible solutions: real-time Vs. off-line records of eye movement data as VAF Theoretical Contributions	Chapter 6 <u>Study IV (RQ4)</u> Instantiating and evaluating attentive information dashboards with individualized VAF for data exploration tasks Chapter 8. Discussion Practical Contributions	Chapter 7 <u>Study V (RQ5)</u> (a) Instantiating attentive information dashboards that support resuming interrupted tasks (b) Evaluating the effect of working memory capacity on different gaze- based task resumption supports Nascent Design Theory
Chapter 5 <u>Study II (RQ2)</u> Exploring attention management problems on information dashboards <u>Study III (RQ3)</u> Exploring possible solutions: real-time Vs. off-line records of eye movement data as VAF Theoretical Contributions	Chapter 6 Study IV (RQ4) Instantiating and evaluating attentive information dashboards with individualized VAF for data exploration tasks Chapter 8. Discussion Practical Contributions owledge Limitat Discussion	Chapter 7 <u>Study V (RQ5)</u> (a) Instantiating attentive information dashboards that support resuming interrupted tasks (b) Evaluating the effect of working memory capacity on different gaze- based task resumption supports Nascent Design Theory ions and Research



2. Conceptual Foundations ¹

2.1. Overview

In this thesis, I focus on the intersection between research streams in three fields of studies, including Human-Computer Interaction (HCI), Psychology, and Information Systems (IS). Figure 2.1 presents an overview of the relevant research streams with selected example studies. Furthermore, the research gaps introduced in Section 1.2 are positioned within these research streams.



<u>*Gap4:*</u> Designing attentive information dashboards for data exploration task <u>*Gap5:*</u> Designing attentive information dashboards with task resumption support

Figure 2.1.: Overview of research streams, theoretical foundations and research gaps.

From the **HCI** perspective, this thesis covers the two research streams including eye tracking technology and IIS. IIS are intelligent systems that people interact with and they emphasize the interplay of intelligent technology with advanced interactions of users (Jameson and Riedl, 2011). The first research gap is positioned at the intersection of these two research streams as eye-based IIS. In this thesis, I mainly focus on a specific type of eye-based IIS that is sensitive to users' attention and aims to support their attentional process, called Attentive User Interface (AUI). From the **Psychology** perspective, I investigate the role of attention and working memory, which play an essential role in

¹This Chapter is based on the following studies which are published or in work: Toreini and Morana (2017), Toreini et al. (2018c), Toreini et al. (2018b), Toreini and Langner (2019), Toreini et al. (2020c), Toreini and Maedche (2020), Toreini et al. (2020a)

Human Information Processing (HIP) theory. Finally, from the **IS** perspective, I investigate the role of BI&A systems and information dashboards as a specific component of BI&A systems. The second research gap focuses on the intersection between information dashboards and the psychological constructs of attention and working memory. The third research gap addresses the intersection between information dashboards and the type of eye-based IIS, especially by comparing the real-time and off-line usage of eye movement data to support information dashboard users' attentional process. Furthermore, the fourth and fifth research gaps address the intersection between all three aspects by investigating how to design attentive information dashboards and individualized VAF to support data exploration tasks and resuming interrupted tasks.

In the following, I first describe the relevant theoretical foundations used in this thesis. After that, I outline existing studies within each research stream as related work.

2.2. Theoretical Foundations

2.2.1. Attention

Following the APA dictionary of psychology 2 , attention is discussed as "state in which cognitive resources are focused on certain aspects of the environment rather than on others and the central nervous system is in a state of readiness to respond to stimuli". However, researchers do not offer a dedicated definition for attention in research area (Anderson et al., 2018) and consider it rather generally as selective processing of incoming sensory information (Driver, 2001). The reason for emphasizing on selective processing is humans' limited attentional resources (Chun et al., 2011). Selective attention initially has been introduced as part of Broadbent's filter theory (Broadbent, 1958). It argues that when humans' perceptual system is overwhelmed by information overload, it starts to process the information through a selective filter. The limited capacity of attentional resources is not fixed and can vary based on different conditions such as task and user's characteristic (Kahneman, 1973). An easy task requires little attention, and a difficult task demands more attentional resources. Furthermore, users with different expertise can have different capacities.

One can differentiate attention in goal-directed and stimuli-driven attention as well as covert and overt attention (Desimone and Duncan, 1995). Goal-directed attention is the voluntary type of attention, whereas stimulus-driven attention is involuntary (Corbetta and Shulman, 2002). Researchers consider goal-directed attention as selective attention that is due to the limited attentional resources of humans. In this case, users select stimuli to allocate attention consciously and based on their intention. Additionally, they refer to stimuli-driven attention when external stimuli capture the user's attention unconsciously. Color, orientation, size, motion, depth, etc. are known as guiding representation that involuntary lead users' attention to salient objects (Wolfe and Horowitz, 2004). Treisman and Gelade (1980) discussed these elements and the role of them as "*Feature-Integration Theory of Attention*". Figure 2.2 shows an example of the two different types of attention.

²https://dictionary.apa.org/attention

In this example, the users are asked to find the IISM logo among four existing logos. On the left side, they control the attention voluntarily and guide it to do the task by finding the IISM logo on the bottom-right position. However, in the right side example, changing the brightness of the four logos leads to attracting users' attention to the logo on the top-right first (KIT logo), while they do not have control over their attention allocation. That takes a while to overcome this and shift attention and achieve the goal of finding the IISM logo.



Figure 2.2.: An example of goal-directed and stimuli driven attention. In both pictures the user has the task to find the IISM logo.

Furthermore, Posner (1980) has distinguished overt and covert attention as two other categories of attention. Overt attention is an extrinsic behavior and aids humans to monitor the environment. Also, overt attention guides the users' head-turning and eye movements to an object (Carrasco, 2011). Researchers measured the users' overt attention by using eye tracking technology in previous studies (Kowler, 2011). Just and Carpenter (1980) have proposed the eye-mind assumption that describes the relationship between the pattern of eve movements and the underlying cognitive processes. Based on this assumption, where users are fixating dedicates their overt attention. Also, both goal-directed and stimuli-driven attention control user's eve movements (Orquin and Mueller Loose, 2013). As the other attention type, the covert attention is an inward activity in which the brain attends to an object without any extrinsic behavior. This attention type influences the brain signals. Researchers measure it by leveraging neuroscience tools such as Electroencephalography (EEG) and Functional Magnetic Resonance Imaging (fMRI). Figure 2.3 shows an example of overt and covert attention. In the overt attention (left), the user's gaze direction indicates the attention of the users. Therefore using eye trackers help to tracks overt attentions. However, in the covert attention (right), the user gaze direction is on the monitor, but the attention is allocated to the time.



Figure 2.3.: An example of overt and covert attention.

The attention term used in this study is considered as goal-directed attention and the overt attention tracked by users' eye movement data. To control stimulus-driven attention, I designed a specific dashboard layout, which is discussed separately for each study.

2.2.2. Working Memory

Following the APA dictionary of psychology ³, memory is "the ability to retain information or representation of past experiences, based on the mental processes of learning or encoding, retention across some interval of time, and retrieval or reactivation of the memory". Humans use their memory daily to process information around them, which has many different forms, e.g., images, sounds, or meaning. To explain how the memory works, Atkinson and Shiffrin (1968) proposed the "Multi-Store Model of Memory". Based on this model, the memory comprises three stores: a sensory memory, short-term memory, and long-term memory. Sensory memory stores the raw information that the brain receives from the five senses. After paying attention to such information, the information is encoded, and then the short-term memory receives the input from the sensory memory. Later, rehearsing of information enables the transfer of information to long-term memory and keeps it for a longer time. As Figure 2.4 shows, based on this model, users process information trough three different memory types in a linear way and attention is the key element to transfer information around the user to the short-term memory.



Figure 2.4.: Multi-store model proposed by Atkinson and Shiffrin (1968).

18

³https://dictionary.apa.org/memory

Later, researchers found that this model is oversimplified, and the short-term memory is a more complicated phenomenon that what is explained in the multi-store model of memory. Baddeley and Hitch (1974) have proposed an alternative model called "Baddeley's Model of Working Memory" and argued that the short-term memory is more than just a store. They argued that the short-term memory includes both store and processing information capability simultaneously. Therefore, they replaced the concept of short-term memory, which can only store information with "Working Memory" that can both store and process information. They argued that working memory could further be divided into three component, including the phonological loop, visuospatial sketchpad, and central executive Baddeley and Hitch (1974). The first two components describe in what kind of form, the working memory is storing information. The phonological loop is responsible for storing and processing verbal information; the visuospatial sketchpad focuses on storing and processing visual inputs such as spatial location, shape, size, and color. The third component is the central executive component, which is the most complicated part. This component describes the use of brain resources with working memory. It is also responsible for monitoring and coordinating the operation of other components and related them to long-term memory. Later, Baddeley (2000) added the fourth component as the episodic buffer that focuses on the relationship between short-term and long-term memory, synthesizing information across modalities, and interacting with semantic knowledge. Figure 2.5 shows the four components of working memory, and the relationship between as integrated by Baddeley (2012).



Figure 2.5.: Updated version of Baddeley's model of working memory by Baddeley (2000).

Working memory has an impact on building contemporary global models of cognition and is involved in many complex cognitive behaviors, such as comprehension, learning, reasoning, and problem-solving (Baddeley, 1992; Engle, 2002). Baddeley (2012) reviewed the working memory research and mentioned that the term "working memory" and "shortterm memory" are still on occasion used interchangeably. However, he uses short-term memory to refer to the simple temporary storage of information and uses the working memory term to consider a combination of storage and manipulation. In this thesis, I follow the distinction between short-term memory and working memory and the term working memory used in the following studies refers to more than just a temporary store and consider as an element that influences users ability to control attentional resources.

Furthermore, working memory has a limited capacity (Cowan, 2010; Miller, 1956). Miller (1956) argued that this capacity could be increased by a process known as "chunking" which is grouping pieces of related information. A limiting number of items that can be recalled is known as memory span. It is known as one important individual difference (Baddeley, 1992). The memory span of individuals for the different forms of information (visual, verbal, digit, etc.) is not similar and there are different psychological tests to extract the users' memory span based on the information type (Conway et al., 2005; Wilhelm et al., 2013). This individual difference has an impact on real-world cognitive tasks (Engle, 2002) and general intelligence (Conway et al., 2003). Furthermore, several studies, such as Engle et al. (1999), Kane et al. (2001), and Kane and Engle (2003), have investigated the role of WMC and attention control. The results show that users with high and low capacity have different abilities to control their attentional resources, impacting their task performance.

2.2.3. Human Information Processing

The humans' mind is a information processing system (Card, 1983). HIP theory describes how individuals encode information, capture it in their memory, and retrieve it when needed. Researchers have considered the multi-store model of memory by Atkinson and Shiffrin (1968) and its linear process of information as HIP. However, this model does not fully consider the role of attention and working memory while processing information and has focused on three types of memory and their connections. Therefore, I use the adapted version of HIP stages by Wickens et al. (2016) in this thesis, which describes the relationships between different components of information processing in more details. Figure 2.6 depicts the stages of information processing. This framework includes four primary components: attention, memory, perception, response selection, and execution. I explain each of them in the following by considering the processing visualized information in the form of information dashboard.

"Attention" is the first component and has connections with all the other components. It is considered as a limited resource and the detailed description of it is discussed in Section 2.2.1. Healey and Enns (2012) explain that allocating limited attentional resources while processing visualized information contains two different steps. First, pre-attentive processing that includes methods for drawing the user's stimulus-driven attention. In this step, users encode a stimulus for a short time based on the elements that attract users' attention and perceive the information through their sense organ (e.g., eye, ear, etc.). Later, the central processing starts as the post-attentive processing that focuses on goal-directed attention and processing the perceived information in detail. Furthermore, the interaction between users and information visualization is through their visual systems. Therefore, overt attentional resources are used in this thesis as the attention component. Haroz and Whitney (2012) have investigated the role of limited attentional resources influence while processing information visualization processing. As a result, they found that this limited
resource strongly changes the effectiveness of information visualizations, particularly the ability to detect unexpected information. Therefore, I assume that allocate attention to information dashboards properly influences the effectiveness of them as well.

"Memory" is the second component and different types of it are discussed in detail in Section 2.2.2. This framework comprises three types of memory and presents their relationship similar to what is introduced by Atkinson and Shiffrin (1968) as the multi-store model of memory and further updated by Baddeley and Hitch (1974) with the working memory concept. Based on that, first, the sensory memory stores the raw information that the brain receives from the sense organ (e.g., color, shape, location, etc. of the stimuli) and keeps it for few seconds. Second, working memory stores information temporarily and executes cognitive functions. Users select stimuli from the dashboard and precept it by allocating attention to the sensory memory's collected information besides previous experience collected in the long-term memory (experience level). Later the precept information is transferred to the working memory, which plays an important role in complex cognitive behaviors, such as comprehension, reasoning, and problem-solving (Engle, 2002). However, as discussed in Section 2.2.2, WMC is a limited resource. Researchers also have defined it as one important individual characteristic while working with visualized information (Borkin et al., 2016,1; Healey and Enns, 2012; Toker et al., 2013). Third, the long-term memory that stores the information for a long time. Rehearsing the information from working memory enables that information transfers to long-term memory.

"Perception" is the third component that helps raw data from the environment be interpreted and decoded. Wickens et al. (2016) have emphasized that perception and sensation are different since perception involves determining the *meaning* of sensory information. Perception of visualized information support users in making decisions in later steps (Ware, 2012). Ward et al. (2010) have distinguished the perception of visualized information as processes of recognizing (being aware of), organizing (gathering and storing), and interpreting (binding to knowledge). Therefore, the perception of information presented on information dashboards is considered as determining the meaning of the presented data.

"Response Selection and Execution" is the fourth component focusing on when users want to trigger an action based on the processed information. The selection of responses and the execution of the action are two separate stages. In the selection stage, the user makes a decision among several options. The process of making a decision can vary depending on the task. Also, the user may make a decision immediately, or store information while a decision is formulated. Later in some cases, this response is executed in a manner that requires muscle co-ordination for moving the body or execute some actions on the User Interface (UI). Hence this execution process may change the environment or UI layout (e.g., filtering a chart based on last year's sales amount); it creates new information that needs to be sensed. This loop is shown as the "Feedback" element in the framework that captures that the goal has been achieved.



Figure 2.6.: A model of HIP stages adapted from Wickens et al. (2016).

2.3. Related Work

2.3.1. Eye Tracking Technology

Human's eyes are considered as a powerful sensory organ that they use them while processing information. With existing muscles around the eyes, humans can move them to place the information that interests them on the fovea. Therefore, eyes support perceiving visual impressions of what happens around us. The human's eye movement data demonstrates cognitive processes and gives a hint about what they are thinking or what may be their intention (Hayhoe and Ballard, 2005; Liversedge and Findlay, 2000; Rayner, 1998). This motivates the usage of it for in IS field (Dimoka et al., 2012) and it is also known as the dominant tool for IS studies that integrated neuroscience tools (Riedl et al., 2017).

Eye tracking technologies developed over a century ago to investigate human visual perception (Rayner, 1998). During the last century, it improved from both discovery eye movement facts and the relationship to cognitive behavior besides the recording systems. Using eye tracking technology and collecting eye movement data can describe the overt attention (Kowler, 2011). Yarbus (1967) has investigated the relationship between users' eye movement data and their intentions. Based on that, the participants received a photo and had to inspect it for different tasks. The results show that eye movements' pattern is different based on the viewer's intent or the task assigned. Therefore, knowing where an individual is looking at and tracking their eye movements patterns provides valuable information about user intentions. This capability have motivated researchers in different fields to use eye tracking technology, including neuroscience, psychology, ergonomics, advertising, and design (Richardson and Spivey, 2004). Eye trackers are not only proper for finding users' conscious behavior (e.g., visual attention, cognitive processes and intentions (Kowler, 2011; Majaranta and Bulling, 2014; Raptis et al., 2016)), they are also acknowledged as a promising approach to obtain insights on unconscious behavior (e.g., boredom (Kim et al., 2018), mind wandering (Bixler and D'Mello, 2016)).

Different types of eve trackers exist in the market and have been used by researchers in the last years (Duchowski, 2017; Holmqvist et al., 2011). Modern eye trackers use mainly nearinfrared technology as well as high-resolution cameras to track the gaze direction. They are principally categorized into either desktop-mounted eye trackers (also called remote devices or screen-based) or head-mounted eye trackers (also called mobile eye trackers, eye tracking glasses, etc.). The desktop-mounted eye trackers are used to trace any desktop-based stimulus, while the head-mounted eye trackers are proper for real-life activities. Desktopmounted eye trackers are more comfortable, easy to use, and faster to setup (Morimoto and Mimica, 2005) but limited the technology to the time that users are working with monitors. But using head-mounted devices is appropriate for everyday usage (Bulling and Gellersen, 2010). Head-mounted devices are very accurate and stable; however, desktop-mounted eye trackers may lose their accuracy when users change their distance from the eye tracker. Eye trackers in both categories tend to be expensive. Therefore, some studies have attempted to employ low-cost or open-source alternatives during the last years, e.g., using webcams (Burton et al., 2014; Zugal and Pinggera, 2014). However, in such eye trackers, high accuracy and compatible analytical software are critical for becoming successful. Besides these types, by enabling eye trackers on Virtual Reality (VR) and Augmented Reality (AR) headsets, the system can record the user's visual behavior for usability purposes or design new interactive features in the virtual environment by integrating the user's eves.

The usage of eye tracking technology increased during the last years since there are cheaper, faster, more accurate, and easier to use eye trackers in the market (Duchowski, 2017). In this thesis, I used the Tobii 4C eye tracker ⁴ for both designing attentive information dashboards with VAF and conducting eye tracking studies to analysis users visual behavior. As shown in Figure 2.7, this apparatus is a desktop-mounted eye tracker with the size of $17 \ge 335 \text{ mm} (0.66 \ge 0.6 \ge 13.1 \text{ in})$ and the sampling rate of 90 Hz. For applying this apparatus for research purposes, I used the relevant license to record and analyze the eye movement data.

Also, different metrics are used to classify the collected data through eye trackers. Holmqvist et al. (2011) and Duchowski (2017) have provided a long list of eye tracking data used in previous eye tracking studies. There are two fundamental concepts regarding eye movement data. First is the fixation, which is known as maintaining the visual gaze on a single location. During fixation, the gaze stops relatively on one object to process it. For that, a fixation needs to be around 200-300 milliseconds long. "Approximately 90% of viewing time is spent in fixations" (Duchowski, 2017, p.15). The second concept is "Saccades" which are known as the fast eye movements in between two fixations. Saccades can explain a voluntary change in the focus of attention (Duchowski, 2017). Besides eye movements, the pupils can give insights about mental processes. The pupil diameter is used to measure the

⁴https://gaming.tobii.com/tobii-eye-tracker-4c/



Figure 2.7.: Tobii 4C eye tracker with the sampling rate of 90 Hz.

user's cognitive load, which is higher for a bigger dilation (Holmqvist et al., 2011, p.393). Pupil dilation can also be a measurement to indicate a user's interest; a larger diameter leads to higher interest. Furthermore, there are several techniques to visualize the eye tracking collected data to understand users' visual behavior. Blascheck et al. (2014) have presented an overview of visualization techniques for eye tracking data and have described their functionality in nine categories.

There are several eye tracking applications that process and integrate eye tracking data. In general, Duchowski (2002) has broadly categorized eye tracking applications into the classes of diagnostic and interactive. In the diagnostic applications, researchers use the eye tracker to provide quantitative evidence of the user's overt attentional processes. Diagnostic eye tracking applications collect user's eye movement data while doing a task, and later, researchers use off-line records for the evaluation. The second category is interactive eye tracking applications that interact with the user based on observed eye movement data. In this case, the system uses the user's eve movement data in real-time and enable eve-based interactions. Furthermore, Majaranta and Bulling (2014) have categorized gaze interaction applications into four categories. Figure 2.8 depicts these applications based on the continuum from real-time to off-line recordings of eye movement data. The first category is "Explicit Eye Input" that refers to the applications which use gaze-based command and control. Different input possibilities have been tested and often compared to input through mouse and keyboard (Jacob and Karn, 2003) such as gaze typing (Majaranta and Räihä, 2002), gaze input (Hutchinson et al., 1989) and interaction with mobile devices (Drewes et al., 2007; Dybdal et al., 2012). Zhai (2003) presented an overview about using eye movements as an input interaction. Although using eyes as input is fast and useful in different cases, humans might look at a specific element for processing information and not intended to give a command. Jacob (1991) has introduced this problem as "Midas Touch" effect. The solution for that can be using other input modalities, such as gesture or voice, besides eye input. The second category is "Attentive User Interface" in which the users' eye movement data is used subtly in the background to provide attention support

features. Eye trackers have the capability to collect eye movement data in real-time and use it for designing AUIs (Bulling, 2016; Henderson et al., 2013; Majaranta and Bulling, 2014; Roda and Thomas, 2006; Vertegaal, 2003). In this thesis, I focus on this type of eye tracking application and present a more detailed description of that in the next session. The third category is "Gaze-based User Modeling" systems in which tracking users' eye movement data provides a way for understanding the users' behavior, cognitive process, or intention. Researchers used eye trackers to identify user's confidence (Smith et al., 2018), intention (Doshi and Trivedi, 2009), workload (Bailey and Iqbal, 2008), engagement (Ishii et al., 2013), task (Steichen et al., 2014), learning curve (Lallé et al., 2015), personality traits (Hoppe et al., 2018), etc. The fourth category is "Diagnostic Applications" that works with passive eye monitoring and has a similar goal of diagnostic applications introduced by Duchowski (2002).



Figure 2.8.: Categories of eye tracking applications by Majaranta and Bulling (2014).

2.3.2. Attentive User Interfaces

The research on systems sensitive to the users' attention arose from the idea that a massive amount of the information surrounds the users in these years while their attention is a limited resource (Anderson et al., 2018; Bulling, 2016). Lack of proper attention allocation can cause different errors, which are known as attentional breakdowns (Roda, 2011). For example, the failure to notice a fully visible but unexpected content in UIs can be explained by the phenomenon named inattentional blindness (Mack and Rock, 1998), or the inability to detect changes because of lack of proper attention allocation as change blindness (Simons and Rensink, 2005). The developed conceptual framework as the result of Study I shows that attention support systems aim to assist users in overcoming such attentional breakdowns by adapting the system without involving users (e.g., (Buscher et al., 2012)) or providing feedback to increase their awareness (e.g., (D'Angelo and Gergle, 2018)). To assist in attention management, systems use user attention indicators such as user presence, proximity, orientation, speech activity, or gaze (Vertegaal, 2003) as an input and provide attention support feature.

This type of system is known as AUI and Vertegaal (2003, p.32) has described it as "Computing interfaces that are sensitive to the user's attention". Later, Roda and Thomas (2006, p.577) have introduced attention-aware systems as another type of attention support system as "systems capable of supporting human attentional processes". They argued that such a system is helpful in different domains, mainly where:

"...(1) attentional switches are very often solicited, or (2) where the users' lack of experience with the environment makes it harder for them to select the appropriate attentional focus, or (3) where an inappropriate selection of attentional focus may cause serious damage to the system, its users, or third parties, or (4) where the very reason for the system to exist is to attract the user attention".

Also recently, Bulling (2016) has suggested using real-time eye movement data beyond specific situations and proposed pervasive AUI. This type of system aims to continuously track the user's eyes and manage their attention for daily life activities. Such systems can simultaneously optimize for both information throughput and subtlety. Future AUI will support users' attention management process in daily life (Bulling, 2016).

In this thesis, I focus on integrating attention support systems while users work with IS applications at workplaces. Therefore, the definitions of AUI and attention-aware systems fit my intention. As both AUI and the attention-aware system have close meaning (Roda and Thomas, 2006) and have the same goal, I choose AUI as the key terminology in my thesis and refers to attentive information dashboard as a type of AUI.

Furthermore, users' eye movements can be used as input for designing intelligent UIs (Henderson et al., 2013), and their usage has grown during the last years (Nakano et al., 2016). This data is known as one of the popular data sources for designing AUIs (Bulling, 2016; Majaranta and Bulling, 2014). Such AUIs are used in different fields so far. Examples of AUIs are: reading assistant by attentive documents (Buscher et al., 2012), attentive information systems (Maglio et al., 2000), attentive recommender systems (Xu et al., 2008a), attentive tutoring systems (D'Mello et al., 2012), support resuming interrupted tasks (Kern et al., 2010; Mariakakis et al., 2015), attentive conversational agents (Ishii et al., 2013). The results from a SLR study as Study I in this thesis provides an overview of existing systems.

For IS application, Maglio et al. (2000) have suggested in integrating users' eye movement data for designing attentive IS. This type of IS application should gather evidence about users' behavior from multiple sources, model the user, and present extra information to support their task. Furthermore, they suggested using users' gaze information as one source of information for attentive IS. Based on the developed conceptual framework in Chapter 3, this type of support is considered a system adaptation focus. However, the best of my knowledge there is no attentive IS focusing on user adaptation and increasing users' self-awareness.

2.3.3. Business Intelligence and Analytics Systems

With the use of advanced information technologies, companies have the opportunity to make informed decisions and take faster actions by using data resources (Phillips-Wren et al., 2015; Torres et al., 2018). The usage of Business Intelligence and Analytics (BI&A) systems has become more important for companies over the last years (Peters et al., 2016; Trieu, 2017).

BI&A systems have gone through its evolution during the past decades. It is not an entirely new phenomenon and can be traced back to the middle of the 20th century (Watson, 2009). In the 1960s, organizations used computers for transaction processing and scientific applications. The focus did not lie on decision support back then and later, the first system to support making decisions was developed to help managers achieve specific business goals. Over the years, various applications for this purpose have been developed and named differently, like Executive Information Systems (EIS), Management Information Systems (MIS), or Decision Support Systems (DSS). In the early 1990s, the term Business Intelligence (BI) was first introduced by Howard Dresner (Power, 2007), who has described it as "an umbrella term for all decision support applications" (Wixom et al., 2011, p.13). The popularization of the BI term in the 1990s was reinforced by the development of IT infrastructures like the internet and large database systems enabling data warehousing (Dinter et al., 2015). In the beginning, the focus of BI was mostly on technological aspects resulting from new developments in the application of analytic systems like Online Analytical Processing and Data Mining (Chaudhuri and Dayal, 1997; Fayyad et al., 1996). Later, other BI specific topics evolved like the integration of mass data with the help of extract, transform, and load processes, the design of different architectures of data warehouse systems, the modeling of the query-oriented data schema, as well as the pursuit of improvement and assurance of data quality (Wang and Strong, 1996). At the beginning of the 21st century, researchers and practitioners shifted their focus from the rather technical aspects to more strategic, organizational, and integrative dimensions of BI. Also, there was a change of focus within the technical aspects like the collection of web-based, unstructured content in social media, content and text analytics, etc. It offered new opportunities for businesses to analyze their customers' opinions, analyze mobile and sensor-based content, etc. More recently, with billions of people carrying smartphones and other GPS devices, location-aware analysis, and person-centered analysis emerge in the context of mobile BI and the internet of things as well (Chen et al., 2012).

Chen et al. (2012) have introduced the BI&A concept as the next version of BI. Whether they are called DSS, MIS, EIS, BI, or BI&A the purpose of all of these systems is the same, namely, to provide managers with information to support their decisions and the terms are often treated synonymous (Chen et al., 2012). Due to the opportunities that are collecting and analyzing data provides, the interest in BI&A is significant these days, and I consider BI&A following the definition have presented by Chen et al. (2012) as: "... the techniques, technologies, systems, practices, methodologies, and applications that analyze critical business data to help an enterprise better understand its business and market and make timely business decisions".

2.3.4. Information Dashboards

To support decision-makers, BI&A systems are generating insights by using analytical techniques and then envisioning them in the form of visualized information. The primary purpose of information visualization is to support users in perceiving patterns, which can be used to build appropriate explanatory models and support improving performance (Purchase et al., 2008). Also, information visualization ease in finding the relationships among data and provide a comprehensive overview of the data (Kelton et al., 2010).

Information dashboards are known as one of the most effective BI&A tools (Negash and Gray, 2008) that show the current status of metrics and key performance indicators of an entire company or sub-division (Pauwels et al., 2009). An information dashboard typically combines numbers, metrics, charts, graphs, etc. on one screen, and users need to explore them at the same time and find relationships between them before making decisions. The information dashboards can be customized for specific roles and show metrics that are optimized for a particular business task. Dashboards usually process data from different sources in real-time and include design features that support users to customize the layout based on their intention. The users of dashboards may have different purposes include consistency, monitoring, planning, and communication (Yigitbasioglu and Velcu, 2012). By using them, the user may search for information or scan the whole dashboard to understand the current business state (Vandenbosch and Huff, 1997). Few (2006, p.34) has defined information dashboards as:

"... a visual display of the most important information needed to achieve one or more objectives; consolidated and arranged on a single screen so the information can be monitored at a glance".

Furthermore, Yigitbasioglu and Velcu (2012) have described an information dashboard for management purposes as:

"collect, summarize, and present information from multiple sources such as legacy, enterprise resource planning, and BI software so that the user can see at once how various performance indicators [...] are performing".

The proper usage of information dashboards and matching it to the decision-makers' tasks are known to affect the success of such systems (Schwarz et al., 2014). Also, having interactive information dashboards are considered as one of the critical elements for BI&A systems (Cindi et al., 2019; Pauwels et al., 2009; Yigitbasioglu and Velcu, 2012). Information dashboards support decision-makers in their thinking process (Bačić and Fadlalla, 2016). Therefore, they should be designed to help decision-makers maintain their cognitive tasks in convenient ways. To reach that goal, they need to be evaluated according to their design features and the way the users interact with them for making decisions. They can also be equipped for new features and support users in conducting their cognitive (Niu et al., 2013; Yigitbasioglu and Velcu, 2012).

The interaction between decision-makers and information dashboards is mainly focused on investigating the information visualization (Tegarden, 1999). In existing dashboard studies, researchers have used eye trackers to evaluate the design features (e.g., presentation formats, colors, size, etc.) by analyzing off-line records of eye movement data (Bera, 2014,1; Burch et al., 2011). Besides that, they investigated the visual analytics strategies of decision-makers to find the relationship between the accuracy, speed, and consistency of decisions (Cöltekin et al., 2010; Vila and Gomez, 2016), user's cognitive effort while working with visualized information (Fehrenbacher and Djamasbi, 2017; Smerecnik et al., 2010), and their situation awareness (Nadj et al., 2020). Researchers also used eye movement data to examine the relationship between user characteristics and visualized information such as perceptual speed, visual and verbal working memory (Okan et al., 2016; Toker et al., 2013). The main focus is on integrating this technology for evaluation purposes (Kurzhals et al., 2016). Silva et al. (2019) have argued that only a few studies investigated using eye movement data in real-time and they have focused on system adaptation (e.g., (Okoe et al., 2014; Shao et al., 2017; Silva et al., 2018)) rather than increasing users awareness. The results from the SLR study in Chapter 3 shows that no study has used real-time eye movement data for designing attentive information dashboards that provide attention support features.

3. State-Of-the-Art and Conceptual Framework ¹

3.1. Study I: Overview

Researchers have studied the collection and usage of user's gaze data in two major modes, the off-line and real-time mode, (Majaranta and Bulling, 2014). In the off-line mode, collected gaze data is analyzed after a task has been performed. Typically, the off-line mode is used for diagnostic purposes and has been the most well-known usage scenario of eye trackers. Using eye trackers for diagnostic purposes is popular among researchers since it provides objective data about users, which is enhancing subjective self-reported data, e.g., collected using surveys. Researchers have intensively used eye trackers for the off-line mode in various fields. They have integrated highly accurate and expensive eye trackers to explore user visual behavior while interacting in digital or physical environments and model user cognitive states while conducting tasks (Hayhoe and Ballard, 2005; Kowler, 2011; Liversedge and Findlay, 2000; Orquin and Mueller Loose, 2013; Rayner, 1998). With the further advancement of eye tracking technology during the last years, researchers have increasingly leveraged the possibility of collecting and processing user eye movement data in *real-time*. With the real-time mode researchers can involve the users' eye movement data in an interactive system and create gaze-responsive applications (Duchowski, 2002). These applications offer new ways of interacting with systems, including two main paradigms: 1) using eyes as an input device for interactive systems or 2) collecting and analyzing user eve movement data in real-time to design Interactive Intelligent Systems (IIS). Integrating user gaze data as input for designing IIS has grown during the last years (Nakano et al., 2016). Besides attention detection as the primary purpose of tracking user eye movement data (Carrasco, 2011), the usage of it is suggested to extend user context with cognitive dimensions (Bulling and Gellersen, 2010) including user's confidence (Smith et al., 2018), intention (Doshi and Trivedi, 2009), workload (Bailey and Iqbal, 2008), engagement (Ishii et al., 2013), task (Steichen et al., 2014), learning curve (Lallé et al., 2015), personality traits (Hoppe et al., 2018). In this state-of-the-art study, I consider systems that process eve movement data in real-time to design IIS as eve-based IIS.

The research gaps regarding this study are already discussed in Section 1.2. In this study, I summarize existing knowledge about eye-based IIS by conducting a SLR study and integrating the results in a conceptual framework. Furthermore, I provide a list of future research directions for eye-based IIS. Best of my knowledge, such review does not exist, and current literature reviews focus on summarizing the usage of eye trackers in specific fields rather than integration for designing IIS. Therefore, I focus on closing this gap by answering the first RQ of this thesis as following:

RQ1: What is the state-of-the-art and potential future research directions for designing eye-based IIS?

¹This Chapter is based on the following working paper: Toreini and Maedche (2020)

This RQ is the step before conducting the DSR project in this thesis and has the aim to provide an overview of existing knowledge in eye-based IIS to support designing attentive information dashboards in the next steps. To answer this RQ, I first introduce the foundations of eye-based IIS in Section 3.2 with an integrated conceptual framework, including key dimensions for designing such systems. Subsequently, I introduce the methodology used to conduct SLR study in Section 3.3. Next, in Section 3.4, I present the findings for each dimension of the developed conceptual framework in detail. Last, Section 3.5 contains discussion and suggestions for future research direction in this field before concluding this study in Section 3.6.

3.2. Conceptual Foundations

In this section, I first introduce the concept of eye-based IIS and define it. Subsequently, I describe the integrated conceptual framework which is derived in this study and is used to report the state-of-the-art review and future research directions.

3.2.1. Eye-based Interactive Intelligent Systems

Jameson and Riedl (2011) introduces IIS as intelligent systems that people interact with, and they emphasize the interplay of intelligent technology with advanced interactions of users a system. By applying intelligent technology, a system embodies capabilities that have traditionally been associated with humans. An example of that is the ability to perceive information, learn, reason and plan. Complementary, the focus on advancing interaction design for intelligent systems emphasizes the need to understand users better and support them when interacting with systems. Thus, IIS combines intelligent algorithms and advanced interactions to promote the interaction between users and the system.

The usage of user gaze data leveraging eye tracking technology as input for designing intelligent systems has grown during the last years (Henderson et al., 2013). The primary usage of the eye movement data in real-time is to track user's attention (Carrasco, 2011). Based on that, Vertegaal (2003) introduces the concept of Attentive User Interface (AUI) as computer interfaces that are sensitive to the user attention mainly by tracking user eyes and model the user's visual behavior. Since an AUI knows the status of user attention, it can raise the interaction between users and the system accordingly by either providing attention support features or adapt the system best on user status. Nguyen et al. (2018) provided a comprehensive review of attentive systems that use saliency cues. Later, Roda and Thomas (2006) introduced attention-aware systems, which has a similar goal to AUIs and are capable of supporting human attentional processes. Similar to AUIs, eye trackers are proposed as the primary device for designing such a system as well. Recently Bulling (2016) has suggested pervasive AUIs and usage of real-time eye movement data constantly. This type of system aims to track the user's eyes continuously and manage their attention for daily life activities. Such systems can simultaneously optimize for both information throughput and subtlety. Attention status of the users is the primary usage of real-time eye movement data and designing AUIs, attention-aware systems or pervasive AUIs are suggested based on that. However, these are not the only use cases for designing intelligent

31

UIs based on real-time eye-movement data. Analyzing eye movements can be used to derive other cognitive dimensions beyond attention and design cognition-aware interfaces (Bulling and Gellersen, 2010; Bulling et al., 2011). Eye trackers are also effective to understand user confidence (Smith et al., 2018), intention (Doshi and Trivedi, 2009), workload (Bailey and Iqbal, 2008), engagement (Ishii et al., 2013), task (Steichen et al., 2014), learning curve (Lallé et al., 2015), personality traits (Hoppe et al., 2018). The systems that used such information as input are called in different ways such as gaze-aware (Tremblay et al., 2018), gaze-reactive (D'Mello et al., 2012), gaze-contingent display (Duchowski et al., 2004) and gaze-directed display (Toet, 2006). A common characteristic of all eye-based systems is that eye tracking technology is the core device for collecting user gaze data. Later, the collected data is used to model user states and on this basis to design advanced interactions for its users. Therefore, I name IIS which leverages eye trackers as eye-based IIS. I assume that this class of systems does not only focus on attention but also covers other user states. Specifically, I define eye-based IIS as follows:

"Eye-based interactive intelligent systems are intelligent systems that use eye tracking technology to collect user gaze data to model user states for designing advanced interactions between users and the system".

3.2.2. Integrated Conceptual Framework

I propose a conceptual framework for research on eye-based IIS after checking papers collected in this study, as depicted in Figure 3.1. In this framework, I distinguish three high-level categories in framework 1) influencing factors, 2) eye-based IIS properties, and 3) outcomes. Each category includes several specific elements. In the following sections, I introduce the conceptual foundations of each category, as well as their dimensions in detail. The proposed framework provides a foundation for capturing the current state of research and future research directions in eye-based IIS.



Figure 3.1.: Integrated conceptual framework for eye-based IIS.

Influencing Factors

The influencing factor category focuses on elements that impact the design of the eye-based IIS and its outcomes. Besides that, I concentrate on eye tracking technology as there are different types to be used for designing eye-based IIS. I also investigate how the developed systems are evaluated in which conditions.

As the first dimension in this category I consider **Context**. The term context has been used in many ways in different research fields. Chen and Kotz (2000) have defined it as *"the set of environmental states and settings that either determines an application's behavior or in which an application event occurs and is interesting to the user"*. Schilit et al. (1994) have named these environmental states as physical context and added user conditions, including the user profile, social situation, location, etc. as another aspect of the designing context-aware systems. For designing eye-based IIS, I consider both the user and physical contexts and integrate it into the framework.

Beside context, I consider **Task** as the second dimension in this category. Context and task are known as two dimensions that guide the design of any interactive system (Benyon, 2013). Previous eye tracking studies have shown that the given task affects user eye movements patterns (Yarbus, 1967). Therefore I added task as another influential factor while the primary goal of eye-based IIS is to support users in increasing their task performance.

The third dimension in this category is the *Eye Tracking Technology*. Different types of eye trackers exist in the market and have been used by researchers in the last years. Researchers in the field of eye-based IIS need to investigate which eye tracker fits their use case by considering different aspects such as usability, accuracy, and price. Modern eye trackers use mainly near-infrared technology as well as a high-resolution camera to track the gaze direction. Next to the camera emits infrared light, which produces a reflection on the cornea of the eye that its position changes depending on where the user looks (Morimoto and Mimica, 2005). Therefore, the point of gaze can be calculated by measuring the changing distance between the pupil and the cornea reflection (Majaranta and Bulling, 2014). Independent from the technical aspects to record the gaze position, eye trackers are principally categorized into either desktop-mounted eye trackers (also called remote devices or screen-based) or head-mounted eye trackers (also called mobile eye trackers, eye tracking glasses, etc.). The desktop-mounted eye trackers are used to trace any desktopbased stimulus, while the head-mounted eye trackers are proper for real-life activities or virtual environments. Details of eye tracking technology are discussed in Section 2.3 in Chapter 2.

Finally, I suggest to consider *Experimental Setup* as an important element. It is important to understand how researchers evaluated the designed eye-based IIS. Therefore I consider the experimental setup as an influencing factor. The experimental method includes laboratory or field study, type of study includes within, between, mixed, etc. The amount of participation in the collected studies can impact the conclusions about designed eye-based IIS.

Eye-based IIS Properties

The second category is focusing on the specific eye-based IIS properties, which are considered as the core of the framework. *Sensing and Reasoning* are key properties to design eye-based IIS. Later, these properties are used to design eye-based IIS, which can have two main *Focuses*: making the system more intelligent and support system adaptation or increase user awareness and support user adaptation.

First, eye trackers are the main tool for sensing dimension in designing eye-based IIS. Different metrics are used to classify the collected data through eye trackers. I divided *Eye Tracking Metrics* into five high-level categories, including gaze-based, fixation-based, saccade-based, pupil-based, and others. Each of these categories comprises subcategories that I named as eye-gaze measures. Later, the collected data is analyzed to detect patterns in and draw conclusions on the user state. The reasoning dimension is about how the collected eye movement data is used for *Modeling User State*. Collected eye movement data provides insights into the cognitive process and gives hints about the anticipated user behavior (Majaranta and Bulling, 2014). It is not only finding of their conscious behavior (e.g., relevance, interest, task, intend, etc.) of the users, but it is also acknowledged as a promising approach to obtain insights on unconscious behavior (e.g., boredom, mind wandering, etc.).

In general, IIS should include both human and artificial varieties of intelligence (Jameson and Riedl, 2011) while they aim to create novel interactions that suited to people's abilities. Eve-based IIS systems use the results from the sense and reasoning processes as an input to design such novel interactions. The designed novel interaction has two main focuses: 1) increasing the intelligence of the system by generating implicit feedback and focusing on system adaptation, 2) increasing user awareness by providing corrective feedback, and focusing on user adaptation. First, in the case of **System Adaptation**, eye-based IIS focuses on intelligent technology and provides sense and reasoning results as input for designing better intelligent algorithms that later support users without involving them directly. In this case, the user is observed by the computer, and the system generates implicit feedback. Therefore, it adapts the interface to the user's needs and intentions while the user may not be aware of what is going on in the back end and the changes in the provided information. However, adapting the interfaces that users are working with can help them to use the system more efficiently. For example, Buscher et al. (2012) used the user's eye movement data as implicit feedback and designing attentive documents. Such documents are used for personalizing and improving the quality of web search. As another example, Rozado et al. (2015) integrated user gaze information as a source for user intention and use it to speed up web navigation and increase user comfort.

The second focus of eye-based IIS is on designing new interaction in which the system focuses on *User Adaptation*. In this case, users are involved in advanced interactions and receive corrective feedback while conducting the task. Here, the corrective feedback role is to increase users' awareness and let them improve their information processing. To design corrective feedback, the computer adjusts the eye tracking data as gaze augmentation in real-time or present user eye movements history in forms of different visualization

techniques such a heatmap, scanpath, last fixated point, etc. For example, D'Angelo and Gergle (2018) investigated how different forms of gaze visualization increase shared attention and therefore influence collaborative performance.

Outcomes

In the last category, I investigate the outcomes of the designed eye-based IIS through three dimensions. The first dimension focuses on *User Perception* in which I study how users clarified their experience while working with the eye-based IIS. User perception is usually measured by conducting a survey or interviews after experiencing a new innovative interaction. As the second dimension, I investigate how their **Behavior** is changed while working with eye-based IIS. Here, I focus on objective data collected during user interaction through different approaches, including data from user mouse and keyboards or by integrating other user tracking devices such as eve trackers. In the eye-based IIS studies, since the eye trackers are already used in the design process, I assume that researchers use the device for the evaluation sections as well and report users' visual behavior. As the third dimension, I consider the *Performance* measurements, which can have two perspectives. First, studies may review how using the eye-based IIS influences users' performance to conduct their tasks. This examination focuses more on the relationship between user performance and the provided innovative interaction by eye-based IIS. From the second perspective, studies may focus on the system's performance and consider the effects of the intelligent technology used in designing eve-based IIS.

3.3. Methodology

For executing the review, I followed the steps proposed by Kitchenham and Charters (2007). Based on that, the review process includes three main steps that include planning, conducting, and reporting. Each of these steps entails specific sub-activities. Figure 3.2 explains the three high levels and the subsections for each step and the activities. In the following, I describe each step for the plan and conduct steps in detail. Later I provide the findings in Section 3.4.

3.3.1. Planing the Review

In the planning, first, the need for the review must be identified. I already discussed the existing research gap and motivation for this review in Section 3.1. In the second part, I developed a conceptual framework in several iterations while exploring collected papers for this study. Later, this framework supports analyzing the collected papers in the review from different perspectives. The detail of this framework is already discussed in Section 3.2. Last, I developed a review protocol that covers the following aspects.

Sub-Research Questions: To follow a structured approach in selecting and later analyzing the collected papers, I articulated sub-research questions: (1) What are the boundary conditions of using eye trackers in designing innovative interactions? (2) How are innovative interactions conceptualized? (3) Which influencing factors are considered while



Figure 3.2.: Steps of the SLR study for designing eye-based IIS following Kitchenham and Charters (2007).

designing eye-based IIS? (4) Which eye-based IIS properties are used for designing eyebased IIS? (5) What are the outcomes of the eye-based IIS?

Search Strategy: Later, I developed a review protocol, which is supposed to be used as documentation for other researchers to re-run the review and get the same results. Once the aim of the study and the review protocol is defined, I generated a boolean string that includes keywords. These keywords are extracted after searching the ACM Intelligent User Interface conference and ACM Transaction of Interactive Intelligent Systems as two HCI outlets relevant to the eye-based IIS topic. I searched for "eye OR gaze" in these outlets, identified some relevant papers, and extracted a set of keywords from them. I categorized these keywords into four groups, as can be seen in Figure 3.3. The first two categories (eye and movement) are related to eye tracking technology and focus on finding the papers that include these devices. The other two categories cover the aspect focus on the domain of IIS. Later, I used the connecting operators in the way to search for each keyword-combination that represent both perspectives and support to extract eye-based IIS studies.

Subsequently, I selected six databases covering different research communities since eye tracking technologies have already been used by researchers in various fields: ACM Digital Library, ScienceDirect, IEEE Explore, Wiley, EBSCOhost, and WebOfScience. I used the search string to search on the title, abstract, and keywords section of the papers. I also chose a time limitation from beginning 2008 until the end of 2019, which covers the last 12 years for the initial search. In the backward search, I considered the same time limitation to provide a complete overview.



Figure 3.3.: Search string with four main categories to conduct the SLR study.

Study Selection Criteria: After collecting the papers, I applied several selection criteria only to include high-quality and relevant studies. First, I included only peer-reviewed publications. Second, I excluded research in progress papers. Third, I removed publications with less than five pages. Fourth, I excluded papers that used eye tracking only for diagnostic purposes. Fifth, I excluded papers that concentrated on increasing the quality of life for users with physical disabilities or cognitive impairments. Sixth, I excluded papers that apply eyes as an input device to directly control the system or use it in the multi-modal interaction set up. Seventh, I removed the papers that focus on modeling user behavior, cognitive processes, intentions, etc. without providing any innovative interactive system. With those selection criteria, I focused on the studies that use real-time eye movement data to model user state and later provide innovative interaction design as used in the definition for eye-based IIS in Section 3.2.

3.3.2. Conducting the Review

In this step, I first executed the search string in the six selected databases and found 2706 papers. Later I filtered relevant papers by reading their title and abstract and considering the pre-defined questions discussed in Section 3.3.1 and selection criteria in Section 3.3.1. The result of this step ends with 139 selected papers. Next, I read these papers in more detail and selected 22 of them that fit the definition of eye-based IIS and the goal of this review. Next, I conducted backward and forward search on these papers and added 25 more papers to the list. In the end, I selected 47 papers that are considered as eye-based IIS and published from the beginning of 2008 until the end of 2019. The list of selected papers can be seen in Appendix A.1.

A descriptive analysis of the final list of collected papers shows that the dominant research community for designing eye-based IIS is HCI by covering 81% of all papers. From the remaining 19% papers, 6% are in psychology and cognition community, 4% are published in the learning community, 4% in the visual computing community, 2% in multimedia, and 2% in system engineering. Looking in more detail into the HCI papers, 57% are published in top HCI outlets ranked by Google scholar ² as well as ETRA conference as one of the important conferences for the eye tracking community. 26% of the papers are published

²https://scholar.google.es/citations?view_op=top_venues&hl=en&vq=eng _humancomputerinteraction

in CHI conferences and 31% in other outlets such as IEEE Transactions on Interactive Intelligent Systems, International Journal of Human-Computer Studies, IUI, UbiComp, ICMI, ETRA.

As depicted in Figure 3.4, one can clearly recognize an upward trend towards more research in the field of eye-based IIS nowadays. 60% of the papers were published in the past six years (2014-2019), while 40% of them are from 2008 to 2013. The reason for an increasing number of studies in this field during the last years may be the improvement of eye tracking technology as well as the availability of cheaper devices in the market. More details on these factors are discussed in the eye tracking technology dimension in Section 3.4.1.



Figure 3.4.: Distribution of collected papers based on years and research fields.

3.4. Findings

I analyzed all 47 resulting papers using coding tables derived based on the developed integrated conceptual framework explained in 3.2. The detail of these coding tables can be seen in Appendix A.2. The results from analyzing the coding tables are used to report the state-of-the-art that I present in the following.

3.4.1. Influencing Factors

Context

Figure 3.5 summarizes the findings regarding the user context. Analyzing the collected studies show the individual interaction between users and the eye-based IIS is the dominant research focus regarding the user social context. 68% of the studies focused on the individual aspect, 30% focused on team-based and 2% on the collaboration between users and conversational agents. Furthermore, the participants' information reported in the conducted studies includes age (68%), gender (62%), experience level (45%), and vision status (19%) are the four metrics that are considered in the studies as the user profile. This distribution shows that the user age and gender are the two popular factors for profiling users. These four factors are the ones that researchers collected while evaluating



Reported User Context about Participants

Figure 3.5.: Descriptive statistics: user context.

the designed eve-based IIS. Besides, user profiling can also be viewed as the user state collected by the eye trackers, which I discuss in Section 3.4.2.

Figure 3.6 presents the results regarding the physical context. First, I classified the physical context into digital and real-world environments. As can be seen, 96% of the collected studies focus on the eye-based IIS that works with digital environments, while only 4% consider designing such a system for real-world physical interactions. Furthermore, I analyzed the type of application that integrated the eye tracking system for designing eye-based IIS as another dimension for physical context. Figure 3.6 shows that collaboration tools are the most popular digital applications that are empowered with eye-based IIS (22%). Collaboration is the mutual work of two or more people either at the same time or consecutively. Lack of joint attention during communication can result in various problems. Therefore, eye-based IIS systems in this category typically focus on enhancing the collaboration between team members by providing a shared gaze feature. The team's participants focus on a common goal, such as writing a text, problem-solving, driving, learning, etc. that are discussed as the list of tasks in the next sections. Next, web-based applications (15%)are the popular physical context for designing eye-based IIS. In this case, the system integrated user eye movement data in real-time and extracted their state while exploring a website. As an example, such information is used to improve the performance of recommender systems in an online shop and provide more accurate items to the web users based on their interest collected through their eyes. The next popular digital application is exploration tools (13%), where users need to process huge amounts of information on the provided UI. In this case, eye-based IIS helps users manage information overload properly and prevent possible attentional breakdowns, such as missing important information or change blindness. For the next digital application, researchers focused on enhancing the effectiveness of training applications (13%) and support the user learning process by tracking their eye movement data in real-time. The collected information is used for adapting the learning content, inform the students about their learning process, inform the teacher about the status of their students, etc. Also, Reading tools (11%) that support the user in reading books, pdfs, texts, etc. as well as the driving simulators (11%), are popular among eye-based IIS researchers. Finally, gaming tools (7%), multi-display applications (5%), VR and AR platforms (2%), and cellphone apps (2%) are less interesting digital applications.



Figure 3.6.: Descriptive statistics: physical context.

Task

Figure 3.7 displays a list of tasks and distribution of them investigated in the collected papers. As can be seen, browsing and search tasks have the highest rate, with 45%of all tasks. This includes browsing or searching for a target on a digital platform or real-world environment. The reason for the high investigation of this task can be the user difficulties in managing limited attentional resources and facing different attentional breakdowns. Besides browsing and search tasks, researchers considered reading (15%), driving (11%), learning (11%), and monitoring (6%) tasks. I assume that the need for having proper attention allocation while conducting all those tasks is the main reason to select them in eye-based IIS studies. Examples of such support are informing users about mind-wandering status while reading a text, notice all critical information while driving, concentrating while learning a new language, and avoid missing changes while performing monitoring tasks using multi-display environments. Furthermore, few studies focused on specific tasks such as remote physical tasks (4%), gaming (4%), writing (2%), and programming (3%). These tasks may not be as complex as others, but I found that most of those tasks are combined with collaboration tools and considered as a collaborative task, which increases the overall complexity.



Distribution of Tasks on Eye-based IIS

Figure 3.7.: Descriptive statistics: tasks.

Eye Tracking Technology

Figure 3.8 shows the distribution of apparatus types used in the collected papers. More than one eye tracker type is used in some studies, so I present the numbers in the percentage format. As can be seen, desktop-mounted eye trackers (71%) are more used than headmounted (17%) eye trackers in eye-based IIS studies. I assume that the focus on the digital environment, lower price, easier setup, more manageable data analysis, etc., are some of the reasons for the higher usage of desktop-mounted compared to the head-mounted versions. Besides these two types, 8% of the used eye trackers are webcam-based devices. In this type, researchers used either the existing algorithm that converts webcams to eye tracking (e.g., OpenGazer) or developed a specific algorithm. As another apparatus type, a few studies used VR headsets equipped with eye trackers (2%) or smartphone cameras (2%). The reason for the low number of VR-based eye trackers may be that this technology is still rather new. Moreover, for smartphone cameras, the small displays and low accuracy level of algorithms to change cameras to eye trackers makes the usage of them difficult.





Figure 3.8.: Descriptive statistics: eye tracking devices.

41

Figure 3.9 presents the distribution of used apparatus among existing eye tracking hardware companies in the market and their products. As can be seen, there are 26 different apparatus types used in the collected studies. 23 of them are either the existing infrastructure in the market or used the existing algorithms to convert webcams to an eye tracker; the remaining 3 developed specific technologies. Among the existing eye tracking companies in the market, Tobii is the leading company for designing eye-based IIS. The results show that 53% of all papers used one of the products from Tobii while the other 47% of the apparatus are coming from various companies or are self-made solutions. Also, there are only six apparatus among existing 26 devices that are used more than three times in the collected studies: Tobii T60, Tobii EveX, Tobii X120, Tobii Tx300, Tobii 4C, and EveLink II. Therefore, I assume that these devices are the most popular devices among researchers for designing eye-based IIS. The remaining 20 apparatuses are used less than two times, which reveals that researchers tested different types of eye trackers during last years, and only a few eye trackers could receive more attention for designing eye-based IIS. Also, Tobii 4C eye tracker is the successor of Tobii EyeX; both are considered as low-cost eye trackers designed for eye-based interaction purposes. Researchers used these devices eight times in total during the last years, and I assume that they are more popular apparatus for designing eye-based IIS than the others.



Figure 3.9.: Descriptive statistics: eye tracking hardware providers.

In previous eye tracking studies, researchers argued about the high price of apparatus and the difficulty of executing large-scale experiments. Based on the report from Farnsworth (2019), an average eye tracker price is around \$17,500. However, during the last years, the eye tracking hardware companies released cheaper apparatus. I categorized the used apparatus in the collected papers by considering eye trackers lower than 1000\$ as a low-cost eye tracker. This categorization is similar to the categorization from Farnsworth (2019) about the price of eye trackers in the market. In this study, I consider the devices with a higher price than that as high-cost eye trackers. Figure 3.10 reveals the distribution of low and high-cost apparatus as well as their usage in the last years. As can be seen, 43% of collected studies employed low-cost eye trackers, while 57% of the studies used high-cost eye trackers for designing eye-based IIS. However, checking the distribution of

those eye trackers during the last years reveals that the usage of high-cost eye trackers decreased, and more studies with low-cost eye trackers are published. This shows that the tendency of researchers to use low-cost eye trackers for designing eye-based IIS started in recent years, and I assume that it continues in the future. The reason for that can be the improvement in the quality of low-cost eye trackers and their reliability in measuring eye movement metrics in real-time as well as the possibility of running cheaper studies.



Figure 3.10.: Descriptive statistics: eye tracking technology prices.

Experimental Setup

As can be seen in Figure 3.11, most studies (96% of all) focus on the controlled lab environment to evaluate their system and only 4% tested their system as a field study. The reason for that goes back to the general difficulty of running eye tracking experiments as field studies independent in all forms include eye-based IIS, eye-based interaction, or for diagnostic purposes. In the eye tracking field studies, researchers have various challenges such as managing calibration, privacy issues, storing a large amount of eye movement data in long term usage, the difficulty integrating the devices to the real-world environments. These challenges are less while conducting a laboratory experiment in a controlled environment.

Furthermore, by analyzing the type of study design, I observed that more studies focused on a within-subject study type (42%). Conducting eye tracking studies requires much individual preparation that makes the experimental duration long. Also, the eye movements of users depend on several individual characteristics that are difficult to control and make the within-subject experiments more reliable with low amounts of participants. Furthermore, 31% of the studies are conducted as a between-subject design that shows this type is also popular among researchers. Also, 17% of the studies used a single user or group to test their system and did not compare the user or system's performance with any other conditions. Finally, few studies leveraged a mixed design (6%) or case study (4%) research approach. Therefore, I can see that researchers preferred to execute the experiment in a within-subject or between-subject manner rather than any other type while testing eye-based IIS.



Figure 3.11.: Descriptive statistics: experimental design.

Besides, I also extracted the number of participants in the collected studies. The results are depicted in Figure 3.12. I found that the number of participants increased slightly during the last years. The average number of participants from 2008 until 2013 (M=20.68, SD=14.19) was lower than the average number of participants from 2014-2019 (M=34.46, SD=33.29). I assume that with cheaper and more reliable low-cost eye trackers; researchers also had a chance to included more participants in their studies. Having a higher amount of participants support the reliability of their findings.



Number of Participants in Last Years

Figure 3.12.: Descriptive statistics: number of participants.

3.4.2. Eye-based IIS Properties

Eye-based IIS properties are at the core of the integrated conceptual framework explained in Section 3.2. In the following, I first discuss specific aspects for sensing and reasoning dimensions by presenting results from eye tracking metrics and model user stats. Later, I investigate the focus of the designed eye-based IIS system in the collected papers.

Eye Tracking Metrics

In this section, I focus on the eye tracking metrics used in designing eye-based IIS. Figure 3.13 shows the distribution of identified metrics in the collected papers. As discussed in Section 3.2, I identified five categories for eye tracking metrics: fixation-based, gaze-based, saccade-based, pupil-based, and others. Since eye trackers provide several measures simultaneously, some studies used more than one measurement while designing eye-based IIS. Overall, 89 measures (19 unique measures) are used in the collected papers. In order to avoid any confusion when identifying the preferred category, I present the results in percentages.

The most common eye tracking metric in designing eye-based IIS is the fixation-based metric (45%). When users are asked to conduct a task using their visual system, they first move their gaze to the relevant information; later, the visual system of the brain needs to be activated through a longer fixation of gaze to recognize an object properly that is consider as fixation (Majaranta and Bulling, 2014). Fixations are characterized through pauses of at least 100 ms and are, on average, between 200 and 600 ms (Majaranta and Bulling, 2014). Fixation position is the most common measure in this category (15%). Furthermore, the number of fixations (13%) and fixation duration (10%) in an Area Of Interest (AOI) is also measured by researchers in this category. This measurement can be used for different purposes, for example, interpreting more fixations and long duration of fixation on an AOI shows that this AOI is mostly related to deeper processing, more complex, less user-friendly areas, etc. On the opposite side, high stress can result in short fixation duration and expertise, which may prolong the duration (Holmqvist et al., 2011). Furthermore, some studies consider the last fixation points (6%). Recording the last fixation point is needed for TRS systems. In this case, the system records the last point of the users before the interruption and provides support when the user wants to resume the task by showing this position as a reminder for the last processed position. Also, one study used the first fixation point (1%) of users to improve the quality of their recommender system (Cheng et al., 2010).

The gaze-based measures are the second popular eye tracking metric (33%) and focus on the collected gaze data of users while working with eye-based IIS. The most commonly used measure in this metric is the gaze position (16%). In this measure, researchers focused on the position of the user gaze and related it to the point of relevance, interest, importance, etc. This can be seen as a fixation position; however, since researchers did not mention it specifically as a fixation point, I consider the gaze position as a separate measurement for the evaluation. The second measure is gaze duration (6%) that delivers insights on the period, the gaze enters an AOI, till the exit. A high value of gaze duration can indicate uncertainty, interest, and difficulties in extracting information, etc. This measure can also be relevant to fixation duration. The next measure is mutual gaze (4%), which is covered in some collaboration scenarios and determines the level of joint attention between partners. Furthermore, researchers investigated the number of gaze transitions (2%) that associates with the movements between two AOIs. Also, in some cases, researchers identified the presence of the users in front of the computer by tracking their gaze presence (2%). Besides, researchers tracked the gaze data to check if an AOI is entered several times and consider it as the user visit counts (1%) within this AOI. One study calculated the 3D focal depth of eye gaze (1%) using online gaze data to calculate when uses attention is engaged in a virtual display space while wearing AR glasses (Toyama et al., 2015).

Besides fixating, users constantly scan the visual environment with the eyes through fast eye movements, called saccades. Saccades are "rapid motions of the eye from one fixation to another" (Holmqvist et al., 2011, p.23) and saccade-based measures shall "indicate the quality of visual cues in the stimulus or extent of visual searching" (Kurzhals et al., 2016). This metric covers 8% of all studies and is less popular than fixation and gaze-based metrics. However, it offers the possibility to identify difficulties with the encoding of visuals by measuring a number of the saccade (3%) or lack of engagement by saccade length (2%). Also, they provide insights into the mental workload (Holmqvist et al., 2011, p.405). Furthermore, the direction of the saccade (2%) can support the system for identifying the user intention (e.g., switch line while reading when the saccade is from top to down)

The next metric is the pupil-based that only covers 8% of the collected studies. The reason for that can be the pupil's sensibility to the light and distance, which makes it challenging to use it out of a controlled environment. Researchers that used this metric consider two measurements include pupil dilation (7%) and dilation speed (1%) as two measures. The pupil diameter is used to measure the user's cognitive load, which is higher for a bigger dilation (Holmqvist et al., 2011, p.393). Pupil dilation can also be a measurement to indicate a user's interest; a larger diameter leads to greater interest. Apart from this, other external factors as diabetes, ages, pain, drugs, or emotion tend to influence the pupil size as well. Also, Vrochidis et al. (2011) used the speed of dilation as the rate of changes in time and considered it as an indicator of interest.

Besides the mentioned category, some papers used the scanpath measure (3%), which combines fixations and saccades of the users. It explains the sequential order of the fixations on the underlying stimulus. Following the path can be interesting to discover the user interactions with the interface. The scanpath is useful when the research concerns the time until an area is first entered. Furthermore, it explains difficulties in handling the task by the user. Beside scanpath, 3% of the studies extracted several eye relevant features and used a mixture of those to model user behavior. This approach is called as "global eye feature" by D'Mello et al. (2017). In this study, the authors used 62 global eye features to identify the mind-wandering status of the users while reading a text.

Analyzing the results shows that considering gaze and fixation based metrics is the most common metric (78% in total). Regarding the measures, gaze and fixation positions were popular among researchers rather than any other measure (31% in total).



Eye Gaze Measures in Eye-based IIS Studies

Figure 3.13.: Descriptive statistics: eye tracking measures.

Reasoning

Figure 3.14 shows the list of user states identified through collected eye movement data. Researchers mainly detected user distribution of attention in a given context and task that is a proxy for relevance or interestingness for the user. Therefore I can see that most eyebased IIS studies are in the direction of AUIs, attention-aware systems, and pervasive AUIs as discussed in Section 3.2.1. The second popular usage is the detection of attention shifts (13%). This identification used to support users when they face an interruption or need to shift their attention in multi-display environments. Next, identification of the engagement (8%) and mind-wandering (5%) are investigated with some eye-based IIS. Last, only a few studies focused on intention 3%, uncertainty 2%, and confidence 2%.



Modelling Visual Behavior based on Eye Tracking Metrics

Figure 3.14.: Descriptive statistics: modeling user states.

Eye-based IIS Focus

Figure 3.15 shows the distribution of eye-based IIS focus among the collected papers. As can be seen, I found that the papers are almost equally distributed among system and user adaptation focuses. 51% of the collected papers focused on system adaptation by providing implicit feedback to the system, while 47% of them focus on user adaptation by increasing user awareness and providing corrective feedback. Furthermore, one study (2%) covered both focuses by providing different types of innovative interactions using user's real-time eye movement data. This shows that the research stream on both directions is receiving attention by researchers in the field of eye-based IIS and is properly balanced. Furthermore, I analyzed these two focuses based on years of publication. I found that in the first years, researchers focused more on system adaptation. However, during the last years, focusing on using user's eye movement data for user adaptation is increased.



Figure 3.15.: Descriptive statistics: focus of eye-based IIS.

3.4.3. Outcome

Perception Outcomes

By analyzing the collected papers, I found that only 60% of the studies collected user perceptions, and 40% ignored it. Researchers used survey questions or interviews to collect user opinions after they experienced the designed eye-based IIS. As a result of this, I identified 12 different perception constructs and measurements leveraged in the collected studies and present them in Figure 3.16. The findings show that perceived usefulness (33%) is the most popular construct to measure while analyzing user perceptions. Furthermore, researchers collected perceived satisfaction (17%) of the users. These two constructs cover overall 50% of all collected measures. The high frequency of using these constructs shows the high importance level of them for the eye-bases IIS designers. The remaining 50% comprises ten other constructs. Some researchers tried to capture the mental workload (10%) of the users while working with eye-based IIS by conducting the NASA-TLX test, asking survey questions, or conducting an interview. Besides, researchers that focused on designed collaborative systems with eye trackers asked participants about the collaboration quality (10%). Also, researchers asked about distracting levels (8%) of the provided feature with eye-based IIS. Some researchers collected ease of use of the system (6%). Also, 4% of the studies checked whether the tracked eye movement data was correct or not based on what users remember about their attention distribution. The measures like confidence (2%), frustration (2%), challenge (2%), fun (2%), and nervousness (2%) were also integrated into some studies.



Figure 3.16.: Descriptive statistics: user perceived outcomes.

Behavioral Outcomes

By investigating the collected papers, I found that 57% of the collected papers include the user behavior in their evaluation section, and the remaining papers ignored it. This distribution is almost similar to the perception outcomes. However, only 36% of all papers both measure user perception and behavior. 21% only did the behavior and did not include any perception measures. 23% focused only on the perception of the users and ignored analyzing the behavior, while 19% did not include any of them. As can be seen in Figure 3.17, I identified 12 specific constructs for measuring user behavior. Since the eye trackers are used in designing eye-based IIS, it is dominant for tracking user behavior with 82% of all constructs. Researchers used eve trackers in the evaluation phase for different purposes including the user's attention allocation (29%), attention shift (11%) engagement (9%)search strategy (7%), gaze overlap in conversations (7%), global bias (2%), changes in attitude (2%), reading speed (2%) and resumption lag (2%). Moreover, some of them checked the user pupil data to measure the workload (9%). Besides eye trackers, 9% of the measures are related to the collected interactions through mouse and keyboards. Also, 9% of them analyzed the conversation between users in collaborative scenarios by checking the content of the user discussion.



Figure 3.17.: Descriptive statistics: behavior outcomes.

Performance Outcomes

Based on the collected papers, 75% of the studies focused on measuring the user's performance. I found them most papers include user performance besides behavior or perception. 49% of them involve both the user performance with behavior, and 51% involve perception with it. Only 30% of them measured all three outcomes include perception, behavior, and user's performance.

As can be seen in Figure 3.18, I identified seven constructs for estimating user performance. Among the constructs, the accuracy and completion time are the most popular measures with 66%. The accuracy is considered as the measure to show how the users were successful in conducting the assigned task. Accuracy is the most popular measure, with 40% of all measures for user performance. Completion time is considered the measure to show the speed to conduct the assigned task. This measure received 26% of all measures for user performance. In addition to these two measures, some researchers considered the reaction time (11%), comprehension (9%), self-reported measure(6%), driving performance (6%), and effectiveness(2%) as other methods for measuring the user performance.

Regarding the system performance, only 32% of the studies included this dimension. The reason for the difference between popularity to measure the user performance and the system performance has roots in two different focuses on designing eye-based IIS that I discussed in Section 3.4.2. Studies focus on user adaptation more involved in measuring user performance; however, the papers with system adaptation focus considered the system's performance as well. Also, I found that only 15% of the papers considered both user performance and the system performance in their reports. As can be seen in Figure 3.19, I identified three factors for measuring the system performance in which the system's accuracy is the most popular measure with 81%. Furthermore, few researchers considered how the system's running time (13%) was affected, besides how it could affect network traffic (6%).



Figure 3.18.: Descriptive statistics: user performance outcomes.



Figure 3.19.: Descriptive statistics: system performance outcomes.

3.5. Discussion and Future Research Directions

Analyzing the state-of-the-art along the dimensions of the conceptual framework supports in identifying the status quo and potential research gaps. In general, I conclude that besides the well-established usage of eye trackers for understanding user cognitive processes by analyzing their eye movement data in an off-line mode, there is an investment in using them in real-time. Particularly, more recently, there has been a growing interest in sensing the user's eyes, derive user state in real-time, and designing eye-based IIS that focuses on either system or user adaptation. Although the results show progress in this field during the last years, I believe that there are interesting research gaps that could be approached by researchers. In the following, I discuss the status quo and the suggestions for future research directions along with the findings by analyzing dimensions in three major categories of the conceptual framework: 1) influential factors 3.4.1, 2) eye-based IIS properties 3.4.2, and 3) outcomes 3.4.3. Table 3.1 presents the suggestions for future research directions for eye-based IIS.

Influential factors	Context and Task	 Increase research on team-level eye-based IIS and go beyond dyadic settings. Investigate eye-based IIS interactions between users and agents in team settings. Broaden research emphasizing the role of more specific user characteristics and design user-adaptive eye-based IIS. Investigate eye-based IIS beyond well-defined, restricted environments. Design eye-based IIS for enterprise environments (professional tasks and applications). Go beyond established monitor displays and investigate eye-based IIS for AR, VR, mobile and tablet users.
	Eye Tracking Technology	 Leverage more user-friendly eye trackers (e.g., no calibration, lighter, smaller, etc.) for eye-based IIS. Design privacy-aware eye tracking technology. Integrate eye trackers into VR and AR devices. Deliver cost-effective eye trackers, including advancing algorithms for converting webcam and mobile phone camera data to be leveraged for eye tracking.
	Experimental Setup	 Go beyond lab experiments and conduct field studies with eye-based IIS. Conduct online experiments with eye-based IIS using crowdsourcing. Increase the sample size in the experiments in order to deliver more generalizable findings.
Eye-based ISS Properties	Sense and Reasoning	 Integrate and test eye gaze metrics. Investigate the combination of eye gaze metrics. Investigate short terms user states such as emotion and stress. Investigate long term user states such as gender, personality, WMC, etc. Combine eye trackers with other users' input resources such as face recognition and biosignals (heart rate variability), EEG, or FMRI.
	Eye-based IIS Focus	 Intensify research on providing corrective feedback and increase user's awareness. Test providing corrective feedback for further contexts and tasks. Solve usability issues with corrective feedback in eye-based IIS. Individualize corrective feedback.
Outcomes	Perception and Behavior	 Standardize reporting about users' perceptions and behavior. Analyze users' behavior measures based on users' eye movement data. Establish evaluation frameworks for perception and behavior measures. Leverage further data resources to understand user states better while working with eye-based IIS.
	Performance	 Emphasize analyzing system performance in future research. Compare eye-based IIS performance with other attention detection techniques (e.g., mouse-based detection).

Table 3.1.: Summary of future research directions based on identified perspectives.

3.5.1. Influential Factors

Contexts and Tasks

Findings in the context analysis presented in Section 3.4.1 show that both individual and team levels received attention by researchers. However, the amount of research on the individual level is much higher than on the team level. Thus, there is a need to conduct further research in a team-level direction. Besides, I found that most existing team-level studies focus on a dyadic setup while, in reality, there are many remote meetings with more than two users and more complex interactions among participants. Therefore, I suggest investigating eve-based IIS systems beyond dyadic settings and focus on designing systems that include more users. Additionally, the interaction between users and agents received less attention in existing research. This interaction can be in dyadic settings with collaboration between one user and the agent or considering an agent as one participant in a setup with more than one user. Also, I suggest considering further user characteristics beyond age, gender, previous experiences, and vision status. Users of eye-based IIS can also be characterized by WMC, personality, reading speed, etc. Individual differences can strongly influence the effectiveness of an eye-based IIS, and it may be necessary to adapt eye-based IIS accordingly. More specific user characteristics can be collected by using survey-based approaches or using user's eye movement data to predict certain user characteristics. Recently, this research direction received more attention in the eye tracking community however they are not integrated with an eye-based IIS focus (Al-Samarraie et al., 2018; Berkovsky et al., 2019; Conati et al., 2017; Haves and Henderson, 2018; Hoppe et al., 2018; Raptis et al., 2016; Toker et al., 2013). I believe that having a deeper look into more specific user characteristics and designing user-adaptive eye-based IIS is an interesting avenue for future research. Regarding the physical context, existing studies are mainly conducted while users engage with digital applications in well-defined, restricted environments rather than in open environments. However, eye-based IIS usage, especially with head-mounted devices, can be used in more open environments such as driving a car, working with industrial tools, or teaching in classrooms. So far, in open environments, the use of eye trackers is mainly limited to collecting eye movements and understanding cognitive processes, and limited studies are focusing on providing eye-based IIS on this basis. Some researchers have emphasized that their eye-based IIS could be used in open environments. For example, Kern et al. (2010) have suggested exploring Gazemarks in the automobile context as interruption handling can highly improve safety.

Besides open environments, I also identified that integrated digital applications rarely include enterprise applications. In existing research, the focus is preferably on private usage, including entertainment, internet usage, online learning, etc. Therefore, I suggest investigating eye-based IIS at the workplace in combination with professional tasks and applications. For example, Tremblay et al. (2018) have called to include their approach in real-time operational support applications.

I expect the usage of eye-based IIS goes beyond desktop-based or open environment scenarios as well. First, with the increased usage of VR and AR technologies, I expect future research in eye-based IIS within these technologies. Combining VR and AR technologies with real-time eye movement data can support both system and user adaptation in the future. Second, I assume further research focusing on tablets, smartphones, smartwatches, etc. since users frequently use them in their daily life. For example, Cheng et al. (2018) have mentioned that their approach would be rather suitable on a smartphone for cross-application usage instead of a cross-device environment.

As context and tasks are tightly interconnected, I expect that by investigating new contexts for eye-based IIS, more tasks should be added to the list of tasks discussed in Section 3.4.1. However, the browsing and search task is assumed to remain the most common task because users have difficulties in managing limited attentional resources while facing a massive amount of information.

Eye Tracking Technology

As discussed in Section 3.4.1, the current focus of researchers is on the usage of desktopmounted eye trackers that fit a well-defined, restricted environment. I assume that the use of different eye tracking technologies will increase by extending research to new contexts and tasks. For example, the usage of head-mounted eye trackers should increase by integrating open environments rather than working with digital applications. For that, there is a general need for more user-friendly eye trackers, with an easier or no calibration process, smaller and easier to carry, etc. Having proper devices can increase the adoption and usage of eye tracking technologies. Besides that, there is a need to integrate eye trackers as an embedded capability in VR and AR systems for designing eye-based IIS. Finally, an essential driver for adopting and using any eye tracking technology is the implementation of privacy features (Kunze et al., 2013; Steil et al., 2019).

Furthermore, future work should develop and evaluate cost-effective eye tracking technologies leveraging webcams. In order to achieve mainstream adoption of eye-based IIS, dedicated eye trackers are still expensive and not commonly available. Webcam and smartphone cameras have the potential to become reliable devices for algorithms to record and analyze gaze data through low-quality images. So far, several studies have focused on the reliability of low-cost and webcam-based eye trackers (Burton et al., 2014; Dalmaijer, 2014; Zugal and Pinggera, 2014), but the findings show that these eye trackers did not receive researchers' attention in the field of eye-based IIS.

Experimental Setup

As presented in Section 3.4.1, existing experimental studies mainly focus on controlled lab environments. As a future research direction, I suggest extending investigations on the usage of eye-based IIS in more natural environments. D'Mello et al. (2012) suggested testing their system in real-world computer-enabled classrooms. Mariakakis et al. (2015) also suggested testing SwitchBack in more natural surroundings. Alt et al. (2012) proposed to examine whether the approach applies to image-based content besides advertisement. Although such intention is already mentioned in most of the identified studies, the findings show that it is not investigated a lot so far. Furthermore, replicating existing eye-based IIS studies in the field, reporting the results, and identifying new requirements would be interesting future contributions. Also, I assume that the average sample size for eyebased IIS studies is low. By extending the sample size, researchers could control for more specific user characteristics and ultimately deliver more generalizable knowledge. I assume that further improvement of eye tracking technology regarding cost-effectiveness and easier calibration processes would enable running experiments with a higher sample size. I assume that by improving webcam and mobile phone camera-based eye trackers, one could conduct large-scale online experiments by leveraging crowd-based services such as Amazon Mechanical Turk.

3.5.2. Eye-based IIS Properties

Sense and Reasoning

In general, as shown in Section 3.4.2, researchers have used a broad spectrum of metrics for designing eye-based IIS. However, future research could add additional sensing metrics, such as smooth pursuit movements or entropy. Also, only a few studies used a combination of eye gaze metrics, while in some studies, using this approach could deliver more reliable findings (e.g., identify mind-wandering studies (D'Mello et al., 2017)). Nevertheless, the metric or the used combination highly depends on which specific user states researchers aim to model. As discussed in Section 3.4.2, so far, the interest and relevance of information is the main goal of collecting eye movement data. However, this source of information can be used for further identification of user states. Furthermore, to increase the quality of findings in user cognitive states and needs, eye trackers can be used in combination with other data resources. As an example, some researchers mentioned to add input variables, especially face detection features, as a valid point for future work (Ishii et al., 2013; Nguyen and Liu, 2016; Tremblay et al., 2018; Vrochidis et al., 2011). Roda and Thomas (2006) suggested using eye trackers beside biosignals like heart rate or EEG, brain signals with fMRI, etc.

Eye-based IIS Focus

As discussed in Section 3.4.2, existing research emphasizes on both user adaptation by increasing user awareness and system adaptation by increasing the intelligence level of the system. However, by analyzing the distribution of papers, I found that the number of papers focusing on user adaptation is slightly lower than designing intelligent technology. Further investigation shows that the number of papers in this direction increased during the last years. Thus, I suggest continuing the trend and ensure balancing both directions. Providing corrective feedback to the users can assist them in recognizing their failures and adapt them. This approach seems easier to be implemented and can be used in various situations than system adaptation. Furthermore, in this type of focus, users who are directly involved in interacting with corrective feedback and usability of such feedback are essential. There is a need to investigate different types of corrective feedback based on user characteristics, task, and context. For example, D'Mello et al. (2012) proposed to identify individual differences and adapt gaze-reactive statements depending on these

differences as an essential focus of further research. Also, Akkil and Isokoski (2016) and D'Angelo and Gergle (2016) have noted that participants struggled to interpret sporadic eye movements; more research is also required for finding the best options. Some studies like Jo et al. (2015), D'Angelo and Gergle (2018) and Newn et al. (2017) have shown the interest of the community in finding the appropriate type of gaze-based highlighting method as corrective feedback; however, they are limited to a specific task, and it is not possible to generalize the results. Besides the representation of such corrective feedback, there is a need for further studies on how often and when eye-based IIS should provide feedback. For example, D'Angelo and Gergle (2016) considered a selective option for users to decide how often and when the gaze visualization is displayed as a need for further improvement. Also, researchers should investigate corrective feedback's personalization since the users may have different cognitive abilities that influence the effectiveness of corrective feedback.

Additionally, most eye-based IIS considered only one type of focus category. However, I suggest searching for synergies between the two paradigms. As eyes are the primary sense to process information while interacting with the environment, researchers can consider designing systems that integrate both implicit and corrective feedback. For example, Mariakakis et al. (2015) suggested implementing more features for SwitchBack, e.g., automatic scrolling when the user reaches the bottom of the text or magnifying glasses to enlarge the current read line of text.

3.5.3. Outcomes

Perception and Behavior

As discussed in Section 3.4.3 and Section 3.4.3, researchers leverage different measures for reporting user perceptions and behavior. It has been recognized that meeting user expectations is a critical success driver for eye-based IIS. However, I believe that the focus on systematically measuring these outcomes should increase in future research. 40% of the collected papers did not include any measures for perception. Knowing more about user perception and reporting it supports the research community in understanding the impact of eye-based IIS on the user and ultimately designing better systems. Delivering eye-based IIS aligned with user expectations is a critical factor for the mass adoption of this technology. Besides user perception, researchers in the field of eye-based IIS can use eve trackers that are integrated into the design procedure for the evaluation as well. Surprisingly, 43% of the collected papers ignored reporting user behavior. Therefore, I suggest providing information about user behavior in the field of eye-based IIS as a new standard in corresponding publications. Researchers can use the same approach to find user states to discover more about user behavior and include them in their findings. Therefore, there is a need to extend the leveraged behavior measures in future studies. Besides, I observed diverse approaches to collect perception and behavioral data by researchers. For perception, interviews and surveys are the most common way, but mixed methodologies such as think-aloud approaches and eye tracking can also be used to measure more accurate perception and behavior. Regarding the behavior, so far, researchers have focused on eye
tracker and log data, however integrating further devices such as EEG, fMRI, and face recognition as advanced evaluation tools could be valuable. Finally, researchers collected perception data through various techniques. To create more accurate results, I suggest developing a framework and a standardized survey and interview protocols for future studies.

Performance Outcomes

As discussed in Section 3.4.3, I investigated performance from two perspectives: user and system performance. Most papers include user performance as the main goal of the eye-based IIS. The way to measure user performance is directly related to the user task. However, only a few papers analyzed system performance. Therefore, I suggest future research should include the system performance of eye-based IIS as well. In general, there is a need to establish a proper methodology for measuring and reporting this information. Furthermore, as there are different ways to compare the attention of users (through mouse movements, EEG, etc.), I suggest future research comparing the different algorithms used in eye-based IIS with other attention-aware systems. For example, Akkil and Isokoski (2016) suggested the future development of GazeTorch by comparing the algorithm with mouse-based pointing systems.

3.6. Summary

As Study I in this thesis, I presented the results of a SLR study focusing on eye-based IIS. This SLR study is based on the suggested approach of Kitchenham and Charters (2007) and covers papers from the beginning of 2008 to the end of 2019. Conducting this study before doing the DSR project support to identify different dimensions for designing innovative IS applications that integrate real-time eye movement data as suggested by IS researchers (Davis et al., 2014; Dimoka et al., 2012; vom Brocke et al., 2013). Furthermore, the identified papers and the results from the analysis serves as the related work for the design cycles and supports highlighting MRs and DPs. Especially the eye-based IIS application that focuses on increasing users' awareness in data exploration tasks and resuming an interrupted task.

By improving eye tracking technology using eye movement data in real-time as an input for designing IIS increased significantly in the last years (Chuang et al., 2019; Nakano et al., 2016). Eye-based IIS provides innovative interactions enabling advanced forms of system and user adaptation. However, existing studies on eye-based IIS are scattered, and there is a lack of a systematic overview of existing research in this field. This study contributes to the field of eye-based IIS by providing a conceptual framework for designing such a system. Later, building on the developed conceptual framework, 47 identified papers were analyzed in detail, and I presented the state-of-the-art of eye-based IIS. Also, I identified research gaps and outlined possible future research directions. Mainly, the results from context and tasks show that the studies rarely include enterprise applications. Additionally, I search the same search stream for this SLR study within a selection of journals in the IS community

known as the Basket of 8⁻³. The findings show the lack of research on integrating realtime eye movement data for designing innovative IS applications. Eye tracking technology is known as the dominant tool for IS studies that integrated neuroscience tools (Riedl et al., 2017), but the focus is on using it for diagnostic purposes. These results are synced with the emphasis of researchers in IS field that there is a lack of research on integrating neuroscience tools such as eye tracking technology to design innovative IS applications (Davis et al., 2014; Dimoka et al., 2012; vom Brocke et al., 2013).

³https://aisnet.org/page/SeniorScholarBasket

4. Research Methodology ¹

4.1. Design Science Research

This thesis follows the DSR paradigm to deliver an innovative solution for real-world problems (Hevner et al., 2004). I mainly addressed two problems associated with limited attentional resources by designing attentive information dashboards with individualized VAF by integrating eye tracking technology. The first problem focuses on managing limited attentional resources during data exploration tasks. The second problem focuses on resuming interrupted data exploration tasks. I primarily address the lack of design knowledge for providing solutions that integrate user's eye movement data in real-time. I adapted the research approach from Kuechler and Vaishnavi (2008) and divided the entire DSR project into three sequential design cycles. In this DSR, I focus on artifact-centric approach (Peffers et al., 2007). The goal is to produce new knowledge by construction and evaluation of software artifacts (Kuechler and Vaishnavi, 2012). Figure 4.1 summarizes the three design cycles of the entire DSR project.

		Cycle 1	Cycle 2	Cycle 3
S	General Design Science Cycle	exploring attention management problems with dashboards and possible solutions exploration	attentive information dashboards with task resumption support	
	Awareness of Problem	literature review & problem exploration through exploratory eye tracking study	further reading and refinement of theoretical grounding	literature review on attentive systems with task resumption support (TRS)
wledge	Suggestion	provide suggestions based on results from literature review and exploratory study	adaptation of DPs based on empirical results and theoretical foundations	adaptation of DPs based on empirical results and theoretical foundations
id Goal Kno	Development	instantiation of suggestions as basic: • attentive dashboard • three VAF types	instantiation of DPs as: • attentive dashboard • individualized VAF	instantiation of DPs as: • attentive dashboard • individualized VAF (gaze-based TRS)
Operation ar	Evaluation	quantitative evaluation of VAF approaches (real-time Vs. off-line) (eye tracking pilot study)	quantitative evaluation of individualized VAF for data exploration (lab experiment)	quantitative evaluation of gaze-based TRS and the role of WMC (lab experiment)
	Conclusion	evaluation analysis and identification of most suitable VAF type	evaluation analysis, hypothesis supported	evaluation analysis and identification of most suitable gaze-based TRS based on WMC
		nascent design theory		

Figure 4.1.: Design cycles integrated in this thesis.

¹This Chapter is based on the following studies which are published or in work: Toreini and Morana (2017), Toreini et al. (2018b), Toreini and Langner (2019), Toreini et al. (2020b), Toreini et al. (2020a)

At the end of this DSR project and after completing all three design cycles, I summarized the findings in the general contribution as nascent design theory (Peffers et al., 2007) following six core components of design theory suggested by Gregor and Jones (2007). Section 8.3 presents the nascent design theory for designing attentive information dashboards with individualized VAF.

4.1.1. Design Cycle 1

I started the DSR project with a problem awareness step. The most common way to perform this step is to conduct a literature review or perform empirical research in forms of surveys or interviews (Hevner et al., 2004). After conducting a literature review, I found that existing research focuses on the role of attention and memory on single visualized information (Healey and Enns, 2012). There is only limited research on the cognitive state of the users focusing on information dashboards and there is a call for more research in this area (Alberts, 2017; Bera, 2016; Pauwels et al., 2009; Yigitbasioglu and Velcu, 2012). Moreover, detecting problems relevant to attention and working memory through surveys and interviews is not reliable since users have subjective views and have limited abilities to judge the effectiveness of their visual behavior (Dimoka et al., 2012). Therefore, as Study II in this thesis, I designed an exploratory study as a controlled laboratory experiment and simulated the data exploration task using an information dashboard. I chose a data exploration task since, in real-world scenarios, decision-makers explore dashboards from time to time to get a better understanding of their business. Periods to revisit dashboards can vary between daily visiting once a week, month, etc. The primary goal of this type of data exploration is to thoroughly investigate all information and try to understand the current status of the business. In this case, having effective attention allocation can increase the information processing of the user. I integrated eye tracking to evaluate user's visual behavior in this task. The usage of eye tracking provides the opportunity to objectively examine the attention of the user by analysing the recorded eye movement data and uncovering existing problems regarding to limited attentional resources. Furthermore, the role of visuospatial WMC of the users in allocating attentional resources is investigated. Based on the findings, I proposed six initial MRs for information dashboards that consider limited attentional resources and working memory of users when performing data exploration tasks. These MRs are classified into two categories, synced with the two focuses for designing eye-based IIS proposed in the framework of the SLR study in Study I. The first category includes three MRs that focus on making the system more intelligent for adapting the dashboard's layout based on the limited cognitive capabilities of users. The second category includes three MRs that focus on the necessity of providing new interactive features to users and focus on user adaptation. These interactive features should support managing limited attentional resources by increasing user's awareness about their previous allocation of the attentional resources. The derived knowledge can be used for articulating DPs and designing innovative features for dashboards.

Subsequently, I focused on implementing the MRs in the second category in Study III and proposed several designs for VAF grounded in research on attention and self-tracking feedback. To reach this goal, I proposed alternative designs for information dashboards that can track user's attention based on existing knowledge in literature. I derived two approaches of VAF types that operate based on eye movement data. One of these approaches is including off-line eye movement data from previous users as VAF and the other using real-time eye movement data of the user to provide individualized VAF. After the development of both approaches, I designed and executed an eye tracking pilot study to investigate each approach's effectiveness. For that, I integrated three groups of users that used VAF types. The first group used general VAF by providing an example of good Attentional Resource Allocation (ARA) integrating off-line records of eye movement data from other users that did the same task on the same dashboard. The second group also received the off-line records of eye movement data from other users but with improper ARA. The third group received individualized VAF that has represented their actual ARA and individualized VAF. Later, I compared the effects of general and individualized VAF, and the findings reveal the positive effects of individualized VAF on information processing compared to general VAF types.

4.1.2. Design Cycle 2

In the second design cycle, I investigated the influence of individualized VAF as the suggested solution for the data exploration problem in more detail. I started by refining the theoretical grounding for designing individualized VAF. For that I refined the corresponding MRs in Studies II and III. Later I derived two DPs based on the proposed system architecture to map these DPs to design features. Later, I instantiated an improved version of the attentive information dashboard and individualized VAF as a running software artifact. Next, following the suggested methods to evaluate DSR projects (Pries-Heje et al., 2008; Venable et al., 2012), I conducted a large-scale controlled laboratory experiment to assess the effectiveness of designed individualized VAF on users' ability to manage their limited attentional resources. In this experiment, I compared two design configurations with DPs activated (individualized VAF) and deactivated (general VAF with a text-based explanation about attention) based on the proposed research model and three hypotheses. The results from this study revealed that individualized VAF positively influences users' information processing by improving their ability to manage limited attentional resources. The results of this design cycle are captured in Study IV.

4.1.3. Design Cycle 3

In the third design cycle, I focused on the situation that information dashboard users are faced with interruptions while exploring information dashboards. In this case, users need to shift their limited attentional resources from the data exploration task (primary task) to a secondary task. There are many interruptions around employees in workplaces (Czerwinski et al., 2004; Mark et al., 2008), and they have difficulty to resume their primary task after finishing the interruption task (Hemp, 2009; Mark et al., 2005). Previous research has shown the need to provide attention management systems to support users in task resumption (Anderson et al., 2018). In this design cycle, I first conceptualized for short-term interruptions.

individualized VAF that supports resuming an interrupted task as, gaze-based TRS. I also presented dimensions that impact designing effective gaze-based TRS by analyzing previous research in the field of interruption and gaze-based TRS collected in the SLR study in this thesis as the Study 1. Based on that, I presented six MRs and three DPs for designing attentive information dashboards that support users' task resumption with gaze-based TRS. Later, I proposed the system architecture to map three identified DPs to the design features. For this study, I suggested three types of highlighting methods for gaze-based TRS as last point, heatmap and scanpath. Such gaze-based TRS works as a memory aid to remember previous visual behavior. For the evaluation part, I compared the effectiveness of provided DPs and designed gaze-based TRS with the situation that users do not receive such support. Following the suggested methods to evaluate DSR projects (Pries-Heje et al., 2008; Venable et al., 2012), I investigated the role of WMC on the effectiveness of each highlighting method for gaze-based TRS in a large-scale laboratory experiment as an exploratory study. The findings suggest that gaze-based TRS, especially heatmap highlighting, is beneficial for low WMC users when face with shortterm interruptions. Furthermore, users with high WMC may not need a gaze-based TRS

5. Design Cycle 1: Attention Management Problems and Possible Solutions ¹

5.1. Study II: Overview

Based on Gartner–Magic Quadrant for BI&A systems, visualizing the information and presenting the results in a comprehensive way is known as a critical capability of BI&A systems (Sallam et al., 2017). The main purpose of information visualization is to support users in perceiving patterns, which can be used to build appropriate explanatory models and improve their performance (Purchase et al., 2008). Information dashboards typically include several visualized information in one screen, and users need to explore them at the same time and find relationships between them before making decisions (Pauwels et al., 2009; Yigitbasioglu and Velcu, 2012). Information dashboards include a lot of compressed and essential information that supports comparing different perspectives and making proper decisions. However, conveying the enormous amount of information presented by information dashboards combined with a complex cognitive task is challenging for users (Niu et al., 2013; Sedig and Pasob, 2013). Decision-makers need to manage to allocate limited attentional resources while working with information dashboards to avoid typical attentional breakdowns, such as missing important information (Roda, 2011).

Furthermore, WMC is known as one of the individual characteristics affecting decision making as the mental representation about the status of the business is created in the users' working memory (Davern et al., 2012). Working memory has an impact on building contemporary global models of cognition and is involved in many complex cognitive behaviors, such as comprehension, reasoning, and problem-solving (Engle, 2002). Researchers found that individuals have different types of memories (Atkinson and Shiffrin, 1968; Baddeley and Hitch, 1974). Visuospatial WMC is an important individual differences variable while processing visualized information (Bačić and Fadlalla, 2016; Healey and Enns, 2012).

As discussed in Chapter 1, Section 1.2, so far, there is limited research on examining the role of attention and working memory while exploring information dashboards. To improve information dashboard design, knowledge about how users act and which of their cognitive limitations affect their performance is essential (Haroz and Whitney, 2012; Niu et al., 2013). To the best of my knowledge, no study has investigated the role of limited attentional resources and working memory while using information dashboards. Therefore, in this study, I investigate users' control on the attentional resources with the help of eye trackers, as suggested by researchers in the IS community, to assess the impact of systems on users' information processing (Dimoka et al., 2012). I also examine the role of visuospatial WMC as a critical working memory type on ARA while users are exploring information dashboards. As the Study II of the thesis, this study aims to find requirements

¹This Chapter is based on the following studies which are published or in work: Toreini and Morana (2017), Toreini and Langner (2019), Toreini et al. (2020b)

for designing more effective information dashboards by addressing the following question, as the second RQ in this thesis:

64

RQ2: What are meta-requirements for designing dashboards that consider users' cognitive limitations of attentional resources and working memory?

As Figure 5.1 shows, this study focuses on understanding existing attention relevant challenges of information dashboard users within the problem awareness section of this DSR project. An eye tracking experimental research approach is chosen for the problem awareness step because of the high reliability of eye movement data in finding problems related to allocating attentional resources compared to self-reported data (Dimoka et al., 2012). Therefore, this study contributes to the IS community and especially information dashboard design for BI&A systems by identifying MRs for designing more effective dashboards. These MRs are used in the remaining steps of the first design cycle, as well as the next design cycles. In the following, I explain the prototype design for the experiment, and the experimental setup. Next, the results are presented to explore different aspects of allocating limited attentional resources. Finally, I derive several MRs that are suggested for designing more effective information dashboards for data exploration tasks based on the results from the experiment and existing research on this field.

		Cycle 1	Cycle 2	Cycle 3
Π	General Design Science Cycle	exploring attention management problems with dashboards and possible solutions	attentive information dashboards for data exploration	attentive information dashboards with task resumption support
	Awareness of Problem	literature review & problem exploration through exploratory eye tracking study	further reading and refinement of theoretical grounding	literature review on attentive systems with task resumption support (TRS)
wledge	Suggestion	provide suggestions based on results from literature review and exploratory study	adaptation of DPs based on empirical results and theoretical foundations	adaptation of DPs based on empirical results and theoretical foundations
nd Goal Kno	Development	instantiation of suggestions as basic: • attentive dashboard • three VAF types	instantiation of DPs as: • attentive dashboard • individualized VAF	instantiation of DPs as: • attentive dashboard • individualized VAF (gaze-based TRS)
Operation an	Evaluation	quantitative evaluation of VAF approaches (real-time Vs. off-line) (eye tracking pilot study)	quantitative evaluation of individualized VAF for data exploration (lab experiment)	quantitative evaluation of gaze-based TRS and the role of WMC (lab experiment)
	Conclusion	evaluation analysis and identification of most suitable VAF type	evaluation analysis, hypothesis supported	evaluation analysis and identification of most suitable gaze-based TRS based on WMC
			nascent design theory	

Figure 5.1.: The focus of Study II in this DSR project.

5.1.1. Laboratory Experiment

5.1.1.1. Experiment Design

The experiment was conducted in a controlled lab environment. As an apparatus, I used Tobii 4C eye tracker with sampling rate of 90 Hz and the relevant license to record and analyze the eye movement data. Figure 5.2 shows the stages of this experiment. The study started by calibrating the eye tracker with the Tobii Eye Tracking Core Software. After that, the screen-based instruction was given to the participants and followed by control questions. The introduction explained the different steps of the experiment and defined the keywords in the designed dashboard to ensure that all participants understood the concepts. Before starting the main part of the experiment, the participants were asked to look at a dot in the middle of the display for a few seconds to check if calibrations have remained stable or not. The experiment's task was designed based on a simulated business scenario in which participants were asked to imagine themselves as the sales manager who recently joined a company. In a few minutes, they will have a meeting with their boss and need to explain the business status of the company during the last six months. To get ready for this meeting, they need to explore the information dashboard for two minutes. This step is considered as the first phase (data exploration phase). The dashboard layout used in this study can be seen in Appendix B as Figure B.3 After the exploration, the participants were interrupted for 30 seconds and were asked to wait while they did not receive any extra information (Appendix B as Figure B.2). After the break, they revisited the same information dashboard for one more minute, which counts as the second phase of the experiment (revisit phase). Besides, the participants got a timer in the footer of the dashboard that informed the remaining time. The eye tracking technology measured the relevant gaze data from each user individually and was stored in a log file through the data exploration. After the experiment, participants had the chance to rest for a few minutes. Later, they joined a psychological test from the free PEBL test battery (Mueller and Piper, 2014) to measure their visuospatial WMC by the visuospatial Corsi Block-Tapping test (Kessels et al., 2000).



Figure 5.2.: Stages of the designed exploratory experiment to investigate the role of limited attention and working memory.

To better investigate the users' attention and working memory in this experiment, I designed a particular information dashboard that includes six graphs, all with the same complexity regarding their appearance. At the same time, each is considered as an AOI. Figure 5.3 shows the AOIs as well as other sections in the dashboard for this study.



Figure 5.3.: The six pre-defined AOIs based on their position.

I acknowledge that this dashboard does not represent a real-world case, but it is explicitly designed for the experiment to enable isolating the effects of attentional guidance elements such as color, size, order, and shape. The idea is to control their effect on users' sensory memory and focus on goal-driven attention rather than stimulus-driven. Moreover, having a dashboard with equally complex elements empowers determining if the attentional resources are distributed appropriately since they need to allocate the same amount of attention on all graphs. Therefore, the six AOIs have the same type (bar chart) to minimize potential distractions of the users' attention allocation by the applied visualized format (Kelton et al., 2010). Each chart includes six chunks of information as a well-designed visualization promotes chunking (Patterson et al., 2014). I chose six chunks as 7 (plus or minus two) chunks of information is known as the magic number for individuals' WMC (Miller, 1956). To control the effect of attracting attention by the interactive options, the dashboard only includes static charts (Zhicheng Liu and Stasko, 2010) and they are ordered

in the same way (high to low value). In addition, the same grey color is used in each AOI to control for color effects (Bera, 2016). With the same visualization format, the number of chunks, color, and the order in all six AOIs, I argue that AOIs in this dashboard have the same complexity. Therefore an appropriate attention allocation while conducting data exploration task should be close to an equal distribution among six charts. As discussed in Chapter 2, Section 2.2.1, this dashboard layout supports controlling stimulus-driven attention and focuses on the user's goal-directed attention. Furthermore, I focus on the overt attention of the users by tracking their eye movement data.

5.1.1.2. Experiment Participants

In total 26 university students (8 female, 18 male) with an average age of 25.11 years participated in this experiment. The original number of participants was 29, but three participants were excluded from the dataset after the study. Two because the recorded eye movement data was available for less than 75 percent of the experiment's actual time, which means less than 90 seconds in the first phase and 45 seconds in the second phase. I assume that this was either an issue with the eye tracker or the participants did not do the task properly by allocating enough attention to the information dashboard. Moreover, one user was removed because of an error and missing results of the visuospatial WMC test.

To group the users based on their WMC, I investigated the collected information by visuospatial Corsi Block-Tapping test (Kessels et al., 2000). This test measures the Corsi Span, which is defined as the longest sequence a user can correctly remember. The higher the number of the Corsi Span, the higher the visuospatial WMC. Scholars have noted that working memory span tasks are the most proper way to compare the individual's WMC with each other (Conway et al., 2005). After running the computer version of this experiment with the help of the PEBL test battery (Mueller and Piper, 2014), the participants were divided into a low and high WMC groups following previously suggested categorization by Lerch and Harter (2001). For that, I performed a median split on visuospatial WMC. The median for the Corsi Span of the participants was 5.75, and I assigned the users with lower values to the *low* WMC group while users with a higher value than 5.75 were assigned to the *high* WMC group. After splitting, the users were equally distributed, with 13 participants in the high WMC and 13 in the low WMC group.

Regarding the sample size, it has been seen that to uncover usability issues, a formative study with eye tracking technology focusing on problem identification, requires fewer participants than a study without eye tracking (Bojko, 2013, p.163). Moreover, in HCI field, one of the main domains of eye tracking studies, a meta-study of 465 publication of the Conference on Human Factors in Computing Systems from 2014, unveiled that the mean sample size for using this method in in-person studies is 21 participants (Caine, 2016). This eye tracking study counts as an exploratory study to investigate users' visual behavior with high and low WMC on information dashboards rather than a confirmatory study. Therefore, I argue that the sample size of 2×13 is sufficient for the goal of this study, which is deriving MRs to design user-adaptive information dashboards in the future.

5.1.1.3. Measurements

Figure 5.4 shows the measurements used in this exploratory study using eye movement data. As can be seen, I mainly focused on three phases of the experiment includes the first visit phase, revisit and end of the task. Furthermore, I focused on three measurements in these phases includes **Attentional Resource Allocation (ARA)** that is measured in all three phases, **Attentional Resource Management (ARM)** that is measured in the first and end of the task and **Resumption Success Rate (RSR)** that is measure in the revisit phase.

ARA: I focused on the distribution of the attention on the six AOIs. These AOIs were named based on their position on the information dashboard, which can be seen in Figure 5.3. I measured the ARA performance of the user on each AOI as this measure is also used in previous eye tracking studies in the IS community (Cheung et al., 2017). Dwell time on each AOIs shows the total time that a user gazed at that AOI through the data exploration phases. Based on the eye-mind assumption, I used the total dwell-time of an AOI to measure ARA (Just and Carpenter, 1980). As an example: a dwell-time of 27 seconds on AOI-3 (Top-Right) provides the total amount of time that this user spent on this AOI in the first phase. I mainly measured ARA of the users in three points of time. First, in the first visit phase, second in the revisit phase, and their end of the data exploration task (first and revisit phase together).

Furthermore, I specifically focused on users' revisit behavior in the second phase of the data exploration task. The revisit phase serves to improve Attentional Resource Management (ARM). The ideal revisit strategy is to increase ARA on the previously low-visited AOIs while decreasing attention on previously high-visited AOIs. Therefore, for proper improvement, the participants need to remember their ARA in the first phase. To measure their ARA in the first phase, I grouped AOIs based on their dwell-time into "previously low-visited" AOIs and "previously high-visited" AOIs. For each group, I identified the three highest and lowest visited AOIs. Then, I measured the ARA of the users based on these two groups in the revisit phase. The results indicate how users can remember their previous visual behavior.

ARM: As the complexity of all six AOIs is considered to be the same, a more even distribution of dwell-time on all six AOIs is regarded as a better ARM. Therefore, I calculated the variance between the six dwell-times. A lower variance value among the dwell-time values indicates a better ARM. I calculated the ARA and ARM for the first phase of the data exploration task in addition to the end of the task.

RSR: Additionally, I investigated the users' task resumption strategy and the success rate. I explore this at the starting point by identifying the first selected AOI in the revisit phase and compared it with the last visited AOI of the first phase. Based on that, I found the Resumption Success Rate (RSR) for each group of participants separately. This value shows the percentage of users that could select the last AOI before interruption as the first AOI after the resumption.



Figure 5.4.: Measures used in the exploratory study.

5.1.2. Data Analysis and Results

5.1.2.1. First Visit Phase

To compare the ARA between the six AOIs in the first phase, I conducted one-way repeated measures Analysis of Variance (ANOVA) for each group separately. Furthermore, the heatmaps for both groups in the first visit phase can be seen in Appendix B Figure B.3.

For users with high WMC, Mauchly's test indicated that the assumption of sphericity was not violated. The results revealed no significant difference in the dwell-time between the six AOIs for users with high WMC (F(5, 60)=1.99, p=0.093). This result highlights that users with high WMC distributed their ARA properly. However, the same test for users with low WMC shows that there was a significant difference (F(5, 60)=2.73, p=0.027) in ARA. It shows that dwell-times are different among AOIs. In the next step, a post hoc comparison using the Tukey test was carried out. There was a significant difference between the dwell-time on AOI-1 and AOI-3 (p=0.023) and AOI-1 and AOI-6 (p=0.036). The users had, on average, around six seconds less dwell-time on AOI-3 and 5 seconds less in AOI-6 in comparison with AOI-1. Figure 5.5 displays the results of the ARA in the first phase for users with high WMC and low WMC. It shows that the ARA for both groups follows almost the same pattern while they allocated less time to AOI-3 (down-left) and AOI-4 (down-middle) than to the other AOIs.

Figure 5.6 shows the ARM of the users as the variance between the six dwell-times for the first visit phase. After checking the normality distribution of the data, an independent t-test was conducted to compare the ARM. The results show that there was not a significant effect between users with high WMC (M=19.69, SD=11.71) and low WMC (M=29.54, SD=17.41), t(21)=-1.647, p=0.11). Therefore, it confirms that both groups had the same performance in managing attention in the first phase. The lower mean values of ARM indicate that the dwell-times tend to be very close to each other, and the users could properly distribute their attention.





Figure 5.5.: Distribution of dwell-times on six AOIs after the first phase of the experiment.



Figure 5.6.: The ARM of participants after the first phase of the task.

70

5.1.2.2. Revisit Phase

Figure 5.7 shows the ARA on each group for users with high WMC (left) and low WMC (right). Furthermore, the heatmaps for both groups in the revisit phase can be seen in Appendix B Figure B.3. The revisit step is considered as an opportunity to improve the management of limited attentional resources. In this step, I investigated the users' ARA in two groups of AOIs: *"previously high-visited"* and *"previously low-visited"* AOIs. The selection of AOIs for each of these groups was discussed in the previous section.



Attentional Resource Allocation

Figure 5.7.: Total dwell-times on previously low and high visited AOI in the revisit phase for users with high WMC (left). Total dwell-times on previously low and high visited AOIs in the revisit phase for users with low WMC (right).

To compare the ARA of previously high-visited and low-visited AOIs for users with low WMC, I conducted paired-sample t-test. The results show that there is not a significant effect between dwell-time on previously high-visited AOIs (M=27.08, SD=6.78) and previously low-visited AOIs (M=24.34, SD=6.41), t(12)=-0.96, p=0.35). Furthermore, the results from the paired-sample t-test for users with high WMC show that there is no significant difference in previously low and high visited AOIs for these users t(12)=-0.34, p=0.73. The mean for scores of previously low-visited AOI was 26.13 (SD=5.87), while the mean of previously high-visited AOIs was 27.12 (SD=5.12). Moreover, running independent tests for comparing the results of previously low and previously high-visited AOIs does not highlight any difference between both groups with (t(23.81)=0.73, p=0.46) for previously low-visited AOIs and (t(22.33)=-0.28, p=0.77) for previously high-visited AOIs. As can be seen in Figure 5.7, both groups repeated their visual behavior of the first phase in the revisit phase and have higher dwell-time on AOIs with previously high values. In fact, they could not distinguish between previously low and high visited AOIs in

their revisit patterns. I can conclude that both groups had difficulties remembering their previous ARA and improving it. These results show that having higher WMC does not necessarily support having a better strategy in the revisit phase.

Furthermore, Figure 5.8 indicates the last selected AOIs by users in the first phase and the first selected AOIs at the beginning of the revisit phase. As can be seen, in the first phase, AOI-6 is selected more often as the last AOI than others, while other AOIs also were selected as the last AOIs by some other users. However, in the revisit phase, the first selected AOIs are AOI-1 and AOI-2 that are selected by 85% of the users and have the highest density. This shows that most users had difficulty resuming their data exploration after the break and decided to start the data exploration task from the AOIs that they already allocated attention more in comparison with the others. Therefore the RSR is low for the users in both groups. This highlights the difficulty of resuming the task after an interruption (Addas and Pinsonneault, 2015) also while exploring visualized information on dashboards. Overall, difficulties in remembering last visited AOI results in task resumption failures (Bailey and Konstan, 2006). Furthermore, starting the revisit phase from the top-left or top-middle, while these two AOIs have almost always the highest dwell-time in the first phase for both groups, indicates the lack of a proper strategy to improve ARM in the revisit phase. The results of dwell-time on previously low and highvisited AOIs in addition to these results, highlight that the users from both groups did not use the revisit phase as an opportunity to improve their information processing and repeated their information processing similar to the first phase.



Figure 5.8.: Comparing last selected AOI in the first phase with first selected AOI in revisit phase of the data exploration task.

5.1.2.3. End of the Task

In this section, I investigate the overall ARA after finishing the data exploration task (after finishing the first phase and revisit phase). The analysis follows the same approach as the first phase by investigating the ARA and ARM of the users. Figure 5.9 shows the dwell-time on each AOIs for users with high and low WMC after finishing the task. Furthermore, the heatmaps for both groups at the end of the task can be seen in Appendix

B Figure B.3. To compare the dwell-time on the six AOIs, I conducted a repeated measure one-way ANOVA. The Mauchly's test indicated that the assumption of sphericity was not violated and the results revealed a significant difference in dwell-times between the six AOIs for both groups of users with high WMC (F(5, 60)=5.149, p=0.000) and users with low WMC (F(5, 60)=5.338, p=0.000). Therefore, the ARA is not distributed well among the six AOIs after finishing the task, and users allocated more attention to some of them. Comparing the two groups reveals that users of both groups allocate more attention to the top left and middle charts (AOI-1 and AOI-2) than to other charts.



Attentional Resource Allocation

First Visit Phase (Two Min.) + Revisit Phase (One Min.)

Figure 5.9.: Distribution of dwell-times on six AOIs at the end of the task.

To test for a difference of allocated attention to AOIs, I conducted post hoc comparisons using the Tukey test for each group separately. Table 5.1 represents the results of this test in both groups. For the users with high WMC there was a significant difference between the dwell-time in five conditions, with three of them being relevant to AOI-1 and two of them to AOI-2 (Top-middle). As can be seen in Figure 5.9, the users with high WMC allocated more attention to the charts on the top left, and middle positions than on top right and down left positions. The differences are between AOI-1 and AOI-3 (p=0.039), AOI-1 and AOI-4 with a highly significant difference (p=0.000) and a weak significant result between AOI-1 and AOI-5 (p=0.046). Moreover, AOI-2 and AOI-4 with a highly significant differences in the conditions. Two of these conditions are relevant to AOI-1 (top-left), while the other four are relevant to AOI-2 (top-middle). AOI-1 is different from AOI-3 (p=0.003) and AOI-4 (p=0.005). Furthermore, the ARA is highly different between AOI-2 and AOI-3 (p=0.000), AOI-4 (p=0.000) while lower significant for AOI-5 (p=0.017) and AOI-6 (p=0.009). These

	Users with High WMC		Users with Low WMC		
Conditions	diff	р	diff	р	
AOI-1 / AOI-2	-0.389	1.000	1.797	0.988	
AOI-1 / AOI-3	-6.725	0.021 *	-10.290	0.003 **	
AOI-1 / AOI-4	-9.671	0.000 ***	-9.981	0.005 **	
AOI-1 / AOI-5	-6.156	0.046 *	-7.207	0.106	
AOI-1 / AOI-6	-5.740	0.078	-7.662	0.070	
AOI-2 / AOI-3	-6.336	0.036 *	-12.087	0.000 ***	
AOI-2 / AOI-4	-9.282	0.000 ***	-11.779	0.000 ***	
AOI-2 / AOI-5	-5.767	0.076	-9.004	0.017 *	
AOI-2 / AOI-6	-5.351	0.123	-9.459	0.009 **	
AOI-3 / AOI-4	-2.946	0.741	0.308	1.000	
AOI-3 / AOI-5	0.569	0.999	3.083	0.882	
AOI-3 / AOI-6	0.985	0.997	2.628	0.937	
AOI-4 / AOI-5	3.515	0.569	2.774	0.922	
AOI-4 / AOI-6	3.931	0.441	2.319	0.963	
AOI-5 / AOI-6	0.415	1.000	-0.455	0.999	
Note: * $p < 0.05$, ** $p < 0.01$, *** $p < 0.001$					

Table 5.1.: The results from Turkey test by comparing the dwell-times of participants on six AOIs after finishing the data exploration task.

results highlight that users with low WMC are mostly focused on AOI-2, and this AOI has a significant difference compared to all other AOIs except AOI-1.

As I controlled for elements that affect ARA of the users, such as color, size, motion, and importance level in the dashboard's design (discussed in Section 5.1.1.1), I can argue that the position of charts in the information dashboard is the main reason for unbalanced ARA. It can be seen that users allocate the attention differently on the elements which are positioned top left and top middle than the other locations. Comparing the ARA in the first and after the revisit phase, I can conclude that the revisit phase did not help users with low WMC improve their ARA. Although users with high WMC had a proper allocation in the first phase, they changed their behavior and focused on some of the AOIs more than on others.

In the next step, I compared the ARM of the users of both groups at the end of the task. Figure 5.10 shows ARM of the users at the end of the experiment. Furthermore, the heatmaps for both groups at the end of the task can be seen in Appendix B Figure B.3 and shows how the ARM of the users was in the end. The results from the independent wilcox test showed that there was a highly significant difference between the variance of dwell-times of users with high and low WMC (W=34, p=0.008). The mean of users with high WMC was 40.33 (SD=29.28), while the mean of users with low WMC was 72.59 (SD=37.90). This shows that users with low WMC have more challenges to manage their attention than high WMC users. In comparison to the ARM in the first phase, the mean value for the performance of low WMC changed from M=29.54 (SD=17.41) in the first phase to M=72.59 (SD=37.90) after finishing the second phase. Whereas for users with

high WMC, this change was from M=19.69 (SD=11.71) to M=40.33 (SD=29.28). This indicates that the revisit strategy does not support any of those groups to improve their ARM. Nevertheless, the overall ARM was worse for users with low WMC compared to users with high WMC.



Figure 5.10.: The ARM of participants at the end of the task.

5.1.3. Derived Meta-requirements

The results from the previous section highlight difficulties of users while exploring dashboards. Based on these results, the theoretical foundation discussed in Chapter 2, Section 2.2 and existing attention and visual memory challenges of users in the information visualization (Healey and Enns, 2012), I extracted six MRs for designing innovative dashboards that consider users limitations in data exploration task. Table 5.2 shows the list of identified MRs. There are three MRs that focus on system adaptation without involving users. These types of MRs concentrate on collecting data about users and make the dashboard more intelligent. The second focus is user adaptation, in which three MRs focus on increasing the user's awareness about their problem by providing feedback and let them improve allocating attentional resources. These two categories are also aligned with the identified focus for eye-based IIS that discussed in the Study I of this thesis.

As the first MR (MR1), I argue that there is a need to recognize users' visuospatial WMC and adopt the dashboard's design based on that. This is known as an important individual characteristic in processing visualized information (Healey and Enns, 2012; Toker et al., 2013), and also previous studies have shown that personalized features affect the information processing and attention of the users (Tam and Ho, 2006). Moreover, this study shows that the chart's position affects the ARA of the users similar to the findings from

other studies that investigated other UI types (Lorigo et al., 2008; Nielsen, 2006). Also, the total number of available information significantly influence the pattern of attention decay (Ahn et al., 2018). In the designed dashboard, the top-left and the middle-top position were visited more than other locations for both groups. Furthermore, the results show that users with low WMC are allocating significantly more attention to the middle-top position than to different positions. At the same time, this is not the case for users with high WMC. Therefore, as the second MR (MR2), I argue that a user-adaptive information dashboard should adopt the position of charts based on their level of importance. As an example, the charts with vital information should be positioned on the left-top or top-middle. Moreover, as the third MR (MR3), I argue that the position of important charts should be considered slightly different for users with high and low WMC.

The results of the ARA indicate low performance for both groups. Moreover, in the revisit phase, they had the chance to improve their ARA by visiting the previously lowvisited AOIs and ignoring high-visited AOIs. However, but both groups failed to select a proper revisit strategy. Therefore, as the fourth MR (MR4), I argue that there is a need to support ARA of users while exploring dashboards. Providing attention feedback is suggested by researchers to support this need (Göbel and Kiefer, 2019; Otto et al., 2018; Qvarfordt et al., 2010). Also, comparing the last selected AOI in the first phase and first selected AOI in the revisit phase uncovered that the users of both groups have difficulties in resuming their data exploration and repeated it from the beginning. Previous studies have shown that this happens when users have difficulty in remembering which activities have been completed (Singh, 1998). Therefore, as the fifth MR (MR5), I argue that there is a need to guide users in selecting a proper revisit strategy that supports them to resume their task in the case of an interruption. Previous studies have shown that providing support for resuming an interrupted tasks with visualization of users eye movement data helps users to have lower resumption lag and improve their performance (Jo et al., 2015; Kern et al., 2010). Also, the results of the ARM are different for users with high and low WMC. It shows that users with low WMC have a significantly lower performance in managing their attention than users with high WMC. As the sixth MR (MR6), I argue that this attention feedback should be individualized and adapted to the users' WMC.

Focus	#	Meta Requirements			
tation	MR1	Information dashboards should be able to adopt their design based of the visuospatial WMC of their users to support data exploration task.			
m Adapi	MR2	11 Information dashboards should be able to adopt the position of chart based on their level of importance in order to be more visible for use			
Syster	MR3	Positioning important charts should be a bit different for users with high and low visuospatial WMC.			
otation	MR4	The information dashboard should support users with high and low visuospatial WMC to control their limited attentional resources by providing attention feedback while exploring information dashboards.			
r Adap	MR5 The information dashboard should be able to support users to proper revisit strategy that helps resuming the interrupted task				
Usei	MR6 The attention feedback should be individualized and adapted to thusers visuospatial WMC.				

Table 5.2.: MRs for designing innovative information dashboards that consider limited attentional resources and working memory.

5.1.4. Summary

As Study I in the first part of the first design cycle, I conducted an exploratory eye tracking study to identify users' attention problems with high and low WMC while exploring information dashboards. Eye tracking technology is used to assess the impact of information dashboard on users' information processing capacity as suggested by IS researchers (Dimoka et al., 2012). Although a single visualized information can support users to have a better overview of the data, several of them on one page can challenge the limited cognitive ability of the users again. Nevertheless, there is a lack of knowledge in the cognitive status of users while using information dashboards (Niu et al., 2013; Yigitbasioglu and Velcu, 2012). Designing effective dashboards requires a detailed understanding of the underlying cognitive processes of their users (Lerch and Harter, 2001). Therefore, as Study II, I investigated the visual behavior of information dashboard users with eye tracking technology. Specifically, I focused on the role of limited attentional resources and visuospatial WMC as two important individual characteristics and checked how these elements affect data exploration tasks on dashboards.

This study contributes to the information dashboard design research by identifying six MRs that should be considered when designing dashboards. As a practical contribution, these findings support dashboard designers to better understand the role of limited attention and working memory and design more effective dashboards by especially considering the visuospatial WMC of their user and their attention allocation strategy. Also, these MRs can be used by researchers in DSR to define DPs, suggest new features, and investigate the effect of them (Gregor and Hevner, 2013). Both groups of MRs offered in this study are suggested by researchers to investigate for future BI&A systems integrating eye tracking

technology (Silva et al., 2019). Generally, there are only a few attentive IS applications focusing on system adaptation with eye tracking technology (e.g., (Maglio et al., 2000) focused on system adaptation). Also, there is a lack of usage on eye tracking technology for increasing user-awareness while working with IS applications (Lux et al., 2018). The results from Study I shows the increasing interest of the community on providing feedback based on users' eye movement data that increase support them to track themselves and increase their self-awareness. However, there is a lack of evidence on using integrating self-tracking such techniques in the workplaces (Rivera-Pelayo et al., 2017). Therefore in this thesis, I mainly focus on the user adaptation category of the MRs (MR4, MR5, MR6) to integrate in the next steps of the DSR project. These MRs are covered by providing attentive information dashboards with individualized VAF for data exploration and resuming interrupted tasks.

5.2. Study III: Overview

The results from the Study II show that users have difficulty allocating their limited attentional resources while exploring information. Attention is a limited resource of humans (Chun et al., 2011), and the huge amount of information can create a poverty of attention (Simon, 1971). When using information dashboards in a data exploration task, a proper allocation of attentional resources is essential for the decision-maker to avoid missing important information. Thus, there is a need to manage attention during dashboard exploration. Also, users need to scan complex displays systematically to extract relevant information (Proctor and Vu, 2006). Missing important information, maybe the consequences of high attentional demand or an inappropriate attention allocation (Roda, 2011). Based on the provided list of MRs, I focus on increasing users' awareness perspectives to provide suggestions. Especially in this study, I focus on providing suggestions for the **MR4**, which is the need to provide VAF to support users.

In the digital environment, researchers have called for designing AUI and specifically attention management support features that preserve users from such attentional breakdowns (Bulling, 2016; Roda and Thomas, 2006; Vertegaal, 2003). The details for AUIs are discussed in Chapter 2, Section 2.3.2. Eye trackers are known as the main tool for designing such systems (Bulling, 2016; Majaranta and Bulling, 2014). Based on the eye-mind assumption, where users are fixating is underlying their cognitive process, such as dedicating their attentional resource (Just and Carpenter, 1980) and users mostly explore dashboards through their eyes. The attentional allocation is the set of processes enabling and guiding the selection of incoming perceptual information (Eriksen and Yeh, 1985). AUIs that provide VAF is known as being supportive of recovering from attentional breakdowns and improving performance in several contexts (D'Mello et al., 2012; Otto et al., 2018; Sarter, 2000; Sharma et al., 2016). Study I provides a list of studies that focus on increasing user's awareness by providing VAF. Also, the research gap for integrating such support is already discussed in Chapter 1, Section 1.2. I focus on closing this gap and address the third RQ in this study:

RQ3: What type of VAF enhances users' ability to manage attentional resources while exploring information dashboards?

As Figure 5.11 shows, this study focuses on the remaining steps in the first design cycle. In this study, I first propose MRs to design attentive information dashboards with VAF and then investigate the effects of two common approaches for designing VAF using eye movement data (off-line usage of eye movement data Vs. real-time usage of eye movement data). For that, I suggested three different VAF types, two based on off-line records and one based on real-time tracking of user's eye movement data. Later, I evaluated the effects of these three VAF types on users' ability to manage limited attentional resources in an eye tracking pilot study and presented the results.

79

		Cycle 1	Cycle 2	Cycle 3
	General Design Science Cycle	exploring attention management problems with dashboards and possible solutions	attentive information dashboards for data exploration	attentive information dashboards with task resumption support
$\prod_{i=1}^{n}$	Awareness of Problem	literature review & problem exploration through exploratory eye tracking study	further reading and refinement of theoretical grounding	literature review on attentive systems with task resumption support (TRS)
wledge	Suggestion	provide suggestions based on results from literature review and exploratory study	adaptation of DPs based on empirical results and theoretical foundations	adaptation of DPs based on empirical results and theoretical foundations
nd Goal Kno	Development	instantiation ofsuggestions as basic:attentive dashboardthree VAF types	instantiation of DPs as: • attentive dashboard • individualized VAF	 instantiation of DPs as: attentive dashboard individualized VAF (gaze-based TRS)
Operation an	Evaluation	quantitative evaluation of VAF approaches (real-time Vs. off-line) (eye tracking pilot study)	quantitative evaluation of individualized VAF for data exploration (lab experiment)	quantitative evaluation of gaze-based TRS and the role of WMC (lab experiment)
	Conclusion	evaluation analysis and identification of most suitable VAF type	evaluation analysis, hypothesis supported	evaluation analysis and identification of most suitable gaze-based TRS based on WMC
		nascent design theory		

Figure 5.11.: The focus of Study III in this DSR project.

5.2.1. Meta-requirements

As the first part, I discuss the initial MRs for designing an information dashboard sensitive to the users' attention. I mainly focus on providing support for data exploration tasks as **MR4** derived in the previous study.

Information dashboards and the included interactive technologies have the potential to bias decisions by focusing attention on a limited set of alternatives, increasing the salience of less diagnostic information, and encouraging inappropriate comparisons (Alberts, 2017; Dilla et al., 2010). What users see or do not see depends on how they allocate their attention while interacting with BI&A dashboards. Missing important information on BI&A dashboards can be explained by phenomena such as inattention blindness (Mack and Rock, 1998) or change blindness (Simons and Rensink, 2005), which has the root in inappropriate attention allocation. Having a comprehensive overview of the presented information on BI&A dashboards and also notice changes from the past are essential for the decision-makers. Therefore, I argue that attentive information dashboards should be able to capture the users' current visual attention. Thus, I propose the first MR (MR1) as the need to monitor the users' visual attention.

There exist subjective as well as objective measurements for the users' visual attention.

Researchers argue that the users' eye movements and their current eye fixations are the approximation for their visual attention and cognitive processes (Hayhoe and Ballard, 2005; Kowler, 2011; Liversedge and Findlay, 2000; Rayner, 1998). Eye tracking technology can be used to detect the users' gaze position and collect the relevant eye movement data (such as dwell-time, fixation and saccade) in real-time (Duchowski, 2002). Based on this data, the users' visual attention can be extracted and use as an input for designing innovative systems (Bulling and Gellersen, 2010; Bulling et al., 2011). Thus, I propose the second MR (MR2) as the need for integrating eye tracking technology to collect users' eye movement data.

Decision-makers face the challenge of making biased decisions by focusing their attention on a limited set of alternatives on BI&A dashboards (Alberts, 2017; Browne and Parsons, 2012; Dilla et al., 2010). Having only a subset of the required information can result in inaccurate decision making. Providing feedback on their current visual attention can enable them to allocate their visual attention more efficiently and avoid attentional breakdowns and miss important information. Feedback refers to sending back information about what action has been done, and various kinds of feedback are available for interaction design (Preece et al., 2015). As discussed in the results of the Study I, how and where users paid attention is known as a valuable source of information to provide corrective feedback. Such feedback can increase users' awareness and direct users to allocate attention to missed important information while exploring the dashboard. Moreover, having efficient attention allocation on the dashboard supports decision-makers to compare the results and find the changes. Thus, I propose the third MR (MR3) as the need for providing feedback on the users' visual attention.

With these initial MRs for designing attentive information dashboards that provide VAF for data exploration tasks, I proposed three types of VAF. Two of them work based on off-line records of eye movement data, and one works by tracking users' eyes in real-time. The development of these VAF types is discussed in the next session.

5.2.2. Instantiation of VAF Types

To better understand the effect of each VAF type, I designed two specific information dashboards that increase the internal validity of the experiment. As can be seen in Figure 5.12, these dashboards have a similar layout and used in two phases of the experiment. Similar to Study II, these dashboards are designed in a way that minimizes the influence of external factors on the user's ARA. Therefore, as discussed in Chapter 2, Section 2.2.1, this dashboard layout supports controlling stimulus-driven attention and focuses on the user's goal-directed attention. Furthermore, I focus on the overt attention of the users by tracking their eye movement data. The actual size of this dashboards with the content can be seen in Appendix C in Figures C.1 and C.2. The designed information dashboards consist of six charts while each is considered as an AOI. I tried to design each AOI in a way to have the same complexity by having the same chart format, amount of information chunks, same size, no color, etc. With having the same complexity, I assume that a proper ARA is close to an even distribution of the ARA on all six AOIs. To select an example

81

of a proper and improper ARA, I considered the dwell-time on six AOIs from the users in Study II. For that, I calculated the variance among six dwell-time values for all users. Later, having lower variance shows that ARA was close to the even distribution, and proper and the higher variance shows the improper ARA examples.



Figure 5.12.: The information dashboard layout used in two rounds of the experiment.

Feedback is known as a constructive element that sends back information about what action has been done while allowing the user to continue with the activity (Preece et al., 2015). Therefore, a proper design of VAF for data exploration tasks on information dashboards should enable users to recognize how their previous ARA was and let them continue their data exploration task. Such support can be either delivered by the usage of off-line records of eye movement data from previous expert users (e.g., (Sridharan et al., 2012)) or by using users' real-time eye movement data (e.g., (Qvarfordt et al., 2010)). Figure 5.13 presents the summary of these three VAF types as suggested for the first design cycle of the DSR project. First, I designed an individualized VAF by considering the user's eye-movement in real-time. This type of VAF displays the actual ARA by presenting the dwell-time on each chart of the dashboard as a time format. I assume that having such information assists users to remember their previous ARA. Consequently, it helps in selecting a suitable data exploration strategy within the revisit phase. The second VAF type is a general VAF that presents an example of improper ARA. The design of this VAF type is equivalent to the individualized VAF, and only the duration values are different in order to provide an example of an improper ARA. Furthermore, the users receive an explanation that informs them that it is an improper example of ARA. The third VAF, is again a general VAF with the same design. In contrast to the second VAF, the values in this VAF type expose an example of a proper ARA. I assume that both second and third VAF types, as general VAF types, give a hint to the participants for recalling their previous ARA from their own memory, consequently, let them plan the revisit phase. The dwell-time values of second and third VAF types are the same for all participants while they changed for the users with an individualized VAF. Moreover, users received a short text on top of each VAF that explains the VAF type. The values of the proper and improper VAF types are coming from the ARA of users with the individualized VAF. I first conducted the study with the users that received an individualized VAF and considered the dwell-time on six AOIs in the first phase of the experiment from these users to find the values for general VAF types. For that, I calculated the variance among the collected six dwell-time values for all participants and selected the ARA with the lowest variance as the proper example of ARA and the highest variance as the improper ARA. Figure C.3 in Appendix C shows the design layout of all VAF types used in this study. Individualized VAF shows the actual ARA of the user and for the general VAF the values in Figure 5.13 are used.

1		G1: Ir	ndividualized	IVAF
ime	First VAF Type	Chart 1 Title	Chart 2 Title	Chart 3 Title
Real-t	Individualized VAF Present actual dwell-time on each AOI	00 m: 27 s <u>Chart 4 Title</u> 00 m: 21 s	00 m: 17 s <u>Chart 5 Title</u> 00 m: 10 s	00 m: 15 s <u>Chart 6 Title</u> 00 m: 22 s
		G2: Imp	roper ARA	example
scords	Second VAF Type	Chart 1 Title 00 m: 23 s	Chart 2 Title 00 m: 07 s	Chart 3 Title 00 m: 10 s
	$\left \begin{array}{c} General \ VAF \ - \\ Improper \ example \\ of \ ARA \end{array} \right \left \begin{array}{c} \text{Example with highest} \\ \text{variance of six dwell-times} \\ \end{array} \right $	Chart 4 Title 00 m: 13 s	Chart 5 Title 00 m: 40 s	Chart 6 Title 00 m: 12 s
ine R		G3: Pr	oper ARA e	xample
off-l		Chart 1 Title	Chart 2 Title	Chart 3 Title
	Third VAF Type	00 m: 21 s	00 m: 20 s	00 m: 19 s
	General VAF - \bigvee Example with lowest variance of six dwell-times	Chart 4 Title	Chart 5 Title	Chart 6 Title
	of ARA	00 m: 22 s	00 m: 18 s	00 m: 20 s

Figure 5.13.: Three VAF types that are suggested and evaluated in Study III.

5.2.3. Eye Tracking Pilot Study

5.2.3.1. Experiment Design

To test the influences of VAF types, I executed an eye tracking experiment in a controlled lab environment. The experimental design was 2×3 mixed design in which with or without VAF was manipulated within subjects and VAF types were manipulated between subjects. In this experiment, each subject had to conduct data exploration tasks in two rounds on two different information dashboards, which both had the same design but different content. These dashboard can be seen in Appendix C in Figures C.1 and C.2. For each data exploration task, participants had three minutes to explore the dashboard in total. After two minutes, they were interrupted for 30 seconds and later resumed the task for one more minute. During the first round, the interruption phase counted as a break, and participants were asked to wait for 30 seconds before continuing. During the second round, each participant got one of the three VAF types presented in the previous section.

The experiment procedure started by calibrating the eye tracker for each participant individually with Tobii Eye Tracking Core Software. In the second step, screen-based instructions were given to explain the experimental steps and illustrated the concepts used in the dashboard. These instructions were followed by control questions to ensure the common understanding of concepts on the dashboard. After that, the main part of the experiment started and can be seen in Figure 5.14. First, participants conducted the first round (without VAF) of the experiment in which they explored the dashboard for two minutes (first phase) and then they had a break for 30 seconds. Next, they had the opportunity to resume the data exploration task for one more minute and finish the first round. Then, the calibration status was checked again, and in case of any errors, the system enforced the execution of a recalibration. Also, a rest phase was included for two minutes to control for carryover effects from the first round. Next, the second round (with VAF) of the experiment is started. In this round, users had a data exploration task similar to the first round but this time with a new dashboard. In this round, participants received one of the designed VAF types after two minutes instead of having a break. As time was controlled in each step, the participants got a timer in the footer of the screen that displayed the remaining time in all phases of the experiment. In the end, demographic questions were asked as a survey.



Figure 5.14.: Stages of the designed eye tracking pilot study.

5.2.3.2. Experiment Participants

In total 29 university students (8 female, 21 male) with an average age of 25.03 years (SD=2.32) participated in this experiment. After checking the collected data, I removed two participants since the total collected dwell-time was less than 3/4 of the time assigned to each phase. This can be because these users ignored some part of the task or an error in the calibration. The remaining 27 participants were distributed across three groups (G). 11 participants were assigned to G1 with the individualized VAF, 8 participants to G2 with the improper example of ARA (general VAF) and 8 participants to G3 with the proper example of ARA (general VAF).

5.2.3.3. Measurements

The data analysis is focused on comparing the ARA performance and Attention Shift Rate (ASR) of the participants as the dependent variables in the revisit phase of both rounds (with and without VAF). The revisit phase is considered as an opportunity to improve the ARA by focusing on previously low-visited charts of the first phase. Thus, for each round, I detected the three low-visited charts in the first phase and measured the dwell-time on these charts in the revisit phase. Also, I captured the focused attention and how they engaged in the task by tracking the number of transitions between the AOIs. ASR shows the centering of attention on a limited stimulus field (Ishii et al., 2013). Here, having a lower number of transitions is considered as having higher focused attention. Figure 5.15 shows the visual representation of measurements considered in this study with and example values.



Figure 5.15.: Measures used to compare three suggested VAF types.

5.2.4. Data Analysis and Results

To test the difference between groups in these two rounds, I conducted a mixed design ANOVA test with groups as between subject and the dependent variables as within subject. I did this test for each dependent variable separately and the results did not show any significant difference among the three groups for both dependent variables. I assume that these results occurred because of the low sample size in this eye tracking pilot study. To investigate the effects of VAF types in more detail, I performed a within-subject analysis for each group separately by conducting a paired-sample t-test. Figure 5.16 shows the ARA for each group. Furthermore, the heatmaps based on collected eye movement data for three groups can be seen in the first visit, and the revisit phase for each round of the experiment can be seen in Appendix C and Figures C.4 and C.5.

For the group with an individualized VAF (G1), the results confirm that there was a significant difference in the ARA performance of the first (M=45.69, SD=14.08) and second (M=64.95, SD=16.38) round of the experiment (t(10)= -3.773, p=0.003). Although Figure 5.16 shows that the ARA performance improved in the second round for all groups, the paired-sample t-test for G2 and G3 does not show a significant difference. Therefore, I can infer that the individualized VAF helped participants to find previously low-visited charts in a better way than the other two general VAF types.





Figure 5.16.: ARA during the revisit phases of experimental rounds.

Moreover, Figure 5.17 shows the ASR for each group. After checking the normality assumptions for transitions, I used the paired-sample wilcoxon test to investigate the effect of VAF types on ASR. For G1, the results reveal that there was a weak significant difference in ASR for the first (M=32.82, SD=17. 86) and second (M=19.55, SD=9.05) round of the experiment (v=56.5, p=0.040). Also, for G2, the results of the paired-sample t-test show a weak significant difference for the first (M=32, SD=16.14) and second (M=25, SD=16.06) round (t(7)=2.3287, p=0.05). However, for G3, the results do not show any significant difference. I can infer that, although the focused attention improved for all three VAF types, the effect of an individualized VAF and an improper VAA example was stronger than the effect of a proper VAA example.



Figure 5.17.: ASR during the revisit phases of experimental rounds.

5.2.5. Discussion

Results from this eye tracking pilot study reveal that individualized VAF supported users to improve both ARA performance and the ASR in comparison with proper or improper examples of ARA as general VAF types. Providing actual values of previous ARA supports users recalling this information from their memory in a better way than receiving the general hints. These results confirm that users with general VAF types had difficulty to recall their previous ARA and to plan for the revisit phase. Regarding the ASR, I found that giving the individualized and improper example of ARA as VAF types supports users in focusing better on the task. However, having a proper ARA example influences neither the ARA performance nor focused attention. The results explain that having an improper ARA example as a hint helps users to better focus during the revisit phase than the proper ARA example. Following the categories highlight in the conceptual framework from in Study I, the individualized VAF has the focus on user adaptation by increasing their awareness about previous visual behavior. Therefore, I conclude that the suggested initial MRs for designing attentive information dashboards that support data exploration tasks with individualized VAF is supportive.

88

Although the existing results provided valuable findings that help the next steps of this DSR project, this study has some limitations that can be covered in future investigations. This study is considered as an exploratory study to get preliminary results of using the three suggested VAF types. Next, there is a need to differentiate the effects of these VAF types on a large-scale for robust theorization and confirmatory studies. Primarily there is a need to conduct experiments for the individualized VAF since there is evidence that it can support users better than the other general VAF types.

5.2.6. Summary

In the Study III, I cover the remaining stages of the first design cycle. This study focuses on providing solutions for attention management problems in the data exploration task by integrating eye tracking technology. Here, the initial MRs for designing attentive information dashboards are identified, and three types of suggestions for attention support features are presented. Following the categories highlight in the conceptual framework from in Study I, the primary focus of suggestions are on user adaptation by increasing their awareness about previous attention allocation. I propose three suggestions for VAF by integrating real-time and off-line recordings of eye movement data and testing the suggestion in an eye tracking pilot study. So far, eye tracking technology was mainly used to evaluate artifacts by investigating the user's visual behavior of the user (Riedl et al., 2017) rather than using them to design neuro-adaptive IS applications suggested by IS researchers (Davis et al., 2014; vom Brocke et al., 2013). Furthermore, these devices are proposed as the primary method for designing AUI that supports users in managing their limited attentional resources (Bulling, 2016). In this study, I focused on designing attentive information dashboards and VAF that contributes to the IS field by enhancing user capabilities (Dimoka et al., 2012) in managing their limited attentional resources. Following suggestions from vom Brocke et al. (2013), in this design cycle, I focused on both the design and evaluation of three suggested VAF types with eye tracking technology. Furthermore, based on the list of possible contributions in the field of NeuroIS provided by Riedl and Léger (2016), this design is related to the eighth contribution, using NeurosIS tools and delivering an IT artifact which tracks and adapts to the user's attentional state. Moreover, I contribute to the ninth contribution by providing individualized VAF as a live bio-feedback that helps users better control their limited attentional resources. In the second design cycle, I focus on improving the design for individualized VAF and investigating its effect on users' ability to manage limited attentional resources in a large-scale experiment for robust theorization and confirmatory study.

6. Design Cycle 2: Attentive Information Dashboards for Data Exploration ¹

6.1. Study IV: Overview

As the users' attention is known as a limited resource (Chun et al., 2011), users cannot attend all stimuli at the same time and need to select on the specific part of while exploring UIs. UI designers try to overcome this limitation by directing users' attention to important items. They attract users' attention by integrating specific design elements (e.g., size, color, animation, etc.) with UI designs considering the user's specific tasks. Treisman and Gelade (1980) discussed these elements and the role of them as the feature integration theory of attention. Also, in the IS research, several studies focused on investigating the role of such guiding representation on users' attention, such as Cheung et al. (2017) and Hong et al. (2004a).

However, in some tasks like data exploration tasks, a comprehensive overview of all information presented on the UI may be required. Users need to scan complex displays systematically to extract the relevant information (Proctor and Vu, 2006). In these tasks, users need to conduct several attention shifts to allocate their attention to all information rather than on guided attention by specific design features. The attention shifts of users can be different based on their tasks (Yarbus, 1967). Therefore, to process all information on the UI, they need to manage their limited attentional resources by themselves. This is similar to the data exploration task on information dashboards. Providing well-designed information dashboards that enable the exploration of data and support proper decision making is emphasized by Gartner-Magic Quadrant for BI&A systems Cindi et al. (2019). However, even with a proper design, users can only focus on a limited set of information and miss other parts while exploring dashboards (Alberts, 2017; Dilla et al., 2010). The results from the Study II confirms that users have difficulty in managing their limited attentional resources while exploring information dashboards. Therefore, there is a need for designing innovative information dashboards that provide support for managing attentional resources in data exploration tasks.

In the Study III of this thesis, I investigated two common approaches for providing attention support using eye movement data for data exploration tasks. These approaches either use off-line recordings of eye movement data from other users and provide VAF or use real-time eye movement data for individualized VAF. I tested these two approaches in an eye tracking pilot study, and the results show that providing individualized VAF is more supportive that the off-line records. Furthermore, investigating the papers collected in Study I with the focus on supporting data exploration on different platforms has shown that using eye trackers to provide feedback about ARA supports users in improving their self-awareness and therefore their performance. For example, Sharma et al. (2016) has shown that a gaze-aware feedback tool significantly improved ARA and students' learning gains. D'Mello et al. (2012) has found that informing students about their information

¹This Chapter is based on the following working paper: Toreini et al. (2020c)

processing behavior supports reorienting their attentional patterns and promotes learning, motivation, and engagement. Sarter (2000) has shown the need for giving feedback for effective ARA to support users managing their limited attention while working with highly complex information-rich environments. Deza et al. (2017) demonstrated the benefit of using eye trackers to improve users' performance in a visual search task since the huge amount of data makes operators susceptible to information overload and ARA inefficiencies. Qvarfordt et al. (2010) and Sridharan et al. (2012) investigated using eye movement data as feedback to improve the inspection method in applications such as radiology and imaginary analysis.

As discussed in Chapter 1, Section 1.2, it has shown that the usage of eye trackers for designing individualized VAF is a research gap in IS. The results from Study III show the effective role of individualized VAF for the data exploration task. However, there is a lack of design knowledge describing how to design individualized VAF for information dashboard users to enhance their ability to manage limited attentional resources. Also, there is a need for a confirmatory study about the effectiveness of individualized VAF in a large-scale experiment as a suggested method to evaluate DSR projects (Pries-Heje et al., 2008; Venable et al., 2012). Therefore, I focus on the fourth RQ in this thesis as follows:

RQ4: How to design attentive information dashboards providing individualized VAF to enhance users' ability to manage attentional resources for data exploration tasks?

As Figure 6.1 shows, I focus on the second design cycle of the DSR project in this study. After highlighting the need for such support in Chapter 1, Section 1.1, Section 1.2 and Study II, I continue this design cycle by proposing two theory-grounded DPs. Next, I instantiate both DPs in a software artifact and evaluate them in a large-scale laboratory experiment. Specifically, I analyze users' eye movement data while exploring the information dashboard in the first visit, after receiving the individualized VAF (revisit phase) and at the end of the task. I compare users that received individualized VAF with users that received general VAF in the form of a simple text explanation about the importance of proper ARA while exploring the dashboard.

General Design Science Cycleexploring attention management problems with dashboards and possible solutionsattentive information dashboards for data explorationattentive information dashboards with task resumption supportAwareness of Problemliterature review & problem exploratory eye tracking studyfurther reading and refinement of theoretical groundingliterature review on attentive systems with task resumption support (TRS)Suggestionprovide suggestions based on results from literature review and exploratory studyadaptation of DPs based on empirical results and theoretical foundationsDevelopmentinstantiation of suggestions attentive dashboard • three VAF typesinstantiation of DPs as: • attentive dashboard • individualized VAF (gaze-based TRS)Versionquantitative evaluation of verstrig pilot study)quantitative evaluation analysis, hypothesis supportedConclusionevaluation analysis and identification of most suitable VAF typeevaluation analysis, hypothesis supportedevaluation analysis and identification of most suitable VAF typeevaluation analysis, hypothesis supportedevaluation analysis, hypothesis supported			Cycle 1	Cycle 2	Cycle 3
Awareness of Problemliterature review & problem exploratory eye tracking studyfurther reading and refinement of theoretical groundingliterature review on attentive systems with task resumption support (TRS)Suggestionprovide suggestions based on results from literature review and exploratory studyadaptation of DPs based on empirical results and theoretical foundationsadaptation of DPs based on empirical results and theoretical foundationsDevelopmentinstantiation of suggestions as basic: • attentive dashboard • three VAF typesinstantiation of DPs as: • attentive dashboard • individualized VAF (gaze-based TRS)instantiation of DPs as: • attentive dashboard • individualized VAF (gaze-based TRS)Evaluationquantitative evaluation of VAF approaches (real-time Vs. off-line) (eye tracking pilot study)quantitative evaluation of individualized VAF for data exploration (lab experiment)quantitative evaluation of gaze-based TRS and the role of WMC (lab experiment)Conclusionevaluation analysis and identification of most suitable VAF typeevaluation analysis, hypothesis supportedevaluation analysis, based on WMC	Operation and Goal Knowledge	General Design Science Cycle	exploring attention management problems with dashboards and possible solutions	attentive information dashboards for data exploration	attentive information dashboards with task resumption support
Oppositionprovide suggestions based on results from literature review and exploratory studyadaptation of DPs based on empirical results and theoretical foundationsadaptation of DPs based on empirical results and theoretical foundationsDevelopmentinstantiation of suggestions as basic: • attentive dashboard • three VAF typesinstantiation of DPs as: • attentive dashboard • individualized VAF (gaze-based TRS)instantiation of DPs as: • attentive dashboard • individualized VAF (gaze-based TRS)Evaluationquantitative evaluation of VAF approaches (real-time Vs. off-line) (eye tracking pilot study)quantitative evaluation of individualized VAF (lab experiment)quantitative evaluation of individualized VAF (lab experiment)Conclusionevaluation analysis and identification of most suitable VAF typeevaluation analysis, hypothesis supportedevaluation analysis and identification of most suitable VAF typeImage: Developmentevaluation analysis and identification of most suitable VAF typemascent design theory		Awareness of Problem	literature review & problem exploration through exploratory eye tracking study	further reading and refinement of theoretical grounding	literature review on attentive systems with task resumption support (TRS)
Of performentinstantiation of suggestions as basic: • attentive dashboard • three VAF typesinstantiation of DPs as: • attentive dashboard • individualized VAF (gaze-based TRS)instantiation of DPs as: • attentive dashboard • individualized VAF (gaze-based TRS)Evaluationquantitative evaluation of VAF approaches (real-time Vs. off-line) (eye tracking pilot study)quantitative evaluation of individualized VAF for data exploration (lab experiment)quantitative evaluation of gaze-based TRS and the role of WMC (lab experiment)Conclusionevaluation analysis and identification of most suitable VAF typeevaluation analysis, hypothesis supportedevaluation analysis and identification of most suitable VAF type		Suggestion	provide suggestions based on results from literature review and exploratory study	adaptation of DPs based on empirical results and theoretical foundations	adaptation of DPs based on empirical results and theoretical foundations
Evaluation quantitative evaluation of VAF approaches (real-time Vs. off-line) (eye tracking pilot study) quantitative evaluation of individualized VAF for data exploration (lab experiment) quantitative evaluation of gaze-based TRS and the role of WMC (lab experiment) Conclusion evaluation analysis and identification of most suitable VAF type evaluation analysis, hypothesis supported evaluation analysis and identification of most suitable VAF type		_ Development	instantiation of suggestions as basic: • attentive dashboard • three VAF types	instantiation of DPs as:attentive dashboardindividualized VAF	instantiation of DPs as: • attentive dashboard • individualized VAF (gaze-based TRS)
Conclusion evaluation analysis and identification of most suitable VAF type evaluation analysis, hypothesis supported evaluation analysis, and identification of most suitable gaze-based TRS based on WMC nascent design theory nascent design theory		Evaluation	quantitative evaluation of VAF approaches (real-time Vs. off-line) (eye tracking pilot study)	quantitative evaluation of individualized VAF for data exploration (lab experiment)	quantitative evaluation of gaze-based TRS and the role of WMC (lab experiment)
nascent design theory		Conclusion	evaluation analysis and identification of most suitable VAF type	evaluation analysis, hypothesis supported	evaluation analysis and identification of most suitable gaze-based TRS based on WMC
			nascent design theory		

Figure 6.1.: The focus of Study IV in this DSR project.

6.2. Meta-requirements and Design Principles

In the first design cycle, I identified initial MRs to design innovative information dashboards that support data exploration (Study II). Also, I provided initial MRs for designing attentive information dashboard with VAF (Study III). Besides, I found preliminary evidence for the effectiveness of individualized VAF that works with real-time eye movement data in comparison with general VAF that works with off-line records of eye movement data. In the second design cycle, I investigate the influence of individualized VAF in more detail. I start by refining the theoretical grounding for designing attentive information dashboards and individualized VAF and refined the corresponding DPs.

As the first refined MR, I propose that the system needs to monitor the ARA of users in real-time (**Refined MR1**). Based on the eye-mind assumption (Just and Carpenter, 1980) user's eye movement data represents their ARA. Scholars have used user's eye movement data as an approximation for overt attention (Kowler, 2011). Also, eye trackers have the capability to collect eye movement data in real-time and use it for designing AUI (Bulling, 2016; Henderson et al., 2013; Majaranta and Bulling, 2014; Roda and Thomas, 2006; Vertegaal, 2003). Tracking the users' eye movement data in real-time provides the opportunity to design innovative IS applications (Davis et al., 2014; Riedl and Léger, 2016; **DP1:** Provide the information dashboard with the functionality to monitor the users' eye movements in real-time in order to analyze the users' ARA on the information dashboard when performing data exploration tasks

Being able to monitor users while exploring dashboards is a prerequisite to assist users in improving their ARA. Providing feedback that informs users about their previous ARA increases self-awareness and supports further improvement in information processing performance. Previous research has shown that tracking users and providing real-time feedback can influence users' behavior (Jung et al., 2010). Particularly, it has been seen that such feedback supports users in allocating their limited attentional resources more appropriately and ultimately improves their task performance while working with UIs with huge amounts of information (Deza et al., 2017; D'Mello et al., 2012; Göbel and Kiefer, 2019; Qvarfordt et al., 2010; Sharma et al., 2016; Sridharan et al., 2012). Therefore, I propose the attentive information dashboard should provide VAF to users before they finish their tasks and enable them to improve their information processing performance as the third refined MR (Refined MR3). VAF enables users to recognize their current ARA and potentially adjust it accordingly. The provided VAF should enable users to improve information processing while they explore the presented information. Therefore, the provided VAF needs to be individualized rather than generic as well as being lean and precise. In fact, an individualized VAF should increase self-awareness of the users about their goal-directed attention by presenting their eye movement patterns to them. I assume that having such feedback supports users to identify their attentional failure, such as missing important information. Therefore, I propose the need for individualized VAF as the fourth refined MR (Refined MR4). The proposed third and fourth MRs inform the second DP (**DP2**) as following:

DP2: Provide the information dashboard with the functionality to display individualized VAF based on the monitoring and analysis of the users' eye movement data to support information processing performance.

Table 6.1 summarizes the design activities of the second design cycle with the four refined MRs and corresponding DPs.
Design Cycle 1	Design	n Cycle 2
Initial MR	Refined MRs	DPs
	Refined MR1: Monitor the ARA of users in real-time.	DP1: Provide the information dashboard with the functionality to
Initial MR : The dashboard should support users in	Refined MR2: Estimate the user's ARA based on eye movement data recorded with eye trackers.	monitor the users' eye movement in real-time to analyze the users' ARA when performing data exploration tasks.
management of attentional resources by providing VAF while exploring data.	Refined MR3 : Provide feedback about the users' ARA to enable self-awareness.	DP2: Provide the information dashboard with the functionality to display individualized VAF based on the manitoring and analysis of the
	Refined MR4 : Provide lean, precise, and non-suggestive VAF by the information dashboard.	users' eye movement data to support users in improving information processing performance.

Table 6.1.: MRs and DPs of designing attentive information dashboards that provide individualized VAF for data exploration tasks.

6.3. Development

To map the DPs to design features, I propose the system architecture depicted in Figure 6.2. The system architecture comprises three subsystems. First, the **Information Dash-board Subsystem** connects to the BI&A system and presents information for decision-makers. A dashboard's layout typically comprises visual features (e.g., chart types, table, etc.) and interaction features (check the interaction list by Yi et al. (2007)) (Pauwels et al., 2009). Dashboard designers develop various layouts based on different purposes (e.g., planning, monitoring, communication, etc.) while users have different types of tasks, levels of knowledge and personality (Yigitbasioglu and Velcu, 2012). Moreover, users typically interact with a dashboard using the mouse, keyboard, or touch to explore the dashboard or search for specific information.

Second, the **Eye Tracking Subsystem** establishes a connection to the eye tracker and provides the functionality to track and store the users' eye movement data to extract the attentional states of users. Previous studies have followed different ways for extracting users' cognitive states as well as attentional status from the user's gaze data (Duchowski et al., 2018; Kowler, 2011). This subsystem provides the ability of real-time extracting of attentional states from users' collected gaze data.

Third, the **Attention-aware Subsystem** focuses on merging the user's attentional state with the dashboard layout and provides individualized VAF. In this subsystem, the attention analyzer component uses information from the eye tracking subsystem besides the visual features' coordination and interactions' log data from the information dashboard subsystem to derive the user's attentional spotlight. Hence, the dashboard becomes sensitive to the user's attention.



Figure 6.2.: System architecture for designing attentive information dashboards that support data exploration tasks.

I map the DP1 to the attentive information dashboard and the DP2 maps to the individualized VAF capability, building on the feedback generator component. The specific individualized VAF design can vary based on the feedback's purpose and the user's task and characteristics. In this study, I focus on supporting users in better information processing by allocating limited attentional resources properly while conducting data exploration tasks. Therefore, the individualized VAF should present the summary of previous ARA behavior to the user to increase self-awareness. To reach this goal, DP2 is instantiated by presenting the actual gaze duration on each visual feature (e.g., chart, tables, etc.) on the dashboard in a time format. I assume that providing such information enables users to assess their previous ARA and subsequently improve it in case it is needed. Figure 6.3 shows an instantiation of individualized VAF that exhibits the user's gaze duration on a dashboard with six visual features. Besides the individualized VAF, the following general text-based explanation was provided:

"Many users have a problem to allocate their attention properly while using information dashboards. In the following, you can see your attention allocation so far based on the time that you looked at each chart. Please think about your attention allocation performance in the previous step and then you will have one more minute to continue exploring the dashboard".



Figure 6.3.: Instantiation of DP2.

Furthermore, I designed a general VAF to compare it with individualized VAF. The general VAF uses similar text as in the explanation part of the individualized VAF. However, I do not provide any further information about individualized gaze duration values. I assume that providing such information supports users to think about their information processing performance and let them judge internally about potential weaknesses. The general VAF shows only a generic text as follows:

"Many users have a problem to allocate their attention properly while using information dashboards. Please think about your attention allocation performance in the previous step and then you will have one more minute to continue exploring the dashboard".

6.4. Hypotheses

Figure 6.4 depicts the underlying research model to study the impact of the suggested DPs in this DSR project. As can be seen, I compare two design configurations with DP1&2 activated (individualized VAF) and deactivated (general VAF). I investigated the users' information processing in three different phases, including the first visit phase, revisit phase, and end of the task. The first phase refers to the first time to explore the dashboard and the period before receiving one of the VAF types. The revisit phase is the period that is used to explore the dashboard after receiving one of the VAF types. Also, I consider the end of the task as the last period for the whole experiment (first and revisit phases). I consider the first visit as a control phase, which demonstrates that users have the same initial visual behavior. Subsequently, I propose hypotheses based on the users' attention after receiving the VAF types (revisit phase and end of the experiment).

To allocate an appropriate ARA in the revisit phase, first, users need to recognize their ARA in the first visit with the support of VAF. Previous research has shown that users have difficulties to remember their previous ARA and repeat their visual behavior in the revisit phases (Cane et al., 2012; Monk et al., 2008; Singh, 1998). Therefore, I argue that users with general VAF have challenges in finding an appropriate revisit strategy in comparison with users that receive individualized VAF. Users with an individualized VAF obtain the memory aid. Previous research has shown that providing VAF guides users to recognize high and low-visited parts of the UI (Göbel and Kiefer, 2019; Qvarfordt et al., 2010) and optimize their behavior in the next steps. Therefore, I define the first hypothesis **(H1)** as following:

H1: Providing individualized VAF results in better ARA performance in the revisit phase in comparison to providing generic VAF.

Moreover, having a proper strategy in the revisit phase leads to centering attention on specific elements on the dashboard rather than switching between different elements. It has been shown that providing VAF increases user's focus while conducting tasks (D'Mello et al., 2012). Centering attention results in having less ASR by users rather than shifting among different parts of the interface. Therefore, I define the second hypothesis **(H2)** as following:

H2: Providing individualized VAF results in less ASR in the revisit phase in comparison to providing generic VAF.

Also, previous research has shown that the stimulus' position in the UI affects how users allocate attention to them (Haugtvedt and Wegener, 1994; Lorigo et al., 2008; Nielsen, 2006). It has been seen that users are processing the information similar to the "F" pattern. In fact, users start from the left side of the UI and then allocate less attention to the information that comes later on the right side. A recent eye tracking study about dashboards by Tableau (Alberts, 2017) showed that users typically focus their attention on specific areas and thereby potentially miss other parts of the dashboard. Also, the results from the left side of the dashboard more intensively. Therefore, I argue that having individualized VAF withdraw users from focusing only on specific areas and support them to have better ARM. I define the third hypothesis (H3) as the following:

H3: Providing individualized VAF results in better ARM at the end of the task in comparison to providing generic VAF.



Figure 6.4.: Research model to investigate the effect of DP1&2.

6.5. Laboratory Experiment

6.5.1. Experimental Software and the Apparatus

To prepare the experiment, I first developed the software artifact. This artifact instantiates DP1&2 incorporating the functionalities for tracking the user's eye movement data in real-time. Furthermore, it collects the data required for further analysis in the evaluation section. As an apparatus that enables both tracking users' eye movement data in realtime and record the relevant eye movement data, I used Tobii 4C eye tracker with the corresponding license to store the data. This eye tracker has the sampling rate of 90 Hz and is considered one of the low-cost eye trackers in the market (Farnsworth, 2019). I selected this eye tracker since I argue that the usage of such devices for designing AUI is applicable for daily working tasks and on large-scale. Furthermore, the results from the Study I shows that this is the primary device for designing systems that integrate real-time eye movement data. I connected the eye tracker to a computer that displays the dashboard on a 21-inch screen with a resolution of 1920x1080 for all participants. I developed the experimental software in the .NET framework by using C programming language since Tobii provides the relevant Software Development Kit (SDK) for developing gaze-aware UI (Tobii Core SDK) and collecting data for research purposes (Pro SDK) on this framework. Also, for calibrating the eye trackers, I used Tobii Pro Eye Tracker Manager before running the developed software artifact. To extract the fixations and visualize heatmap in the data analysis section, I used PyGaze (Dalmaijer et al., 2014) as an open-source toolbox for eye movement analysis.

Since the quality of the research design should be judged on the basis of the factors that affect users' ARA, I maintained internal validity from four different perspectives in the experiment. *First*, I evaluated artifacts in a laboratory experiment, ensuring high internal

validity by minimizing the influence of external factors that affect the user's performance. Second, I minimized the influence of external factors that could affect the quality of collected eye movement data, such as movements and light conditions. For that, I controlled the calibration's quality several times during the experiment with the developed experimental software. Third, I used the collected eye movement data to select the users that conducted the experimental task as I asked in the instruction. Based on that, I removed a few users that ignored processing the dashboard or eye tracker could not collect their eye movement data. The dataset includes the participants who process the information dashboard for a minimum of 75 percent of the total dedicated time on each step.

Fourth, I controlled the elements that affect the stimuli-driven attention of the users while exploring. Figure 6.5 displays the dashboard layout that I designed and used for this experiment. As can be seen, this dashboard includes six charts, which I designed in a way to have almost similar complexity from their appearance. To reach that, all six charts have the same type (bar chart) to minimize potential distractions of the visualized format (Kelton et al., 2010). Moreover, all charts, words, and numbers have an equal size to avoid size influence (Alberts, 2017). Also, all charts include six chunks of information as a well-designed visualization promotes chunking (Patterson et al., 2014). I chose six chunks as seven (plus or minus two) chunks of information is known as the maximum capacity for individuals' WMC (Miller, 1956). To control the influences of attention by interactive options (Zhicheng Liu and Stasko, 2010), the dashboard includes only static charts. Besides, the gray color with the same variation is used in all AOIs to manage for color impacts on attention (Bera, 2016). Therefore, with the same visualization format, size, the number of chunks, no interactive options, and gray color, I argue that the six charts have an almost similar complexity from an information representation perspective. In this study, I also collected and analyzed the users' eye movements based on pre-defined AOIs on the dashboard. The dashboard includes six charts, and I consider each of them as one AOI. As Figure 6.5 shows, I named six AOIs based on their position on the dashboard layout. I use AOI's naming to discuss the results in the following sections.

Using elements that influence the users' stimulus-driven attention is common in the realworld dashboards and highly impacts ARA of the users (Alberts, 2017; Pauwels et al., 2009; Yigitbasioglu and Velcu, 2012). I acknowledge that a dashboard with the same complexity of all charts does not represent a real-world scenario. However, this design empowers to track the goal-directed attention of the users, as discussed in Chapter 2, Section 2.2.1. I followed this approach to maintain internal validity for the ARA, ASR, and ARM of the users and not a biased one based on stimulus-driven attention. Furthermore, I focus on the overt attention of the users by tracking their eye movement data.



Figure 6.5.: The designed information dashboard to control for stimulus-driven attention.

6.5.2. Experimental Design

To evaluate the effects of the DPs on dashboard users, I instantiated them in a running software artifact and conducted a laboratory experiment. For that, I applied a mixed model design with two groups (DP1&2 activated with individualized VAF and DP1&2 deactivated with general VAF) as between subjects and time (before and after receiving the VAF types) as the within-subject. I assigned participants randomly to either individualized VAF (treatment condition) or general VAF (control condition) before the experiment. Both groups had the same task in the experiment, and I limited the execution time in each experimental phase.

I started the experiment by calibrating the eye trackers using Tobii Pro Eye Tracker Manager. Next, the participants started a screen-based instruction that introduced different steps of the experiment. As a scenario, I told participants to imagine themselves as sales managers in a company. They recently joined the sales organization and are about to have a meeting with their boss. A few minutes before the meeting, they received the last six months' sales report in the form of an information dashboard. They required to prepare for this meeting by exploring the company's status with regards to sales data. Furthermore, I informed them about the experiment's timing and let them know that the experimental software included a timer for tracking the remaining time for each step. The instruction ended with control questions to ensure that participants understood the experimental steps and the provided information on the dashboard appropriately. Next, I provided a simplified variant of the dashboard used in the experiment without VAF (Check Appendix D, Figure D.1. This version included a dashboard with only two charts and supported users to get familiar with the experimental task and steps. Next, I asked them to rest for two minutes before starting the main part of the experiment. I added this break to control for carry-over effects between the trial and the main part of the experiment.

Figure 6.6 shows the steps for the main part of the experiment. In the "First Visit Phase" of data exploration, participants received the dashboard and explored it for two minutes. The information dashboard for this study can be seen in Appendix D Figure D.2. After that, they were interrupted for 30 seconds. In this step, participants received one of the two designed VAF types (individualized VAF or general VAF) based on the group that they were assigned randomly before the experiment. Examples of these VAF types can be seen in Appendix D, Figure D.3 and D.4. Later, in the "Revisit Phase", the participants were asked to revisit the same dashboard for one more minute. In the last step, participants and get ready for WMC tests. Finally, the two WMC (Corsi span and Digit span) tests were performed using the PEBL test battery (Mueller and Piper, 2014).



Figure 6.6.: The experiment's procedure used to evaluate the effects of DPs.

6.5.3. Participants

In total 92 university students (35 female, 57 male) with an average age of 23.45 (SD=3.39) participated in this experiment. I used student participants for the laboratory experiment, as this comes with two key advantages. First, in contrast to employees in organizations, students are not specifically trained in working with dashboards and are not biased from contextual information. Therefore, they have – similarly to novice users – little or no prior knowledge of the underlying experiment's process. Second, it is easier to reach student participants in a relatively large sample size with reasonable effort to achieve adequate statistical power. Consequently, students can be considered an adequate and representative sample in the experimental setup (Burton-Jones and Meso, 2008).

The students received 10 Euro as a financial incentive for participating and completing the experiment. I recruited participants from an experiment pool and randomly assigned 48 participants to the control group (general VAF) and 44 to the treatment group (individualized VAF). The initial number of participants was 107, but 15 participants were removed from the sample. For 12 participants, the available recorded eye movement data was less than 75 percent of the overall time (basically less than 90 seconds in the first visit or less than 45 seconds in the revisit phase). I assume that these 12 participants did not seriously engage in the data exploration task, or the eye tracker had technical problems recording their eye movement data. Two participants were excluded from the dataset because they did not correctly answer the control question in the post-experimental survey. Finally, I excluded one more participant because of self-reported health problems with the eyes.

6.5.4. Measurements

	Construct	Definition	Measurement	References
ion Processing isit Phase)	Attentional Resource Allocation (ARA)	Users performance in allocating attention on previously low-attended AOIs and ignore previously high-visited AOIs	Eye Tracking (fixation duration, number of fixations)	(Just and Carpenter, 1980) (Cheung et al., 2017) (Qvarfordt et al., 2010)
Informat (Rev	Attention Shift Rate (ASR)	The number of directing attention while processing information	Eye Tracking (total number of transitions pairs)	(Hong et al., 2004) (Blascheck et al., 2014)
Information Processing (End of the Task)	Attentional Resource Management (ARM)	The ability to distribute the attention properly among all stimulus on the screen	Eye Tracking (variance of fixation duration and number of fixations on six AOIs)	Self-defined
	Visuospatial WMC The capacity of user's visuo-spatial memory		Corsi Span	(Kessels et al., 2000)
Controls	Digit WMC Capacity of users regard to memorizing digit numbers		Digit Span	(Conway et al., 2005)
	Demographics	Gender, Age, Experience level	Survey	Self-defined and (Moore, G.C. 1989)

I measured several dependent and control variables during different steps of the experiment. Table 6.2 displays a summary of all measurements used in this study.

Table 6.2.: The dependent variables and controls used in Study IV.

As the dependent outcomes, I explored users' ability to manage limited attentional resources regarding ARA, ASR, and ARM. I measure the users' ARA following the approach used by Cheung et al. (2017). Based on this study, user's fixation duration and the number of fixations on each pre-defined AOIs are known as the user's ARA on that AOI. To recognize the user's performance on their ARA, I compared the fixation duration and number of fixations of the first visit with the revisit phases based on six AOIs. By chance, each chart would receive (100/6=16.67 percentage) of the ARA. This was subtracted from the actual ARA percentage, therefore yielding a score reflecting whether a given AOI was attended more (or less) than the theoretical average. I consider the revisit phase as the opportunity to enhance information processing performance. Therefore, the better performance results in having higher ARA on the previously low attended charts in the revisit phase. Furthermore, I measured users' ASR in the revisit phase. This measure is considered as centering of attention on a limited stimulus that exists in the field rather than shifting attention among all of them. It assumes that when the users have a clear strategy for visiting the information dashboard, they have less ASR. The eye tracking researchers have used the ASR of users among AOIs to explain how focused is user's attention (Bednarik and Tukiainen, 2006; Ishii et al., 2013). The ASR between AOIs is measured by the number of transitions, which is the movement of eyes from one AOI to another (I ignored the transitions within the same AOI). Consequently, the transition matrix represents the ASR between all possible combinations of AOIs (Ponsoda et al., 1995). In fact, the transition matrix is a descriptive summary representation of collected data that supports the analysis of users' data exploration behavior (Blascheck et al., 2014; Burch et al., 2011; Kurzhals et al., 2016). Furthermore, I measured the users' ARM at the end of the data exploration task. As explained in Section 6.5.1, all six AOIs on the dashboard have the same complexity and importance level from the presentation style. Therefore, I consider a more even distribution of attention on all six AOIs as a better ARM performance. For that, I calculate the standard deviation between fixation durations and the number of fixations on all six AOIs at the end of the data exploration task. Lower standard deviation values intimate that these six numbers are closer to each other, and the user properly distributed attention. Also, a higher standard deviation value demonstrates lower ARM.

Second, I measured several participant-specific control variables (demographics as well as two different WMC types). Regarding demographics, I captured gender, age, and compatibility with previous experience (Moore, 1989) to work with dashboards through survey questions (Appendix D in Table D.1). I considered the users experience since it plays a role in users eye movement data and managing limited attentional resources while exploring visualized information (Gegenfurtner et al., 2011). I also measured the users' WMC from two perspectives. The reason for choosing WMC as a control variable is its importance with regards to processing information, as described in Chapter 2, Section 2.2. WMC predicts the attention control of users (Kane et al., 2001; Kane and Engle, 2003) and has been defined as one important individual characteristic while working with visualized information (Borkin et al., 2016; Haroz and Whitney, 2012; Healey and Enns, 2012; Toker et al., 2013). Scholars have noted that working memory span tasks are the most proper way to compare the individual's WMC with each other (Conway et al., 2005). Also, individuals have different capabilities in remembering different information types. Consequently, different working memory spans exist to measure. In this study, I measure two types of users' working memory span as digit and visuospatial WMC. The reason is that dashboard users mostly deal with digits as well as visualized information on the dashboard. I collected the users' visual WMC by running a visuospatial Corsi Block-Tapping test (Kessels et al., 2000) and the digit span test (Conway et al., 2005) to measure the number of digits that can be memorized by a specific user. Both tests report the working memory span value that is the longest sequence a user could correctly repeat in each test. The higher the working memory span value, the higher the WMC.

6.6. Data Analysis and Results

6.6.1. Manipulation and Control Checks

Before testing the hypotheses, I checked whether the random-assignment between-participant conditions was successful or not by testing if the two groups do not differ concerning WMC and the three demographic controls assessed in the final survey.

The chi-squared test for comparing participants' gender per condition (individual VAF and general VAF) was not significant (chi-square=.558, p>.45). Therefore, random assignment for gender was successful. Moreover, the results from the wilcoxon signed-rank test for all the other control variables (age, experience level, Corsi span and Digit span) did not show any difference between the two groups (Appendix D, Table D.2) and the random assignment was successful. Also, to ensure that all users had the same visual behavior in the first visit phase, I analyzed the users' eye movement data and compared ARA, ASR and ARM between two groups. The result shows that users from both groups had similar visual behavior before receiving VAF types. I present the details in the following sections.

6.6.2. Attentional Resource Allocation

Figure 6.7 shows the heatmaps based on the user's ARA within both groups. The left column displays the ARA of the first visit phase, and the right column shows the revisit phase. In the first visit phase, visual behavior did not differ between the groups and the ARA was influenced by the position of the AOIs. For both groups, the left charts (AOI-1, AOI-4) received more attention than the one in the middle (AOI-2, AOI-5) and the middle AOI received more attention than the right side (AOI-3, AOI-6). Also, a column-based observation reveals that charts on the first row (AOI-1, AOI-2, AOI-3) have a higher ARA in comparison to the corresponding charts in the second row (AOI-4, AOI-5, AOI-6). These results confirm that the users are biased to allocate their attention to the left and top of the dashboards similar to other UI types (Lorigo et al., 2008; Nielsen, 2006).

During the revisit phase, the results from the general VAF group show that users repeated their visual behavior while the users in the individualized VAF changed it. Investigating through the rows shows that for the general VAF group, the left-sided AOIs have higher values than the right-sided AOIs. Also, column-based investigation indicates that the general VAF group had higher values for the AOIs in the upper position again in comparison with the AOIs in the lower position. However, the users in the individualized VAF group had more ARA to the right-sided AOIs as well as having higher ARA on AOIs positioned in the lower position. Furthermore, the distribution of fixation duration and number of fixation on each AOI can be seen in Appendix C in Figure D.5 and D.6.

Overall, by qualitative analyzing the visual behavior of both groups via heatmaps, I find that users with individualized VAF improved their ARA in the revisit phase, while the users with the general VAF tend to repeat their visual behavior after receiving a second chance to explore the dashboard.



Figure 6.7.: Heatmaps of both groups in the first and revisit phases.

To test the ARA performance hypothesis in the revisit phase (H1), repeated-measures regression analysis was carried out based on the percentage fixation duration and number of fixations. Besides, prior to the analysis, the ARA scores were centered around the mean average percentage (100/6). Following this, zero reflects the average percentage of ARA spend on an AOI at a given point in time. In the first model, I predicted the fixation duration per AOI in the revisit phase from the fixation duration of that AOI in the first visit, the experimental condition (0=general VAF group; 1=individualized VAF group), and their interaction. First, the effect of fixation duration percentage in the first visit was significant, b=1.145 percentage, SE=0.223 percentage, t(548)=5.129, p<.001; indicating that in the general VAF group, the fixation duration of an AOI was a strong predictor of the fixation duration of the same AOI in the revisit phase. In other words, participants show consistency in their behavior as expected regard to the fixation duration. Figure 6.8 shows on the left part the positive slope in the general VAF group. An AOI that received a higher percentage of ARA (in terms of fixation duration and number of fixations) in the first visit also received more ARA in the revisit phase. Second, there was a significant interaction of fixation duration percentage in the first visit and the individualized VAF

group, b=-0.75 percentage, SE=0.138 percentage, t(548)=-5.468, p<.001. This shows that the effect of the fixation duration of the first visit on the fixation duration of the revisit phase was compensated by the individualized VAF (Figure 6.8, left side). Thus, compared to the general VAF group, an AOI with a high fixation duration in the first visit had relatively less fixation duration in the second phase for the individualized VAF group. Vice versa, a previously AOI with low fixation duration had high fixation duration in the second phase.

Similar to fixation duration, an analogous analysis for the number of fixations also yielded two significant effects: First, the effect of the number of fixations in the first visit was significant, b=1.211 percentage, SE=0.221 percentage, t(548)=5.467, p<.001; indicating that in the general VAF group, the number of fixations on an AOI was a strong predictor of the number of fixations to the same AOI in the revisit phase, and participants show consistency in their behavior regard to the number of fixations. Second, the effect of the number of fixations in the first visit on the number of fixations in the revisit phase was compensated by the individualized VAF (Figure 6.8, right side). Crucially, the results show that there was a significant interaction of the number of fixations in the first visit and the individualized VAF group, b=-0.812 percentage, SE=0.133 percentage, t(548)=-6.067, p<.001.



Figure 6.8.: The interaction between and after VAF types for both groups.

To summarize, the results from the qualitative analysis of the heatmap as well as quantitative analysis for fixation duration and number of fixations show that ARA performance of users with the individualized VAF improved in comparison with the users with general VAF. Participants with the general VAF were consistent, and AOIs that were highly attended to in the first phase, also received more attention in the revisit phase. However, for the users with individualized VAF, an AOI that received more ARA (in terms of fixation duration and the number of fixations) in the first visit received less ARA in the revisit phase, and vice versa. Therefore, H1 is supported.

6.6.3. Attention Shift Rate

Figure 6.9 displays the transition proportions among six AOIs in the first and the revisit phase for both groups. In these matrixes, the number in the cell represents ASR in percentage between each possible AOI pairs. The reason to show the proportions rather than the actual number of transitions for each pair is the difference between the data exploration time in the first visit (two minutes) and the revisit phase (one minute). In addition, the color scaling shows the differences between values on these matrixes to ease the qualitative analysis.

	First Phase						Revisit Phase								
	total number of transitions (Mean=83.56 , SD=21.66)					total number of transitions (Mean=45.52, SD=13.67)									
	AOI1 AOI2 AOI3 AOI4 AOI5 AOI6						AOI1 AOI2 AOI3 AOI4 AOI5 AOI6								
VAF	AOI1	-	9.82%	1.37%	5.53%	$\mathbf{2.24\%}$	0.30%		AOI1	-	10.25%	1.24%	5.13%	$\mathbf{2.61\%}$	0.37%
	AOI2	10.07%	-	6.51%	1.15%	3.47%	0.55%		AOI2	10.89%	-	6.73%	1.05%	3.52%	0.73%
neral	AOI3	0.97%	4.99%	-	0.70%	1.05%	6.66%		AOI3	0.87%	5.72%	-	2.29%	0.82%	6.45%
Geı	AOI4	5.46%	1.47%	0.27%	-	5.06%	0.72%		AOI4	4.53%	1.28%	0.14%	-	5.31%	0.82%
	AOI5	2.69%	4.94%	0.60%	4.69%	-	5.41%		AOI5	2.75%	3.84%	0.78%	5.17%	-	5.35%
	AOI6	0.32%	0.62%	5.81%	1.07%	5.48%	-		AOI6	0.46%	1.01%	5.49%	0.87%	5.58%	-
	total number of transitions (Mean=77.66, SD=24.21)					total number of transitions (Mean=37.52, SD=15.38)									
r		AOI1	AOI2	AOI3	AOI4	AOI5	AOI6			AOI1	AOI2	AOI3	AOI4	AOI5	AOI6
VAF	AOI1	_	10.77%	1.55%	5.50%	2.17%	0.38%		AOI1	_	8.12%	1.45%	3.39%	1.33%	0.48%
ized	AOI2	10.01%	-	7.37%	0.91%	3.86%	0.70%		AOI2	7.45%	-	6.84%	0.73%	4.18%	1.27%
Individual	AOI3	1.26%	6.00%	-	0.79%	0.88%	5.59%		AOI3	1.21%	5.88%	-	0.79%	1.03%	7.75%
	AOI4	5.74%	0.88%	0.26%	-	5.03%	0.64%		AOI4	3.94%	0.67%	0.36%	-	4.85%	0.85%
	AOI5	2.93%	4.77%	0.70%	4.65%	-	4.77%		AOI5	1.70%	4.24%	0.67%	4.97%	-	7.69%
	AOI6	0.59%	0.67%	4.77%	0.97%	4.86%	-		AOI6	0.73%	1.33%	7.15%	0.97%	8.00%	-

Figure 6.9.: Transition matrix of the users in both groups.

Figure 6.9 shows that in the first visit, both groups had similar investigations by focusing mostly on AOI-1 and AOI-2. However, comparing the transition matrix in the first and revisit phase shows that the users with individualized VAF changed their strategy and investigated the relationship between AOIs on the right side of the dashboard. For this group, the transitions between AOI-5 and AOI-6 have the highest value, while for the general VAF group the transitions between AOI-1, and AOI-2 remain as the highest value. Comparing the heatmaps with the transition matrixes indicates that the users with individualized VAF not only had higher ARA on previously low attended AOIs also investigated the relationships between them more. Also, users in the general VAF group repeated their ARA and investigated the relationship between them instead of focusing on investigating new relationships.

As discussed in Section 6.5.4, the total number of transitions in each phase represents the ASR of the user in that phase. As the time for the revisit phase (1 minute) was lower than the first visit phase (2 minutes), the total number of transitions is lower in the revisit than the first visit phase for both groups. Figure 6.10 shows the amount of ASR for each group before and after VAF. For comparing the ASR in the first visit phase, I conducted an independent t-test between individualized VAF (M=77.66, SD=24.21) and general VAF (M=83.56, SD=21.66) groups that do not show any significant difference t(86.59)=-1.22, p=0.22. Therefore, I argue that both groups had the same ASR in the first visit. However, in the revisit phase, the results from the wilcoxon rank-sum test indicate that users with individualized VAF (Mdn=35) had significantly lower ASR than the users with general VAF (Mdn=45), W= 661.5, p=0.002, r=-0.321. Thus, the analysis of the empirical data also supports H2.



Figure 6.10.: ASR of users before and after receiving VAF types.

6.6.4. Attentional Resource Management

Figure 6.11 indicates the interaction plot for fixation duration and the number of fixations that shows how the ARM of the users changed during the experiment. As can be seen, the ARM of users' with individualized VAF improved massively while it is not the case for users with general VAF (the lower standard deviation values among six AOIs represent better ARM). Furthermore, the amount of ARM based on fixation duration and number of fixations for both groups can be seen in Appendix D and Figures D.7 and D.8



Figure 6.11.: Interaction effect of groups and phases on ARM performance.

To compare the ARM, I first conducted the wilcoxon's rank-sum test to investigate differences between two conditions at the end of the first visit (between-subject analysis). As can be seen in Table 6.3, there is no difference in the ARM of the users between two groups at the end of the first visit phase for both fixation duration (p=0.77) and the number of fixations (p=0.14). This is aligned with the previous findings that users of both groups have the same visual behavior in the first visit. However, the results show significant difference for both fixation duration (p=0.01) and the number of fixations (p=0.03) at the end of the task. Second, I investigated ARM by comparing each group in two phases (within-subject analysis). The results from wilcoxon signed-rank test show that the ARM of users with individualized VAF for both fixation duration (p<.001) and number of fixations (p<.001) differ significantly. However, the general VAF did not support users to improve their ARM significantly.

The findings from within and between analysis of the ARM show that the users with individualized VAF had better ARM than the users with general VAF at the end of the data exploration task. Thus, H3 is supported.

Between Subject Analysis								
DV	Exp. Phase	Median	W	Р	R			
	End of the First	General VAF	5.4	1010	0.77	0.020		
ARM performance	Phase	Individualized VAF	5.37	1019		-0.029		
(based on fixation durations)	End of the	General VAF	5.33	1964	0.015*	0.051		
	Task	Individualized VAF	4.14	1364		-0.251		
	End of the First	General VAF	4.77	000	0.143	-0.15		
ARM performance	Phase	Individualized VAF	5.23	868				
(based on No. of fixations)	End of the Task	General VAF	4.3	1000	0.037*	-0.216		
		Individualized VAF	3.56	1322				
Within Subject Analysis								
DV	Conditions	Exp. Phase	Median	\mathbf{V}	Р	R		
	C IVAE	End of the First Phase	5.4	754	.089 <.001 ***	0.172		
ARM performance	General VAr	End of the Task	5.33	704		-0.175		
(based on fixation durations)	Individualized VAF	End of the First Phase	5.37	207		0.409		
		End of the Task	4.14	807		-0.403		
ARM performance (based on No. of fixations)	General VAF	End of the First Phase	4.77	707	.066	0.107		
		End of the Task	4.33	101		-0.187		
	Individualized	End of the First Phase	5.23	0.60	<.001	0.400		
	VAF	End of the Task	3.56	869	***	-0.498		
Note: * p<0.05, **p<0.01, ***p<0.001								

Table 6.3.: Comparing ARM performance of the users in both groups.

6.7. Discussion

The second design cycle of the DSR project in this thesis includes several steps that discussed in previous sections. Based on that, I first derived MRs and DPs from existing prescriptive knowledge in the AUI field besides descriptive knowledge described in attention and HIP theories explained in Chapter 2, Section 2.2. These DPs focused on supporting users in managing goal-directed attention that refers to the voluntary type of attention (Corbetta and Shulman, 2002). Furthermore, these DPs focus on providing individualized VAF based on overt attention, which can track through an extrinsic behavior such as eye movements (Posner, 1980). Also, the individualized VAF can increase the awareness of the users about previous attention based on the eye-mind assumption that indicates where users are fixating is underlying their cognitive process, such as allocating their attention (Just and Carpenter, 1980). Furthermore, I presented the system architecture to map DPs to design features and provide the fundamentals for the development section. This information can help researchers and practitioners develop attention information dashboards that support data exploration with individualized VAF. Later, the identified key constructs leveraged in this study to measure users' ability to manage limited attentional resources. For that, I focused on measuring the attention of users before and after receiving the individualized VAF as well as at the end of the task. First, I measured ARA in the revisit phase that examines the ability to ignore high-visited charts previously and allocating attention to the previously low-visited charts. I also measured the ASR that investigates the ability to select limited charts to process in the revisit phase rather than allocating attention to many charts. Besides, I measured users' ARM at the end of the task that explains their ability to allocate attention to all dedicated charts on the dashboard properly. Next, I formulated three hypotheses for evaluating the effects of the DP1&2 on the ability to manage limited attentional resources. I tested these hypotheses based on analyzing user's eye movement data. The findings described in the previous section show that the proposed DPs and their instantiation in a software artifact increase ARA and ASR performance in the revisit phase (H1&H2) and ARM performance at the end of the task (H3).

All in all, I can argue that the findings of this experiment reveal that the suggested solution (attentive information dashboards with individualized VAF) to support data exploration tasks on information dashboards help users in managing limited attentional resources. Based on the results from ARA, ASR, and ARM, the users with individualized VAF have a higher level of awareness about their previous visual behavior in allocating attention and could plan to improve it in the revisit phases.

6.8. Summary

The second design cycle of this DSR project is motivated by users' challenges in managing limited attention while exploring information dashboards. I provide a solution for that following the DSR paradigm and articulate theoretically grounded DPs for designing innovative software artifacts, attentive information dashboards for data exploration task. This software artifact tracks users' eye movement data in real-time and provides individualized VAF. Based on Gregor (2006), the proposed MRs and DPs are considered as type V theory contribution (Design and Action), since it gives explicit prescriptions for constructing attentive information dashboards that provide individualized VAF for data exploration tasks. Furthermore, vom Brocke et al. (2013) have emphasized that there are limited contributions in the DSR community that make actual use of the potential of neuroscience tools (e.g., eye tracker) to design advanced built-in functions for IT artifacts. To the best of my knowledge, this design cycle and the proposed individualized VAF is the first project that investigates the integration of real-time eye movement data as a built-in function for IT artifact, in this case, information dashboards.

Furthermore, I evaluated the proposed DPs in an eye tracking laboratory experiment with 92 participants. The findings reveal the positive effect of using individualized VAF on managing limited attentional resources focusing on ARA, ASR, and ARM. The findings described in the previous section show that the proposed DPs and their instantiation in a software artifact increase ARA and ASR performance in the revisit phase (H1&H2) and

ARM performance at the end of the task (H3). It shows that when an information dashboard features individualized VAF, users can manage their limited attentional resources in a better way in comparison with users without such support. Based on Gregor (2006), this contribution is considered as II theory (Explanation) since it provides explanations about the effects of using DP1&2 on enhancing users ability to manage their limited attentional resource during data exploration tasks. According to the DSR contribution framework by Gregor and Hevner (2013), the proposed attentive information dashboards to support data exploration is an improvement since I successfully developed a new solution (DP1&2) to the existing problem (managing limited attentional resources).

7. Design Cycle 3: Attentive Information Dashboards with Task Resumption Support¹

7.1. Study V: Overview

In the age of information, it is hard to find someone that is not challenged by interruptions. During the last years, many observational studies identified the disruptive role of interruptions in daily life (Borst et al., 2015). The workplace is also accompanied by frequent interruptions of task execution (Czerwinski et al., 2004; Mark et al., 2008). Interruptions are known to have various negative impacts, such as higher task completion time, the number of errors, or anxiety in multitasking situations (Borst et al., 2015,1; Czerwinski et al., 2000; O'Conaill and Frohlich, 1995). Even though task disruptions can sometimes be unavoidable, e.g., due to the importance of the secondary task or the specific context of work (Dostal et al., 2013), it is necessary to design systems that provide advanced support in better managing interruptions (Anderson et al., 2018).

In the domain of HCI, research on interruptions and task resumption has a long tradition. Several studies focused on explaining how users are coping with that (Borst et al., 2015; McFarlane and Latorella, 2002). In a multitasking environment, users need to shift their attention from the primary task to the secondary task and allocate their attentional resources to the execution of the secondary task. Afterward, users need to shift their attention to the primary task once again, which is sometimes challenging due to remembering previous behavior and resuming the primary task properly. Therefore, capturing and remembering representations of tasks may be useful to assist users in switching among them. A promising solution supporting users in better managing interrupted tasks are AUIs (Anderson et al., 2018; Bailey and Konstan, 2006). Leveraging such a system to detect an interruption and provide TRS by highlighting previously attended areas proved to support users in managing attention (Göbel and Kiefer, 2019; Jo et al., 2015; Kern et al., 2010; Mariakakis et al., 2015). Eye tracking technology is known as a tool to provide TRS. Researchers have used this technology to track the user's eye movements to understand the moment of task switching (Chen et al., 2013) and subsequently visualize the user's eye movement data as an indicator of the most recent attention area (Dostal et al., 2013; Jo et al., 2015; Kern et al., 2010; Mariakakis et al., 2015). Gaze-based TRS makes previous cognitive processes more explicit by providing memory aids. It gives hints to the users to remember what they were thinking or what might have been their intention before shifting to the interruption task (Majaranta and Bulling, 2014). Furthermore, in previous studies, the most common highlighting method for gaze-based TRS was the last point highlighting as a placeholder (Jo et al., 2015; Kern et al., 2010; Mariakakis et al., 2015). Moreover, heatmap and scanpath are standard visualization methods of eve movement data to un-

¹This Chapter is based on the following studies which are published or in work: Toreini et al. (2018c), Toreini et al. (2018b), Toreini et al. (2020a)

derstand the previous visual behavior of users. However, using these visualizations has not been investigated so far as highlighting methods for gaze-based TRS.

Following the findings from Study II, the information dashboard users also have difficulty in resuming an interrupted task, and there is a need to support them. This is considered as MR5 in the identified MRs for designing innovative information dashboards. Also, as discussed in Chapter 1, Section 1.2 and findings from Study I, investigating how to design gaze-based TRS for information dashboards is a research gap. Therefore, I focus on closing this gap and address the first part of the fifth RQ in this thesis:

RQ5a: How to design attentive information dashboards providing individualized VAF to enhance users' ability to manage attentional resources in resuming interrupted tasks?

The impact of interruptions varies and depends on individual user characteristics. Previous studies showed that user characteristics such as WMC play an essential role in managing interruptions (Cane et al., 2012; Foroughi et al., 2016; Mark et al., 2008; Ratwani and Trafton, 2008; Werner et al., 2011). Users with higher WMC can better remember their previous visual behavior and thus can better deal with task resumptions. However, as discussed in Chapter 1, Section 1.2 and the results from Study I shows that investigating the impact of gaze-based TRS under consideration of WMC is a research gap in this field. Also, as the suggested MR6 from Study 2, the provided feedback to support users should consider the WMC of the users. Therefore, I focus on closing this gap and address the second part of the fifth RQ in this thesis:

RQ5b: What is the role of working memory capacity in effectiveness of gazebased task resumption supports?

As Figure 7.1 shows, this study focuses on the third design cycle of the DSR project. First, I review related work on designing gaze-based TRS and conceptualize it by identifying different dimensions that impact designing effective gaze-based TRS. Second, I present MRs and derive DPs for designing attentive information dashboards that support users' task resumption after interruptions. Third, I discuss the development step by presenting the system architecture and different highlighting methods suggested for gaze-based TRS. Fourth, I test the DPs in an exploratory laboratory experiment by investigating the role of WMC on the impacts of gaze-based TRS. Fifth, I present the results from the experiment and discuss different highlighting methods based on the role of WMC.

General Design Science Cycleexploring attention management problems with dashboards and possible solutionsattentive information dashboards for data explorationattentive information dashboards with task resumption supportAwareness of Problemliterature review & problem exploratory eye tracking studyfurther reading and refinement of theoretical groundingliterature review on attentive systems with task resumption support (TRS)Suggestionprovide suggestions based on results from literature review and exploratory studyadaptation of DPs based on empirical results and theoretical foundationsadaptation of DPs based on empirical results and theoretical foundationsDevelopmentinstantiation of suggestions as basic: • attentive dashboard • three VAF typesinstantiation of DPs as: • attentive dashboard • individualized VAF (gaze-based TRS)instantiation of gaze-based TRS)Evaluationquantitative evaluation of vAF approaches (real-time Vs. off-line) (eye tracking pilot study)quantitative evaluation of individualized VAF for data exploration (lab experiment)quantitative evaluation of adaptace of most suitable VAF typeevaluation analysis and identification of most suitable VAF typeevaluation analysis, hypothesis supportedevaluation analysis and identification of most suitable VAF type			Cycle 1	Cycle 2	Cycle 3			
Awareness of Problemliterature review & problem exploration through exploratory eye tracking studyfurther reading and refinement of theoretical groundingliterature review on attentive systems with task resumption support (TRS)Suggestionprovide suggestions based on results from literature review and exploratory studyadaptation of DPs based on empirical results and theoretical foundationsadaptation of DPs based on empirical results and theoretical foundationsDevelopmentinstantiation of suggestions as basic: • attentive dashboard • three VAF typesinstantiation of DPs as: • attentive dashboard • individualized VAF (gaze-based TRS)instantiation of gaze-based TRS and the role of WMC (lab experiment)Conclusionevaluation analysis and identification of most suitable VAF typeevaluation analysis, hypothesis supportedevaluation analysis and identification of most suitable VAF type	Operation and Goal Knowledge	General Design Science Cycle	exploring attention management problems with dashboards and possible solutions	attentive information dashboards for data exploration	attentive information dashboards with task resumption support			
Oppositeprovide suggestions based on results from literature review and exploratory studyadaptation of DPs based on empirical results and theoretical foundationsDevelopmentinstantiation of suggestions as basic: • attentive dashboard • three VAF typesinstantiation of DPs as: • attentive dashboard • individualized VAF (gaze-based TRS)instantiation of DPs as: • attentive dashboard • individualized VAF (gaze-based TRS)Evaluationquantitative evaluation of VAF approaches (real-time Vs. off-line) (eye tracking pilot study)quantitative evaluation of nidividualized VAF for data exploration (lab experiment)quantitative evaluation of suitable VAF typeConclusionevaluation analysis and identification of most suitable VAF typeevaluation analysis, hypothesis supportedevaluation analysis and identification of most suitable VAF type		Awareness of Problem	literature review & problem exploration through exploratory eye tracking study	further reading and refinement of theoretical grounding	literature review on attentive systems with task resumption support (TRS)			
Off pointinstantiation of suggestions as basic: • attentive dashboard • three VAF typesinstantiation of DPs as: • attentive dashboard • individualized VAF (gaze-based TRS)Evaluationquantitative evaluation of VAF approaches (real-time Vs. off-line) (eye tracking pilot study)quantitative evaluation of individualized VAF for data exploration (lab experiment)quantitative evaluation of gaze-based TRS and the role of WMC (lab experiment)Conclusionevaluation analysis and identification of most suitable VAF typeevaluation analysis, hypothesis supportedevaluation analysis, based on WMC		Suggestion	provide suggestions based on results from literature review and exploratory study	adaptation of DPs based on empirical results and theoretical foundations	adaptation of DPs based on empirical results and theoretical foundations			
Evaluation quantitative evaluation of VAF approaches (real-time Vs. off-line) (eye tracking pilot study) quantitative evaluation of individualized VAF for data exploration (lab experiment) quantitative evaluation of individualized VAF for data exploration (lab experiment) Conclusion evaluation analysis and identification of most suitable VAF type evaluation analysis, hypothesis supported evaluation analysis, and identification of most suitable gaze-based TRS based on WMC		Development	instantiation of suggestions as basic: • attentive dashboard • three VAF types	instantiation of DPs as: • attentive dashboard • individualized VAF	 instantiation of DPs as: attentive dashboard individualized VAF (gaze-based TRS) 			
Conclusionevaluation analysis and identification of most suitable VAF typeevaluation analysis, hypothesis supportedevaluation analysis, identification of most suitable gaze-based TRS based on WMCnascent design theory		Evaluation	quantitative evaluation of VAF approaches (real-time Vs. off-line) (eye tracking pilot study)	quantitative evaluation of individualized VAF for data exploration (lab experiment)	quantitative evaluation of gaze-based TRS and the role of WMC (lab experiment)			
nascent design theory		Conclusion	evaluation analysis and identification of most suitable VAF type	evaluation analysis, hypothesis supported	evaluation analysis and identification of most suitable gaze-based TRS based on WMC			
			nascent design theory					

Figure 7.1.: The focus of Study V in this DSR project.

7.2. Background and Conceptualization

This section first introduces related work for TRS studies that used eye tracking technology. Furthermore, I provide a conceptualization of gaze-based TRS and investigate different dimensions that are needed to design an effective gaze-based TRS.

Gaze-based TRS and Highlighting Methods

The generic task interruption process was initially conceptualized by the "Memory for Goals" theory introduced by (Altmann and Trafton, 2002). Based on this theory, each task comes with a goal. When a primary task is interrupted, the goal is stored in the user's memory. After finishing the secondary task, the primary task needs to be resumed, which means the goal needs to be retrieved from the memory. This step is known to be difficult for users and takes dedicated time. A phenomenon that can be experienced in a multitasking environment is work fragmentation and difficulty in retrieving information from memory to resume tasks. Thus, the user's memory is under the competition of several goals at once and is challenged by recalling specific goals for tasks in an efficient way.

Gaze-based TRS applications use the information gained from the user's attention to assist the resumption of an interrupted task. Visual attention is inferred by an eye tracker or a camera system that records the user's eye movements and presence (Carrasco, 2011; Kowler, 2011). The examination of fixations on certain UI elements helps the system to understand how users' attention is distributed between the elements. TRS applications provide feedback that highlights certain points or areas of the previous high visual interest of the user to assist the user in resuming an interrupted task. These systems have shown to be supportive during task resumption as they decrease the resumption lag and increase the overall individual performance (Cheng et al., 2018; Jo et al., 2015; Kern et al., 2010; Mariakakis et al., 2015; Taylor et al., 2015).

For example, Kern et al. (2010) developed a gaze-based TRS application called Gazemarks that delivers a visual placeholder to ease attention switching between two main tasks in a multi-monitor setup. The authors of this study used last fixation point visualization as a visual placeholder. Jo et al. (2015) developed EyeBookmark in which they used eye trackers to support users recovering from the last reading position. They developed four different highlighting methods and investigated their effect on the resumption lag and the reading performance. The first highlighting method was the last fixation point visualization in which they highlighted the last read word in the text. Second, they highlighted a block that includes a fixed number of lines in addition to the last fixated word. Third, they highlighted the whole sentence in which the user was interrupted. Finally, they highlighted the previous sentence before the interruption as a reminder for the reader. In another study, Mariakakis et al. (2015) developed SwitchBack to support reading a text on mobile devices with smaller screen sizes. This TRS also guides the user back to the appropriate region by highlighting the last row of the text that was read before the external distraction. Also, Taylor et al. (2015) developed EyeFrame as a system that used the collected eye movement data in previously high-visited AOIs and drew a pale red line around them as a memory aid. Also, Cheng et al. (2018) developed Smooth Gaze in which the last fixation visualization was used as a highlighting method to support task transfer and recovery across devices when users needed to shift their attention among them.

The last fixated point highlighting method is the most common highlighting method in previously developed gaze-based TRS. Furthermore, highlighting borders of high-visited AOIs was used as a gaze-based TRS. However, the standard visualization techniques for eye tracking data such as heatmap and scanpath are not investigated so far in the context of gaze-based TRS. These common visualization techniques represent the summary of attention allocation and attention shifts of the users while visually process some information and can increase their self-awareness. Furthermore, there is a lack of research comparing these highlighting methods with each other. Comparing different highlighting methods as gaze-based support tools are tested in collaboration tasks (D'Angelo and Gergle, 2018) or online strategic games (Newn et al., 2017). To close this gap, I focus on comparing the last point, heatmap, and scanpath highlighting method for supporting task resumption in this study.

115

Gaze-based TRS Conceptualization

Figure 7.2 depicts the four main dimensions I identified for designing effective gaze-based TRS by analyzing existing research in the field of interruptions and gaze-based TRS. In this study, I tried to control for the task, context, and interruption characteristics and focused on different user characteristics.



Figure 7.2.: Conceptualization of gaze-based TRS.

The first dimension of the gaze-based TRS concept is the type of primary task in which interruptions are disrupting. Previous eye tracking studies showed that the given task affects users' eye movement patterns (Yarbus, 1967). Therefore, to find the proper highlighting method for gaze-based TRS, it is necessary to know the type of task, its complexity, priority, being an individual or group task, etc. influence the design of the gaze-based TRS. In previous studies, reading tasks (Cheng et al., 2018; Jo et al., 2015; Mariakakis et al., 2015), monitoring task (Taylor et al., 2015), browsing (Kern et al., 2010) received attention by researchers. However, I believe that it is necessary to study gaze-based TRS in more diverse tasks.

Additionally, context and task are known as two elements that guide the design of any interactive system (Benyon, 2013). The environmental states are considered as the physical context. Previous studies in this field have focused on the situation in which users are working with digital applications. (Kern et al., 2010) tested a gaze-based TRS system in driving simulators and checking the navigation system, whereas (Taylor et al., 2015) used TRS in a gaming application. (Mariakakis et al., 2015) tested TRS in a mobile application, and (Cheng et al., 2018) examined their gaze-based TRS in reading applications. More contexts such as social aspects, another type of digital application, real-world interactions with augmented reality or glasses can be further options for investigation.

The third dimension is the characteristics of the interruption as a secondary task, which can be seen by interruption's source, duration, and its relevance. Generally, there are two different types of interruptions with regards to their source (Dabbish et al., 2011). First, external interruptions are triggered by events in the environment, like a phone ringing, IT-mediated interruptions such as notifications from groupware or instant messaging tools or short talks with colleagues. Second, internal interruptions are caused by internal reasons related to attentional shifts, such as thinking about vacations, shopping or past events. Besides the source, the duration of interruption also affects the task resumption. A longer and more demanding interruption leads to a slower task resumption (Monk et al., 2004,0). Also, the moment in which interruptions happens have different influences within task execution (Adamczyk and Bailey, 2004). Finally, task-relevant interruptions have been identified as an improvement of user's performance, while irrelevant interruptions are known as disruptive (Stothart et al., 2015).

The fourth dimension is the user characteristics that influence their ability to manage the interruption and resumption. Based on user characteristics, the type of gaze-based TRS might be different. User demographics such as age, gender, expertise level, and WMC are examples of user characteristics. Although previous research examined the role user characteristics on handling interruptions (Cane et al., 2012; Foroughi et al., 2016; Mark et al., 2008; Ratwani and Trafton, 2008; Werner et al., 2011), this is rarely investigated in the context of TRS. For example, user-specific reading behavior can enhance the detection of mind wandering and the reduction of internal task disruption in the process (Bixler and D'Mello, 2016). To the best of my knowledge, none of the existing studies investigated the role of individual characteristics in the context of gaze-based TRS. It is not clear if gaze-based TRS should be designed for all users in the same way or if individualized gaze-based TRS would positively impact user performance.

As an individual characteristic for this study, I emphasize WMC because of the following reasons: First, it plays a vital role in complex cognitive tasks, such as comprehension, reasoning, and problem-solving (Engle, 2002). Besides, theoretical frameworks about interruption emphasize the role of working memory in handling interruptions (Altmann and Trafton, 2002; Borst et al., 2015). Also, it has been shown that users with different WMC can handle interruptions in different ways (Cane et al., 2012; Foroughi et al., 2016; Ratwani and Trafton, 2008; Werner et al., 2011). Moreover, it is considered as an essential individual characteristic while users are working with visualized information such as on information dashboards (Haroz and Whitney, 2012; Healey and Enns, 2012; Toker et al., 2013). I focus particularly on the visuospatial WMC of the user since it reports how well the user can reproduce a sequence of locations. I assume that when users are exploring information dashboards, they read a sequence of AOIs. The ability to remember their previous exploration sequence can help them to have a better resumption performance.

117

7.3. Meta-requirements and Design Principles

Deriving MRs for this study is based on the **initial MR5** found in Study II. Based on that, an innovative information dashboard should support users to resume an interrupted task. In this study, I identify refined MRs for this initial MR. Furthermore, I integrate MRs provided in Study IV to design attentive information dashboards for data exploration and refined them to derived MRs for attentive information dashboards with TRS.

To ease the MRs description, I first provide an example scenario representing the steps for the interruption in Figure 7.3. In this scenario, I assume that a user works with an information dashboard and is busy with a primary task. Later, a team member opens the door and asks for a request, which results in an attention shift from the primary task to the secondary task (in this example, external interruption happens, however, other types of interruption such as IT-mediated and internal interruptions can be the case). After finishing the interruption task, the user of the information dashboard returns to the primary task. This is the moment that can get the benefit of a memory aid that supports handling the resumption performance.



Figure 7.3.: Example scenario to represent steps for providing TRS.

Processing visualized information of information dashboards is done through visual perception. Eye movement data is known as a resource to analyze users' cognitive processes (Hayhoe and Ballard, 2005; Kowler, 2011; Liversedge and Findlay, 2000; Rayner, 1998). Furthermore, eye tracking technology is seen as the leading technology for designing attention support systems (Bulling, 2016). Therefore, as the **MR1**, attentive information dashboards should be able to track user's eye movement data in real-time while processing an information dashboard. Nowadays, employees are often distracted by interruptions originating from different sources such as IT-mediated notifications, phone calls, or colleagues (Czerwinski et al., 2004; Mark et al., 2008). These type of interruption that has a source from external element referred to as external interruptions. Although some of those interruptions have a positive effect on the performance, some have negative effects or both (Addas and Pinsonneault, 2015). In the case of interruption, users need to shift their attention from primary to secondary tasks. Eye movements are effective in measuring the user's attention status and infer the user's intention, e.g., shift their attention (Carrasco, 2011; Hayhoe and Ballard, 2005; Liversedge and Findlay, 2000; Orquin and Mueller Loose, 2013). Therefore, as the **MR2**, attentive information dashboards should be able to estimate the user's attention based on the tracked eye movement data in real-time.

Interruptions are provoking a shift of attention from the primary task to the secondary task (Speier et al., 1999). Recognizing attention shifts due to the external interruption is essential for designing TRS (Altmann and Trafton, 2004; Roda, 2011). This recognition can be by tracking users' interaction data or users' cognitive state through external devices such as eye tracking. Therefore, as the **MR3**, attentive information dashboards should be able to recognize the occurrence of external interruptions. Furthermore, to assist the user in resuming the primary task, the end of the secondary task must be recognized. This can be identified by the start of resuming the primary task. Recognizing a resumption means that the eye tracker detects the user's eye movements upon returning to the application for the primary task (Mariakakis et al., 2015). Therefore, as the **MR4**, attentive information dashboards should be able to recognize the occurrence of task resumptions.

Studies show that interrupted primary tasks are not resumed right away (O'Conaill and Frohlich, 1995). When starting the resumption of the primary task, users tend to think again about their main goal and the amount of the task they did so far. Previous studies have shown that providing support during this recovery process helps the users' performance and reduces the resumption lag (Jo et al., 2015; Mariakakis et al., 2015). Moreover, supporting users with task interruption is known as one of the critical issues in the IS community (Addas, 2010). Therefore, as the MR5, attentive information dashboards should be able to support the users' recovery process. In addition to that, studies show that task resumption is primary a memory-based process (Altmann and Trafton, 2002) and some other general human cognitive, perceptual, and motor processes (Salvucci, 2010). Resuming the primary task comprises remembering to restart the interrupted task and restoring the context of the primary task (Bailey and Konstan, 2006; Cane et al., 2012). Previous studies have shown that providing gaze-based TRS by highlighting the last visual behavior is supportive for users' task resumption performance (Jo et al., 2015; Kern et al., 2010; Mariakakis et al., 2015). Providing gaze-based TRS gives a hint about what was in the visual interest of them before the interruption. This information is assisting users back to the relevant points again after facing the interruption. Therefore, as the **MR6**, attentive information dashboards should provide the previous visual behavior of the users as the support for the users' recovery process.

Based on the model of memory for goal (Altmann and Trafton, 2002), each task has a goal, while users need to allocate attentional resources to conduct their tasks. Furthermore, the goal is associated with the users' eye movements (Ratwani and Trafton, 2010). Attention-aware systems are known as a type of systems that support users in different phases of interruptions (Bailey and Konstan, 2006). Existing research shows that attention-aware systems assist the user by integrating the eye movement data to infer the user's visual attention before providing TRS (Jo et al., 2015; Kern et al., 2010; Mariakakis et al., 2015)). As discussed, tracking the users' eye movement data and extract attentional states is needed for such a system (MR1, MR2). Therefore, I suggest the first DP (DP1) as providing the information dashboard with eye tracking technology to track users' attention allocation.

In order to provide TRS, there is is a need to detect the interruption as well as the resumption (MR3, MR4). Although these interruptions can have different resources, this research is focused on external interruptions and detecting attention shifts in real-time. Traditionally, mouse and keyboard actions are used to measure the interruption and resumption lag (Adamczyk and Bailey, 2004; Altmann and Trafton, 2004; Iqbal and Horvitz, 2007) or detecting strategies for task resumption (Dragunov et al., 2005). Lately, users' ability to manage the interruptions was measured with other resources such as psycho-physiological sensors (Züger and Fritz, 2015) and eye tracking technology (Cane et al., 2012). Therefore, I suggest the second DP (**DP2**) as provide the information dashboard with the ability to identify attention shifts to the secondary task and resumption of the primary task.

Moreover, the need for interfaces that aid task resumption was discussed by researchers (McFarlane and Latorella, 2002). AUIs use gaze-based TRS to assist the user after an interruption such as highlighting either by color or a spotlight the last element that the user was looking at before the interruption (Jo et al., 2015; Kern et al., 2010; Mariakakis et al., 2015)). Kern et al. (2010) have stated that providing gaze-based TRS after task resumption might also be a potential benefit in standard workplaces where interruptions might be longer. Czerwinski et al. (2004) found that approaches capturing and remembering representations of tasks may be useful to assist users in switching among tasks. Therefore, based on MR5 and MR6, I suggest the third DP (DP3) as provide the information dashboard with gaze-based TRS in order to support users to resume their primary task.

Table 7.1 shows the summary of six identified MRs and three derived DPs for designing attentive information dashboards that support users in task resumption.

120

Design Cycle 1	Design Cycle 3				
Initial MR	Refined MRs	DPs			
	Refined MR1: The information dashboard should be able to track user's eye movement data in real- time with an eye tracking device while processing information dashboard.	DP1: Provide the information dashboard with eye tracking tashpalage in order to track the user's			
Initial MR: The information dashboard should be able to support users to select a proper revisit strategy that helps resuming the interrupted task.	Refined MR2: The information dashboard should be able to estimate the user's attention based on the tracked eye movement in real-time.	attention allocation.			
	Refined MR3: The information dashboard should be able to recognize the occurrence of external interruptions.	DP2: Provide the information dashboard with the ability to identify			
	Refined MR4: The information dashboard should be able to recognize the occurrence of task resumptions.	attention shifts to secondary task and resumption of primary task.			
	Refined MR5: The information dashboard should be able to recognize the occurrence of task resumptions.	DP3: Provide the information			
	Refined MR6: The information dashboard should provide the previous visual behaviour of the users as the support for support the users' recovery process.	order to support users to resume their primary task.			

Table 7.1.: Summary of MRs and DPs for designing attentive information dashboards that support users' task resumption.

7.4. Development

System Architecture

To map the DPs to design features, I propose system architecture as can be seen in Figure 7.4. This architecture includes four subsystems which are structured along with the three identified DPs discussed in the previous section.

The first subsystem is **Information Dashboard Subsystem** that involves the BI&A system and information dashboard layout. The data referring to the information dashboard is transferred to both interruption handling and attention-aware subsystems for further process. The second subsystem is **Eye Tracking Subsystem** that enables recording the eye movement data with eye tracking technology. This subsystem covers the DP1 proposed in the previous section. Later the collected eye movement data is processed, and the visual attention of the users is detected. This information is transferred to the interruption handling and attention-aware subsystems as well.

The third part is **Interruption Handling Subsystem** which represents the DP2 and includes three components. The first component, "Dashboard Status Identifier", continually checks whether the information dashboard is the primary running software on the user's computer or other software is activated. In case it is not the main running software, the system considers that an it-mediated interruption happened and communicates the status to the interruption analyzer component. The IT-mediated interruption can be both relevant or irrelevant; in all cases, this is an indicator that the users shift the attention to the secondary task. The second component is the "User Status Identifier" that tracks the external interruption of users through eye movement data and identifying the users' presence. This component considers the moment that the user is not looking at the monitor as the occurrence of an external interruption. Both components are connected to the "Interruption Analyzer" component that distinguishes the start and the end of the second task. The end of the second task is assumed to be the start of the primary task.

The fourth subsystem is **Attention-aware Subsystem** that represents the DP3 and includes two components. "Attention Analyzer" component receives eye movement data as well as dashboard layout information and discovers the user's attentional spotlight on the dashboard. Later this information is used to generate gaze-based TRS as memory aid feature by "Feedback Generator" component. This component is responsible for providing the proper highlighting method as gaze-based TRS.



Figure 7.4.: System architecture for designing attentive information dashboards that support resuming interrupted tasks.

Highlighting Methods

I leverage three different highlighting methods, including the last point, heatmap, and scanpath in this study, as depicted in Figure 7.5. In all cases, the dashboard is blurred, and the previous visual behavior of the users is highlighted. In previous studies, the last point highlighting method is the most common method for gaze-based TRS. Besides the last point, users' attention can also be visualized with the two established eye movements visualization methods (Blascheck et al., 2014): heatmap, and scanpath. Heatmaps visualize how long a user looked at a certain point compared to the rest of the screen. Presenting such a highlighting method as gaze-based TRS can support users to recall their previous attention allocation on the AOIs and not only the last position. Moreover, the scanpath method shows the user's gaze path, including fixations on certain points of the screen. This may help perform the data exploration task on information dashboards since users need to jump between the different information pieces to get a comprehensive overview of the status quo. To extract fixations and to visualize the heatmap and scanpath, I used the PyGaze (Dalmaijer et al., 2014) as open-source toolbox. To present the last point, I used a circle shape. The center of this circle is calculated based on the mean value for the x and y of the last two fixations and the radius of 50 pixel with red color as the border.



Figure 7.5.: Three suggestions for DP3 as gaze-based TRS.

Information Dashboards

For this study, I designed a simple static information dashboard and control for elements that influence stimulus-driven attention, such as color-coding, size, chart type. The dashboard comprises six bar graphs, which are defined as an AOI. Furthermore, each graph has six chunks of information, and all elements are in gray. Therefore, I argue that the six AOIs of the dashboard have the same complexity regarding the design. I designed four dashboards with such design for this experiment but with changing the content of the graphs (sales, marketing, customer service, and human resource). The participants received these dashboards in random order during the experiment. An example of the dashboard design can be seen in Figure 7.6. These information dashboards with the actual content can be seen in Appendix E and Figures E.1, E.2, E.3, E.4. As discussed in Chapter 2, Section 2.2.1, this dashboard layout supports controlling stimulus-driven attention and focuses on the user's goal-directed attention. Furthermore, I focus on the overt attention of the users by tracking their eye movement data.



Figure 7.6.: Example of the information dashboard used in Study V.

7.5. Laboratory Experiment

7.5.1. Experimental Software and the Apparatus

Figure 7.7 shows the different stages of the task, interruption, and resumption process with the gaze-based TRS that I adapted from (Trafton et al., 2003). The first stage is the primary task that the user is diligent. The usage of visualized information in the form of an information dashboard increased these days (Yigitbasioglu and Velcu, 2012). Therefore, I picked a task on an information dashboard for this study, which was derived from the real-world workplaces. The user's task with such an application can be distinguished into search tasks and data exploration tasks (Vandenbosch and Huff, 1997). In a search task, the user tries to find information while in data exploration tasks, the user investigates all information provided on the dashboard to get a comprehensive understanding of the current status. In this study, I focus on data exploration. Also, an eye tracker records the user's eye movements to track the user's visual behavior while exploring the information dashboard. Also, to extract fixations, I used Tobii Pro SDK provided with Tobii 4C eye tracker and sensitive mode in recording fixations.

A few seconds after starting the primary task, I evoke a situation in which they face an interruption and need to shift their attention to the interrupting secondary task. In this study, the secondary task is also inspired by real-world workplaces. Here, I ask the user to answer some irrelevant emails as it-mediated interruptions. These emails are not relevant to the data exploration task and also need to be answered in a short time and do not require a long answer. Therefore, I consider them as short-term interruptions. Irrelevant and short-term interruptions are widespread and frequently happen in workplaces.

After ending the secondary task and upon resuming the primary task, the user receives one of the highlighting methods as gaze-based TRS. This gaze-based TRS presents the previously recorded eye movement data during the primary task. I assume that providing such a reminder about previous visual behavior supports users in having a better resumption performance, including a shorter resumption lag and avoiding repetitive behavior while continuing the data exploration task.



Figure 7.7.: Stages of interruption and resumption adapted from Trafton et al. (2003).

As an apparatus, I used a Tobii 4C eye tracker with a frequency of 90 Hz and the required license to record and store eye movement data for research purposes. So far, the high price of eye trackers is known as one of the obstructions to using this type of device for designing gaze-based TRS. However, Tobii 4C eye tracker is one of the low-cost eye trackers in the market (Farnsworth, 2019), and using it is in alignment with the goal of this thesis by bringing eye trackers to workplaces.

I apply a 2×4 experimental design with users' WMC (high, low) as between-subject factors and four highlighting methods for gaze-based TRS (last point, heatmap, scanpath) as within-subject factors. I fully counterbalanced the order of the 24 unique elements, which includes the primary task (four dashboard design), interruption as the secondary task (four types of emails), and TRS condition (three gaze-based TRS and one control). To randomize this order, I prepared two boxes before the experiment and assigned 24 unique orders on each box. Before the experiment, I asked participants to draw one item of the 24 unique elements randomly, and I insert them manually into the experimental software at the beginning of the experiment. Moreover, the experimental software randomly assigned the order of the secondary tasks for each participants.

At the beginning of the experiment, I calibrated the eye tracker using the 7-point calibration with the Tobii Pro Eye Tracker Manager. The participants then received screen-based introductions, which included explanations of the provided information dashboards and the data exploration task. Later, the experimental software introduced the scenario of this experiment. The participants were asked to imagine that they just joined a fictitious company as the executive assistant to the CEO. It is their first working day at the company, and in a few minutes, they will have a meeting with the CEO and several managers to discuss the current sales, marketing, human resource, and customer service performance of the company. In order to prepare and get ready for the meeting, the task was to explore the corresponding dashboards, which they received one by one in this experiment. These dashboards can be seen in Appendix E and Figures E.1, E.2, E.3, E.4. Besides this important primary task, they had to work on other small administrative tasks in parallel and received four emails regarding their onboarding process that needed to be replied to immediately. These emails can be seen in Appendix E and in Figures E.5, E.6, E.7, E.8. The flow of the main part can be seen in Figure 7.8. These steps were repeated four times with changing dashboards and emails, which were randomly assigned to the participants.



Figure 7.8.: The main steps of the experiment.

Before starting with the primary task, all users received a fixation cross on the bottom of the page outside the position of AOIs for a few seconds to control their gaze position and rest. The position for fixation cross can be seen in Appendix E in Figure E.9. As the first step, they received an information dashboard that marked the start of the primary task. After 30 seconds, the participants faced an interruption, and the secondary task phase is started. In this step, they received an email that they had to read and answer. The email was irrelevant to the data exploration task and only included some information about the onboarding process. All emails were of a similar length, and typing the answer required a similar amount of time. After sending the email, they received one of the TRS conditions. Three gaze-based TRS were designed for this experiment in addition to the without TRS condition as a control condition. In all TRS types, the users saw the TRS on top of a blurred dashboard with a similar layout. Therefore, the users did not have access to the content of the dashboard and could not continue the data exploration task while having the TRS. For the control condition, all users were forced to look at a empty screen (Appendix E Figure E.11) to wait for 10 seconds and then continued the experiment. Similar to the TRS cases, in this case, users did not have access to the dashboard content. Therefore I argue that the users in all conditions have a similar level of chance access the content of information dashboards during the experiment. After the TRS, they received a fixation cross outside the position of the six AOIs to control for the task resumption measurements. The position for the fixation cross is randomized and can be seen in Appendix E in Figure E.10.

In the revisit phase, they received the same dashboard as before the interruption to continue the data exploration. To finish the revisit phase, the participant had to press the finish button at the bottom of the dashboard. In this phase, all users were forced to explore the data for a minimum of 30 seconds and a maximum of four minutes. Therefore, the clicking function of the finish button just gets activated after 30 seconds. I chose 30 seconds as the minimum revisit period in order to have comparable time frames for the first visit and the revisit phase for all users. This procedure is repeated four times to explore the four dashboards (sales, marketing, social media, customer service, human resource).

After exploring the four dashboards, the participants answered a questionnaire about their demographics and a survey so they could rest for a few minutes. As the last step, they completed the Corsi Block-Tapping test from the PEBL software (Mueller and Piper, 2014) to measure their WMC.

7.5.3. Participants

To test the role of WMC on the gaze-based TRS, I conducted a controlled laboratory experiment. In total, 48 university students (22 female, 26 male) with an average age of 22.72 (SD=2.24) years participated. Using students fits the scenario that I examine in this study since they could imagine themselves in the defined role as a new employee in a company. Thirty-one of them did not use any visual aid during the experiment while 11 had glasses, and 6 had contact lenses as a visual aid. They received 10 Euro as compensation for their participation.

After completing the experiment, I asked the participants to rest for a few minutes, and then I analyzed their visuospatial WMC by conducting the visuospatial Corsi Block-Tapping test (Kessels et al., 2000) using PEBL test battery (Mueller and Piper, 2014). The Corsi span refers to the longest list of items that participants could memorize and repeat back in the correct order, and it is calculated by PEBL software. The higher the number of the Corsi span, the higher the visuospatial WMC. It has been seen that the average Corsi span for this test is 6 (Kessels et al., 2000). Scholars have noted that working memory span tasks are the most proper way to compare the individual's WMC with each other (Conway et al., 2005). For participants, the average WMC was 5.77 (SD=1.06). Also, the categorization of users based on their WMC used in previous studies (Lerch and Harter, 2001). To conduct the analysis, I performed a median split on visuospatial WMC to group the users based on their WMC. The median of the Corsi span for the participants was 5.75, and I assigned the users with a lower value to the low WMC group (M=4.89, SD=1.08) while I assigned users with a higher value than 5.75 to the high WMC group (M=6.64, SD=1.07).

7.5.4. Measurements

I classified the measurements in this study into four categories. The first category, control variables, focuses on the measurements used in this study to increase the quality of the results. Second, I checked the task resumption performance of the users after experiencing each gaze-based TRS. Third, I investigated how the usage of gaze-based TRS influenced task performance. Finally, I investigated the relevance of gaze-based TRS to the users by checking it from different perspectives, including user interaction data and their perception.

Control Variables: As discussed in previous sections, task, context, interruption characteristics, and user characteristics impact the effectiveness of the gaze-based TRS. I tried to control these factors in the experimental design as well as checking users' visual behavior and interactions after the experiment. In this study all the participants received the same tasks and information dashboards, they attend the experiment in the same laboratory environment with the same monitor size and resolution, all dashboards have similar visual complexity and are controlled for elements that can bias attention allocation and all participants received the same secondary tasks as an interruption. Also, users in each group have almost similar WMC as the control for user characteristics. After the experiment, I investigated the users' visual behavior in the first visit phase by checking their fixation duration among the six AOIs to ensure the similarity among users regarding the task. I also checked the secondary task period's length as an interruption based on collected log data to guarantee that users had the same interruption characteristics. Moreover, I checked if there is any difference between the dedicated time for the control condition (without TRS for ten sec.) and the processing time for the most simple gaze-based TRS, last point highlighting method.

Task Resumption Performance: I measure the resumption performance of users from four perspectives. *First*, I study the RSR by investigating the number of users in each group that could select the last AOI before interruption as the first AOI after the resumption.
I considered a minimum one-second fixation duration as the threshold for the selection of an AOI in the revisit phase. If the first AOI that the user select is the same as the last selected AOI before the interruption, I considered the users as successfully resuming. Conversely, if the first selected AOI is different from the last selected AOI, the user's resumption is considered unsuccessful. Later, I calculated the percentage of successful resumptions for each group in different conditions. Second, I checked the resumption lag, which is the time-period between finishing the second task and resuming the primary task (Altmann and Trafton, 2002). For that, I determined the last fixated AOI of the primary task before the interruption and calculated the time between starting the revisit phase and selecting the same last fixated AOI again. Third, I investigated the attention allocation performance of the users during the revisit phase. In this section, I checked their ability to ignore previously high-visited AOIs in the revisit phase. As the first visit is always 30 seconds, I calculated the attention allocation performance during the first 30 seconds of the revisit phase. For that, I checked the percentage of fixation duration on the two previously high-visited AOIs in this period. A lower fixation duration on previously highvisited AOIs during the revisit phase is considered as a higher performance in attention allocation. Fourth, I measure the centering of users' attention on a limited stimulus field by the number of transitions after resumption (Blascheck et al., 2014). I investigated the changes in the number of transitions between six AOIs before and after receiving TRS. I considered this as their ability to focus their attention on limited AOIs rather than shifting between several of them. A lower transition rate is regarded as more focused attention after the resumption.

Task Performance: As explained in the experimental design, the users could choose to finish the data exploration task after 30 seconds. I measure task performance by tracking the time that users required until clicking on the finish button as the task completion time. A faster finishing of the data exploration task is considered as better task performance.

Gaze-based TRS Relevance: This category includes four measurements that were collected via survey questions. First, I asked users about the usefulness of each gaze-based TRS with the questions adapted from Davis (1989). Second, I asked questions about how easy it was to use each of the highlighting methods as gaze-based TRS. For that, I asked questions adapted from Davis (1989). Third, I collected users' opinions about how each of those gaze-based TRS influenced their self-awareness level by asking questions adopted from Twenge et al. (2007). Fourth, I asked users about the probability that they use each of the gaze-based TRS in the future (behavioral intention) questions adapted from Venkatesh et al. (2003). List of items for each of these constructs can be seen in Appendix E in Table E.1.

Table 7.2 shows the list of measurements used in the third design cycle of the DSR project.

	Measurements	Explanation
trol Variables	Task	 All participants received similar task. Visual behavior of the participants investigated by checking fixation duration among six AOIs before the secondary task.
	Context	 Lab environment, same monitor size, resolution, etc. All participants received the same information dashboards. All four received dashboards are visually similar. Each information dashboard included six AOIs with the same level of complexity.
Coi	Interruption Characteristics	All participants received irrelevant external (it-mediated).The length of interruption are compared with the log data.
	User Characteristics	• Users in each group have the same level of WMC.
	Resumption Accuracy	• The percentage of successful resumption for each gaze-based TRS by comparing the last selected AOI before the interruption and the first selected AOI after resumption.
nption ance	Resumption Lag	• The time-period between finishing the second task and resuming the primary task based on users' fixations.
sk Resun Perform:	Attentional Resource Allocation	• The percentage of fixation duration on the two previously highly visited AOIs in the revisit phase.
Tau	Attention Shift Rate	• It represents the users' ability to focus their attention on limited AOIs rather than shifting between several of them. I compared the changes in the number of transitions between six AOIs before and after receiving gaze-based TRS.
Task Performance	Task Completion Time	• I measured how fast users could finish the data exploration task by checking their interaction data.
S	Perceived Usefulness	• User's opinion about the usefulness of each gaze-based TRS.
sed TF vance	Perceived Ease of Use	• User's opinion about the easiness of each gaze-based TRS.
aze-ba: Rele	Self-awareness	• User's opinion about the influence of each gaze-based TRS. on self-awareness
G	Behavioral Intention	• User's opinion about the probability to use each gaze-based TRS in future.

Table 7.2.: Summary of measurements used in Study V.

7.6. Data Analysis and Results

Control Variables

Figure 7.9 shows the proportion of all fixations on the six AOIs of all users during the first visit of the dashboard. Furthermore, the heatmaps of users' visual behavior in the first visit phase of the experiment can be seen in Appendix E and in Figure E.12 for low WMC users and Figure E.13 for high WMC users. As can be seen, the attention of the users from both groups is biased by the location of the AOIs for all rounds of the experiment. Users from both groups had a similar pattern and dedicated more fixations to the first and second AOI, which are located in the top-left and top-center position. Also, the low number of fixations of other AOIs highlights that users could not finish the primary task before receiving the secondary task. Therefore, this confirms that the secondary task counts as an interruption for all four conditions.



Figure 7.9.: ARA of users in the first visit phase.

In addition, I conducted a Friedman's ANOVA test revealing that the secondary task duration did not significantly differ over the four conditions, for both users with high WMC $(\chi^2(3)=1.95, p=.58)$ and users with low WMC $(\chi^2(3)=0.58, p=.89)$. Moreover, comparing the two groups with each other in all four conditions does not show any significant difference. Therefore, the findings show that the secondary task was similar for the users of both groups and among four conditions.

I also identified that the average process time of the last point highlighting method as the most simple method is 11.43 (SD=4.13). This processing time is close to the selected 10 seconds for the without TRS condition. Therefore, I argue that selecting 10 seconds without TRS condition was a proper decision.

7.6.1. Task Resumption Performance

Resumption Success Rate

Figure 7.10 shows the RSR. As can be seen, 58% of users with high WMC could resume their tasks successfully in the control condition while only 37% of the users with low WMC were successful. Providing the last point as gaze-based TRS also shows almost the same proportion with 54% successful users in the high WMC group and 33% in low WMC users. Nevertheless, providing heatmap shows different results. This highlighting method could support 50% of users with low WMC to successfully resume the task while only 30% of the users with high WMC were successful. Regarding the scanpath, a low amount of users in both groups (42% of high WMC, 33% of low WMC users) could successfully resume.

Overall, the results from without TRS show that both groups have difficulty in resumption (42% of the high WMC and 63% of the low WMC were not successful). Also, these findings suggest that for the high WMC group, the highest number of successful users was when I did not provide any highlighting method. None of the highlighting methods results in better performance than the baseline for this group. These findings suggest that providing these three highlighting methods as gaze-based TRS is not supportive for high WMC users. However, for the users with the low WMC, the heatmap highlighting method was supportive, and in all the other cases, they had an almost similar amount of successful users.



Resumption Sucess Rate

Figure 7.10.: RSR of users in the revisit phase.

Resumption Lag

Next, I checked the resumption lag as the measure of resumption performance. For the users with high WMC, Friedman's ANOVA results show that the resumption lag of participants did significantly differ over the four TRS ($\chi^2(3)=8.1$, p=.04). Post hoc tests were used with Bonferroni correction applied. It appeared that there is no difference between the without TRS condition and any other TRS condition. However, the resumption lag significantly differs from the heatmap (Mean=19.47, SD=22.69) to the last fixated point (Mean=6.74, SD=9.16) (difference=24). In all cases, the critical difference was 23.59729 (α =.05 corrected from the number of tests). For the users with low WMC, the resumption lag of participants did not significantly differ over the four conditions ($\chi^2(3)=0.15$, p=.98). Also, comparing two groups regarding their resumption lag does not show any difference.

Attentional Resource Allocation - Revisit

Figure 7.11 shows the ARA performance of both groups. To examine the difference between ARA performance, I used repeated measure ANOVA for both groups separately. First, the Mauchly test indicated that the assumption of sphericity was not violated for both the high WMC group (W=0.97, p=.98) and the low WMC group (W=0.80, P=0.43). Therefore, the results of this test are reliable for both groups.

For high WMC users, the results show a significant difference in the ARA performance among four conditions (F(3,69)=2.93, P=0.03). Post hoc comparisons using the Tukey HSD test indicated that the mean score for without TRS condition (Mean=28.20, SD=22.24) was significantly different from the last point condition (Mean=41.10, SD=17.17), (P=0.02). These findings suggest that the last point misleads users with high WMC while they had better performance without this support. The other comparisons do not show any difference. For users with low WMC, statistical results do not show significant main effects (F(3,69)=0.9, p=0.44).

Furthermore, running between analysis shows a significant main effect for WMC groups (F(3,138)=3.11, P=0.02). Executing post hoc analysis indicated that users with high WMC (M=41.10, SD=17.17) allocated more attention to previously highly attended AOIs than low WMC (M=27.02, SD=20.65) when they receive the last point as highlighting method (t(44.52)=2.56, p=.01).

As can be seen in Figure 7.11, the usage of any gaze-based TRS for users with high WMC leads to slightly higher ARA on previously high-visited AOIs (worse ARA performance) in comparison with the without TRS condition. This is the opposite for the users with low WMC while providing any gaze-based TRS supports them in a better ARA performance.



Figure 7.11.: ARA of users on previously high-visited AOIs.

Attention Shift Rate

Later, I checked the difference in the ASR in all conditions for both groups. For the users with low WMC, wilcoxon signed-rank comparison reveals that the last point highlighting leads to a significantly lower number of transitions after receiving TRS (Mean=13.46,SD=7.32) than before (Mean=16.08,SD=6.79) (Z=0.53, p=.01). For users with high WMC, the results from paired sample t-tests reveal that receiving a scanpath feedback results in less number of transitions in the revisit phase (Mean=13.83, SD=5.98) compared to the first visit phase (Mean=18.33, SD=8.53) (t(23)=2.85, p=.009). In all the other conditions, there is no difference in the ASR for both groups.

7.6.2. Task Performance

Task completion time refers to the time that users required to end exploring the information dashboard by clicking on the finish button as a part of the revisit phase. The results from Friedman's ANOVA reveal that the task completion time did not significantly differ over the four conditions, for both users with high WMC ($\chi^2(3)=2.55$, p=.46) and with low WMC ($\chi^2(3)=2.65$, p=.44). Moreover, comparing two groups with each other in all four conditions does not show any significant difference. Therefore, these findings suggest that the three highlighting methods as gaze-based TRS do not support finishing the task faster for any user group.

7.6.3. Gaze-based TRS Relevance

Figure 7.12 shows the opinions of participants from both groups regarding the perceived usefulness, ease of use, self-awareness, and behavioral intention for the three presented highlighting methods.

Perceived Usefulness

In general, users with low WMC tend to rate all highlighting methods more useful in comparison with users with high WMC. This indicates that users with high WMC are more confident in managing short-term interruptions. Comparing the perceived usefulness within each WMC group does not show any difference between the three highlighting methods. Nevertheless, comparing both groups with each other running independent t-tests shows that there was a significant effect for heatmap (t(45.96)=-2.0886, p=.04) and low WMC (M=5.15, SD=1.44) attaining higher scores than high WMC (M=4.29, SD=1.4). These results show that low WMC users found heatmap more useful than users with high WMC.

Perceived Ease of Use

Regarding the ease of use, the results from the wilcoxon signed-rank comparison show that users with high WMC found heatmap (Mean=5.57, SD=1.22) easier to use than the scanpath (Mean=4.64, SD=1.18) (Z=0.60, p=.007). The same applies for users with low WMC, heatmap (Mean=5.60, SD=0.98), scanpath (Mean=4.68, SD=1.61) (Z=0.59, p=.008). Also, there is no difference between the last point and heatmap for users in both groups.

Self-awareness

Participants from both groups believe that the scanpath and heatmap increase their selfawareness more than the last point. For users with high WMC, wilcoxon signed-rank comparison for each pair reveals that they found to be significantly less self-aware with the last point highlighting (Mean=2.24, SD=1.08) than with the heatmap (Mean=4.75, SD=1.30) (Z=0.91, p<.0001) and the scanpath (Mean=4.58, SD=1.33) (Z=0.88, p<.0001). For users with low WMC, the same test shows that they have a similar opinion. The last point highlighting (Mean=2.97,SD=1.75) lead to a significantly lower self-awareness than the heatmap (Mean=5.04,SD=1.47) (Z=0.79, p=.0003) and the scanpath (Mean=4.99,SD=1.55) (Z=0.68, p=.002). Conducting a between-subject analysis depicts that there is no difference between groups.

Behavioral Intention to Use

Regarding the behavioral intention to use, conducting a paired sample t-test for the users with low WMC shows that they prefer to use heatmap (Mean=4.31, SD=1.73) more than the scanpath (Mean=3.47, SD=1.74) (t(23)=2.11,p=0.04). There is no difference between using the last point and heatmap. For users with high WMC, there is no difference

between the three highlighting methods. Moreover, users with low WMC tend to use these highlighting methods in the future more than high WMC users. By comparing the user groups, these findings suggest that especially users with low WMC have a higher intention to use heatmap than users with high WMC (t(45.54)=-2.24, p=0.02).



Figure 7.12.: Users' opinions about Gaze-based TRS.

Table 7.3 shows a summary of the results collected in this study. Comparing the results indicates the importance of considering WMC when designing gaze-based TRS.

Using the last point method, the participants from both groups were not influenced strongly. However, it is a popular highlighting method in the field of gaze-based TRS. Providing this could only affect users with low WMC to have fewer attention shifts. I believe that providing a new task and interface (data exploration on information dashboards) is the primary reason for that. This finding shows the importance of designing different gaze-based TRS based on the dimensions that discussed in Section 7.2 and challenges to generalizing one highlighting method. By considering the role of WMC, I also uncovered that participants with different WMC found using the last point method easy to use, but they do not have a strong intention to use it in the future in comparison with the other gaze-based TRS types. Therefore, these findings suggest that providing last point gaze-based TRS in the specific case as performing data exploration tasks in information dashboards with short-term interruptions is not helpful for any of the user groups.

These findings show that the heatmap method is effective for participants with lower WMC, while it is not that useful for users with high WMC. For users with low WMC, this supports better task resumption performance by supporting to improve the success rate and a slightly better revisit attention allocation. Moreover, participants perceived it useful, easy to use, and supportive with regards to self-awareness. They also have a high intend to use it in the future. For users with higher WMC, this highlighting method does not support resumption performance. Besides, they have a lower intention to use in comparison with users with low WMC, although it is easy to use and can support increasing self-awareness.

Regarding the scanpath highlighting method, these findings suggest that it was supportive for users with high WMC to reduce their attention shifts and also to increase their selfawareness. However, they perceived it as not easy to use in comparison with the other gaze-based TRS types. For the participants with lower WMC, this TRS type could only positively influence the self-awareness in comparison with the last point TRS and does not have any other consequences.

Overall, these findings suggest that providing gaze-based TRS for short term interruptions is more supportive for users with low WMC than high WMC. Providing any of those highlighting methods could not play an extensive supportive role for users with high WMC. However, users with low WMC could get the benefit of them from different aspects, especially using the heatmap highlighting method.

	Measurements	Results
	Resumption Success Rate	 For high WMC users, the no TRS condition lead to a higher resumption success rate. For the low WMC users, heatmap condition lead to a higher resumption success rate. For both groups, providing the last point highlighting is almost like the without TRS condition.
tesumption formance	Resumption Lag	 For high WMC users, providing the heatmap results in higher resumption lag in comparison with the last point. For low WMC users, there is no difference between conditions. For both groups, providing any gaze-based TSR does not support reducing the resumption lag in comparison with the without TRS condition.
Task F Per	Attentional Resource Allocation	 For high WMC users, providing any gaze-based TRS, tends to decrease attention allocation performance. Especially the last point highlighting method has a strong effect on decreasing performance. For low WMC users, providing any gaze-based TRS, tends to support improving attention allocation performance.
	Attention Shift Rate	 For high WMC users, scanpath highlighting supports improving focused attention. For low WMC users, last point highlighting supports improving higher focused attention.
Task Performance	Task Completion Time	• For both groups, providing any gaze-based TRS does not improve task completion time.
	Perceived Usefulness	• Users with low WMC perceive the heatmap more useful than high WMC users.
TRS ce	Ease of Use	 For high WMC users, using heatmap is easier than scanpath. For low WMC users, using heatmap is easier than scanpath. For both groups, there is no difference between last point and heatmap regarding ease of use.
e-based Relevan	Self-awareness	• For both groups, users believe that heatmap and scanpath increase their self-awareness more than the last point highlighting method.
Gaz	Behavioral Intention	 For low WMC users, the intention to use heatmaps is higher than scanpath. For high WMC users, there is no difference between the intention to use of any highlighting method in the future. Users with low WMC have a higher intention to use heatmap compared with the users with high WMC.

Table 7.3.: Summary of findings by comparing gaze-based TRS and the role of WMC.

7.8. Summary

The third design cycle is focused on supporting information dashboard users in resuming an interrupted task. This was identified as one of the MRs in the Study II. These days, interruptions are a crucial phenomenon in private and professional life and users are getting used to shift their attention to secondary tasks while working on a primary one frequently. The increased amount of such situations highlights the need for TRS systems. Eye tracking is a promising technology to provide TRS using gaze-based highlighting methods. With such support, users can be reminded of their previous visual behavior and assisted in resuming the primary task more efficiently in case of interruptions. In this study, I first conceptualized individualized VAF for TRS after an interruption as gazebased TRS. Later, I presented dimensions that impact designing effective gaze-based TRS by analyzing previous research in the field of interruption and gaze-based TRS. Especially, I analyzed the collected papers in the Study I that focused on providing TRS with realtime eye movement data. Based on that, I present MRs and DPs for designing attentive information dashboards with gaze-based TRS as an individualized VAF. Furthermore, I proposed the system architecture for the development part and mapped identified DPs to the design features. For gaze-based TRS I suggested and developed three types of highlighting methods for gaze-based TRS as last point, heatmap and scanpath. Based on Gregor (2006), the presented MRs, DPs, and design features are considered as type V theory (Design and Action), since it gives explicit prescriptions for constructing attentive information dashboards that provide gaze-based TRS.

Also, the results from Study II show the need for individualization of attention support based on users working memory. By analyzing the existing research in gaze-based TRS, I identified that it is not well understood how individuals' WMC influences gaze-based TRS effectiveness. Therefore, I designed an exploratory laboratory experiment to evaluate three gaze-based highlighting methods (last point, heatmap, scanpath) after a short term IT-mediated interruption of a data exploration task. The results suggest that designing gaze-based TRS should be different for users based on their WMC. Especially heatmap, is supportive for low WMC users, while users with high WMC may not need gaze-based TRS. With this study, I provide evidence for the necessity of designing personalized gazebased TRS considering WMC in the future. Based on Gregor (2006), the results from this exploratory study are a type II theory (Explanation), since it provides explanations about the effects of providing different gaze-based TRS after a short-term interruption.

This study contributes to the field of gaze-based TRS from different perspectives. First, I extended the current study in the field of gaze-based TRS by testing new highlighting methods, including heatmap and scanpath, and compared them with the last point highlighting as a common gaze-based TRS. Second, I extended the research field by considering a new task. To the best of my knowledge, it is the first study that investigates TRS in the context of data exploration tasks using information dashboards. This specific task is expected to influence the effectiveness of gaze-based TRS since the user's eye movement patterns depend on the context and the task. Third, I investigated the role of the user's WMC on the effectiveness of different highlighting methods.

8. Discussion ¹

8.1. Theoretical Contributions

To classify the theoretical contributions of this thesis, I rely on the taxonomy of theory types in IS research provided by Gregor (2006). Table 8.1 presents an overview of the theoretical contributions of each study.

In Study I, I presented a conceptualization for the systems that integrate real-time usage of eve movement data to develop innovative interaction, considered as eve-based IIS (see Section 3.2.2, Figure 3.1). Although the development of such systems increased during the last years (Chuang et al., 2019; Nakano et al., 2016), there is a lack of conceptualization for designing such systems. In an attempt to do so, I investigated to provide a conceptual framework for research on eye-based IIS. Based on Gregor (2006), this framework is a type I theory (Analysis) since it describes eye-based IIS and analyses the relationship between different dimensions of research within designing such systems. In this framework, I distinguish three high-level categories as 1) influencing factors, 2) eye-based IIS properties, and 3) outcomes. Each category involves several particular elements. The influencing factor category focuses on elements that impact the design of the eye-based IIS. This category covers context, task, eye tracking technology, experimental setup. The second category, eye-based IIS properties, is considered as the core of the framework. It includes sensing and reasoning as the fundamental elements to design eye-based IIS. Later, these properties are used to design eye-based IIS, which can have two main focuses: making the system more intelligent (system adaptation) or increasing user awareness (user adaptation). In the last category (outcome), I investigate the outcomes of the designed eye-based IIS through three dimensions: perception, behavior, and performance. Furthermore, there is a lack of an overview of existing research on the systems that integrate real-time eye movement data for designing innovative interactions. Therefore, I conducted a SLR study and developed a conceptual framework by analyzing the collected papers. Consequently, I presented a descriptive analysis based on all elements of the framework in detail. These findings reveal several potential future research directions for each dimension that guide academic researchers for further investigation of eye-based IIS.

In **Study II**, I used eye tracking technology for diagnostic purposes and conducted an exploratory study to investigate the visual behavior of users with high and low WMC while exploring information dashboards. The results from the eye movement analysis reveal that both groups of users with low and high WMC have difficulties in managing limited attentional resources while exploring information dashboards. Also, the results show that users have difficulty in resuming an interrupted task. Furthermore, in the case of a second chance to improve information processing performance, both groups of users have

¹This Chapter is based on the following studies which are published or in work: Toreini and Morana (2017), Toreini et al. (2018c), Toreini et al. (2018b), Toreini et al. (2018a), Hummel et al. (2018), Toreini and Langner (2019), Toreini et al. (2020b), Langner et al. (2020), Toreini et al. (2020c), Toreini and Maedche (2020), Toreini et al. (2020a)

difficulty selecting a proper revisit strategy and repeat their behavior. Based on Gregor (2006), the results from this exploratory study represent a type II theory (Explanation) since it provides explanations about managing limited attentional resources of users with high and low WMC while exploring information dashboards. The findings from this study highlight the need for designing an innovative information dashboard that considers the limitations of attentional resources and WMC of the users. Therefore, I extracted six MRs for designing attentive information dashboards. Based on the developed framework in Study I, three MRs focus on designing an attentive information dashboard that reflects system adaptation focus, and the other three focus on user adaptation by increasing users' awareness with providing VAF. Following Gregor (2006), the presented MRs are considered as type V theory (Design and Action), since they provide explicit prescriptions (MRs) for constructing attentive information dashboards.

In Study III, I investigated different attention support solutions for information dashboard users. First, I presented three suggestions for designing VAF for the information dashboard users by integrating real-time and off-line records of eye movement data. Later, I investigated the effects of these three VAF types in an eye tracking pilot study. The first suggestion is an individualized VAF that functions by tracking the users' eye movement data in real-time and presents the users' gaze duration on each chart of the dashboard. The second VAF suggestion is a general VAF, which is an example of a proper visual attention allocation based on off-line records of eye movement data of other users who conducted the same data exploration task on the same dashboard. The third VAF type is another general VAF that follows the same approach as the second VAF but presents an example of improper VAA using off-line eve movement data. The results from comparing these three VAF types reveal that providing individualized VAF assists users in managing their limited attentional resources in a better way in comparison with providing off-line records of eye movement data as VAF (proper or improper example of visual attention allocation as VAF types). Based on Gregor (2006), the results from this eye tracking pilot study is a type II theory (Explanation), since they provide explanations about the effects of using real-time and off-line records of eye movement data as VAF on managing limited attentional resources while exploring information dashboards.

In Study IV, I investigated the design of attentive information dashboards for data exploration. For that, I presented two theoretically grounded DPs for designing attentive information dashboards that support the data exploration task with individualized VAF. vom Brocke et al. (2013) have emphasized that there are limited contributions in the DSR community that make actual use of the potential of neuroscience tools (e.g., eye tracker) to design advanced built-in functions for IT artifacts. Eye tracking technology has recently been extended in IS research (Dimoka et al., 2012; Riedl et al., 2017); however, the usage of it was limited to analyzing off-line eye movement data to understand human visual activities while working with IS applications (Riedl et al., 2017; Vasseur et al., 2019). In this study, I contribute to the field of IS by integrating real-time eye movement data to support users in managing their limited attentional resources while exploring information dashboards. Furthermore, I presented a system architecture for attentive information

dashboards that support data exploration with three components. This system architecture promotes mapping two provided DPs to design features. Based on Gregor (2006), this contribution is considered as type V theory (Design and Action), since it gives explicit prescriptions for constructing attentive information dashboards that provide individualized VAF for data exploration tasks. Later, I instantiated both DPs in a software artifact and evaluated them in a large-scale laboratory experiment as a suggested method to evaluate DSR projects (Pries-Heje et al., 2008; Venable et al., 2012). In this experiment, I compared two design configurations with DPs activated (individualized VAF) and deactivated (general VAF with a text-based explanation about attention). For the evaluation, I suggested an underlying research model and three hypotheses examine the impact of the proposed DPs. For the data analysis, I focused on collected eye movement data during the experiment. I mainly analyzed users' eye movement data while exploring the information dashboard in the first visit, after receiving the individualized VAF (revisit phase) and at the end of the task. The results of this study reveal that individualized VAF enhances users' ability to manage limited attentional resources. Based on the findings, the users without DPs had difficulty in managing their limited attention from different perspectives (Attentional Resource Allocation, Attention Shift Rate, and Attentional Resource Management) while users with DPs performed better in all perspectives. Based on Gregor (2006), this contribution is considered as II theory (Explanation) since it provides explanations about the effects of using DPs on enhancing users ability to manage their limited attentional resource during data exploration tasks.

In Study V, I first conceptualized individualized VAF that supports resuming interrupted tasks, gaze-based TRS. Later, I presented dimensions that impact designing effective gazebased TRS by analyzing previous research in the field of interruption and gaze-based TRS. The detailed conceptualization is discussed in Section 7.2. Based on that, I presented six MRs and three DPs for designing an attentive information dashboard that supports users' task resumption with gaze-based TRS. Next, I proposed the system architecture for developing such systems. This system architecture helps to map three identified DPs to the design features. Also, for this study, I suggest three types of highlighting methods for gazebased TRS as last point, heatmap, scanpath. Based on Gregor (2006), the presented MRs, DPs, and design features are considered as type V theory (Design and Action), since it gives explicit prescriptions for constructing attentive information dashboards that provide gaze-based TRS. Furthermore, this study extends the current knowledge in the field of gaze-based TRS by integrating new highlighting methods (heatmap and scanpath) for gaze-based TRS. Based on the collected papers in the Study I, the previous studies focused mainly on last point highlighting methods, and the comparison of different highlighting methods was not investigated. This study also extends the gaze-based TRS research field by considering a new context and task by using information dashboards for data exploration tasks. Later, I investigated the role of the user's WMC on the effectiveness of different highlighting methods for short term IT-mediated interruptions in a large-scale exploratory study as a suggested method to evaluate DSR projects (Pries-Heje et al., 2008; Venable et al., 2012). The results suggest that the need for gaze-based TRS types is different for users with high and low WMC. Notably, the heatmap highlighting method

is supportive for low WMC users, while users with high WMC may not need gaze-based TRS for short-term IT-mediated interruptions. Based on Gregor (2006), the results from this exploratory study are a type II theory (Explanation), since it provides explanations about the effects of providing different gaze-based TRS after a short-term interruption.

	Study	Theory Type	Main Contributions
SLR	I	Type I theory: Analysis	 Conceptualization of eye-based IIS along with the three high-level categories: influencing factors, eye-based IIS properties, outcomes. Descriptive analysis of research on using real-time eye movement data for designing eye-based IIS. Formulation of future research directions for eye-based IIS.
Design Cycle 1		Type II theory: Explanation	• Users with high and low WMC have difficulty in managing limited attentional resources while exploring information dashboards.
	Π	Type V theory: Design and Action	• Formulation of MRs for designing attentive information dashboards that focus on making the system more intelligent and support system adaptation or increasing user awareness and support user adaptation.
	III	Type II theory: Explanation	 Provide suggestions as VAF types (utilizing real-time or off- line records of eye movement data). Individualized VAF (utilizing eye movement data in real- time) supports managing limited attentional resources in a better way compared with the proper and improper examples of attention allocation as VAF types (utilizing off-line records of eye movement data).
Cycle 2	IV	Type V theory: Design and Action	• Formulation of MRs, DPs, and DFs for designing an attentive information dashboard that supports data exploration tasks with individualized VAF.
Design		Type II theory: Explanation	• Individualized VAF support users in managing their limited attentional resources during data exploration tasks.
Cycle 3	V	Type V theory: Design and Action	• Formulation of MRs, DPs, and DFs for designing an attentive information dashboard that supports task resumption after an interruption with gaze-based TRS.
Design (v	Type II theory: Explanation	• Users with different WMC, need different types of highlighting methods as gaze-based TRS.

Table 8.1.: Theoretical contributions according to theory types by Gregor (2006).

8.2. Practical Contributions

The results of this thesis also are beneficial for at least two groups of practitioners. The first group is information dashboard designers who concentrate on designing information dashboards for existing BI&A platforms. The second group comprises eye-based application designers who are mainly HCI practitioners that focus on integrating real-time eye movement data for designing innovative interactions in different contexts and tasks.

Information Dashboard Designers

Designing effective systems requires a detailed understanding of the underlying cognitive processes while users working with them (Lerch and Harter, 2001). The results from the Study II support information dashboard designers to better understand the role of limited attentional resources and working memory while users explore information dashboards. Researchers have used eye trackers to evaluate some specific design features on information dashboards (e.g., presentation formats, colors, size, position, etc.) (Alberts, 2017; Bera, 2014,1; Burch et al., 2011; Nadj et al., 2020) and limited attention and working memory challenges of information dashboard users are not investigated. For instance, dashboard designers need to highlight critical business states in their design since users have difficulty managing their attentional resources and may miss some information. Previous studies have addressed the need for providing such feedback to inform users about critical business states proactively (O'Donnell and David, 2000; Yigitbasioglu and Velcu, 2012). They can also consider designing personalized information dashboards based on users' WMC or their ability to manage limited attentional resources. Furthermore, users' visual behavior in Study I, and control groups of Study IV and Study V show that users repeat their visual behavior in the revisit phase. This happens since they have difficulty to remember those activities which have been completed (Singh, 1998). This can result in missing changed information on dynamic dashboards (Healey and Enns, 2012). Therefore, information dashboard designers should consider highlighting changes when users revisit information dashboards to avoid missing important information.

So far, commercial BI&A tool providers, such as Tableau, use eye trackers for understanding user's behavior while exploring dashboards (Alberts, 2017). But, the usage of eye movement data in real-time for creating attentive information dashboards is not yet integrated into BI&A platforms (Silva et al., 2019). Within the last years, technology firms have recognized the potential of neuroscience technologies in advancing user and IS applications (vom Brocke et al., 2013). So far, the high price and complexity of neuroscience tools challenged using these devices in workplaces. However, the usage of eye tracking technology increased in recent years because of the availability of cheaper, faster, more accurate, and easier to use eye trackers (Duchowski, 2017). In this DSR project, I used the Tobii 4C eye tracker ² as one of the cheapest eye trackers in the market (Farnsworth, 2019) to develop attentive information dashboards with two types of individualized VAF. This DSR project provides prescriptive knowledge that guides the design of attentive information dashboards for practice (Gregor and Hevner, 2013; Peffers et al., 2007). Particularly the

²https://gaming.tobii.com/tobii-eye-tracker-4c/

presented information in Study IV and V can support commercial BI&A tool providers to enhance their capabilities by designing attentive information dashboards with eye trackers. They can replicate the design of individualized VAF for data exploration tasks or provide support for resuming interrupted tasks. Also, they can use the attentive information dashboard to design further individualized VAF that supports other tasks. For example, they can integrate attentive information dashboards in collaborative scenarios and enhance the team's joint attention with individualized VAF (Toreini et al., 2018a).

Eye-based Application Designers

The usage of eye trackers has moved from the controlled lab environment to everyday settings (Chuang et al., 2019). The number of applications that work with eye tracking increased during the last years. Tobii company as one of the leading companies in this field has announced that eye tracking technology is coming to the devices we use every day and enterprises should get ready for that (Eskilsson, 2019). Besides, Microsoft recently released the usage of eye control on Windows 10 to ease the interaction between users and the system (Microsoft, 2019). This thesis's findings can support eye-based application designers to design new eye-based features. The results from the Study I can support eyebased application designers to understand different dimensions of integrating real-time eye movement data for designing innovative IIS. Also, it provides a comprehensive overview of existing research studies in this field from different perspectives. Furthermore, the results from Study III and Study IV support eye-based application designers for designing data exploration support features for enterprise applications beyond information dashboards. For example, SAP considered the usage of eye trackers for the next version of enterprise systems (Galer, 2019), and such feedback can be integrated into other enterprise applications that users need proper attention allocation. Also, it can be integrated into self-tracking dashboards in workplaces such as MyAnalytics dashboard³ developed by Microsoft and support them to manage their limited attentional resources. The results from the Study V can be integrated into applications and tasks that users need for several attentional shifts. For example, such feedback can be for the employees that need to work on several monitors or to work in situations with high interruption rates.

Furthermore, the developed attentive information dashboards and individualized VAF types can be transferred for the situations that need to wear head-mounted eye trackers. For example, the results from the Study III and Study IV can be integrated into google glass enterprise edition (Kothari, 2019) to help business employees to work smarter by managing their attentional resources while exploring information in real-world. In another example, the results from this DSR project, especially the Study III, IV, and V can be integrated into the VR scenarios. Eye tracking technology enables new forms of interactions in VR and AR, and the suggested individualized VAF in this DSR project can be integrated into VR glasses that have built-in eye tracking such as Vive Pro Eye 4 or AR such as Microsoft HoloLense 2 5 .

 $^{^{3}}$ https://docs.microsoft.com/en-us/workplace-analytics/myanalytics/use/dashboard-2

⁴https://www.vive.com/eu/product/vive-pro-eye/

⁵https://docs.microsoft.com/en-us/windows/mixed-reality/eye-tracking

To summarize the DSR project findings, I use the six core components of design theory that Gregor and Jones (2007) have introduced. Table 8.2, Table 8.3, Table 8.4 and Table 8.5 show the results of this DSR project in the form of nascent design theory.

The first component is the system's purpose and its scope. This component describes the goal of the system which is developed in this DSR project. The findings from this perspective are presented in Table 8.2. In this DSR project, I aimed to design and develop attentive information dashboards sensitive to the users' attention by tracking their eye movement data in real-time. The focus of developed attentive information dashboards is on increasing users' awareness about their previous visual behavior by providing individualized VAF. In this thesis, I focused on providing individualized VAF for two situations in which users have difficulty managing their limited attentional resources. First, I proposed supporting users during data exploration tasks on information dashboards (design cycle 2). In this case, an individualized VAF aims to increase users' self-awareness during data exploration and ultimately improve information processing performance. Second, supporting users in resuming an interrupted task (design cycle 3). In this case, I proposed gaze-based TRS as an individualized VAF when users return to their primary task after an interruption. I described the detailed motivation and goal of each DSR design cycle in Chapter 1, Section 1.1 and at the beginning of each design cycle.

The second component is justificatory knowledge. This component describes the theory that gives basis and explanation for the design. The summary of this component is presented in Table 8.2. For this DSR project, I used the knowledge described in attention, working memory, and Human Information Processing (HIP) theory. For attention, I used selective attention theories that focus on limitation of attention following Broadbent's filter theory (Broadbent, 1958), goal-directed attention that refers to the voluntary type of attention (Corbetta and Shulman, 2002), overt attention is considered as an extrinsic behavior such eye movements (Posner, 1980) and the eye-mind assumption that indicates where users are fixating is underlying their cognitive process such as allocating their attention (Just and Carpenter, 1980). For explaining different types of memories and their relationship, I focused on multi-model memory (Atkinson and Shiffrin, 1968). I also integrated the concept of working memory from Baddeley's model of working memory (Baddeley and Hitch, 1974) and the updated version of it (Baddeley, 2000). Moreover, I used the adapted version of HIP stages by Wickens et al. (2016) for explaining different stages of processing information on the dashboard and the role of attention and working memory on that. The foundations regarding these theories are discussed in Chapter 2, Section 2.2. Furthermore, for design cycle 3, I integrated the stages of interruption and resumption from (Trafton et al., 2003) to explain the users' attention shift procedure and the role of individualized VAF to support resuming interrupted tasks. The details are discussed in Section 7.5.1. Besides these theories, I leveraged prescriptive knowledge from existing studies in eye tracking, AUI, BI&A, and information dashboards explained in Section 2.3. Furthermore, I leveraged the identified studies in the conducted SLR study in Chapter 3 that focused on data exploration, and gaze-based TRS.

The third component describes the key constructs leveraged in this DSR project. Participants had the same task as data exploration in all four studies of the DSR project. Also, in all studies, the data exploration task had two phases: the first visit phase in which the participants started to explore the information dashboard and revisit phase in which participants had the chance to explore the same dashboard again after receiving VAF types. In addition to these two phases, in the Study V, the users had to conduct a secondary task in between these two phases as an interruption task. Table 8.3 shows a list of constructs for each DSR design cycle. As can be seen, in all design cycles, I used eye tracking data to analyze users' visual behavior from different perspectives. All studies involve the participants' performance in "Attentional Resource Allocation" in the revisit phase. With this construct, I examine the ability to ignore previously high-visited AOIs and to allocate attention to the previously low-visited AOIs. It was essential for tracking how the individualized VAF supported users in managing their limited attentional resources and compare it with the situation that users did not receive such support. Also, in all studies of the DSR project, except Study II, I measured the "Attention Shift Rate" to investigate the ability to select limited charts to process in the revisit phase rather than allocating attention to several AOIs. Besides, in studies II and IV, I measured user's "Attentional Resource Management" at the end of the task, explaining the participant's ability to allocate attention to all dedicated charts on the dashboard properly in a limited time. In studies II and IV, I also investigated the task resumption performance of the participant based on their eye movement data. For that, I measured the "Resumption Success Rate" construct that shows the percentage of users that could select the last AOI before interruption as the first AOI after the resumption. Furthermore, in Study V, I investigated the "Resumption Laq" construct as the time-period between finishing the secondary task and resuming the primary task. The usage of user's log data was only integrated with Study V to measure the "Task Completion Time" since the data exploration time was not fixed like in the other studies. I considered this construct as task performance for Study V. Besides that, I evaluated three designed highlighting methods as gaze-based TRS of Study V through survey questions. In this survey, I collected data for four constructs. First, "Perceived Usefulness" as the user's opinion about the usefulness level of each gaze-based TRS. Second, "Perceived Ease of Use" as the user's opinion about how easy it was to use each of the gaze-based TRS. Third, "Self-awareness" as the user's opinion about the influence of gaze-based TRS on self-awareness. Fourth, "Behavioral Intention to Use" as the user's opinion about the probability of using each gaze-based TRS in the future. The detail of how each of these constructs is measured for the conducted studies are discussed in each study separately.

The **fourth component** is capturing the principles of form and function. Table 8.4 shows a list of DPs derived by this DSR project for each design cycle. In design cycle 1, I conducted an exploratory study using eye tracking and reviewed existing literature to identify six MRs for designing attentive information dashboards. Second, I tested two general approaches for supporting attentional resource allocation by integrating the eye tracking technology in an eye tracking pilot study. One approach was the usage of eye movement data in real-time and the other one off-line usage of eye movement data from

other users (e.g., experts) as VAF. Later, I derived two theoretically grounded DPs in design cycle 2 and three theoretically grounded DPs in design cycle 3 based on existing literature. First DP in both design cycles has a similar goal by enabling the information dashboard with tracking user's eye movement data in real-time. So, all in all, I presented four distinct DPs for designing attentive information dashboards that both support users in data exploration and resuming an interrupted task. Later I evaluated the effects of these DPs quantitatively in two separate large-scale eye tracking laboratory experiments as a suggested method to evaluate DSR projects (Pries-Heje et al., 2008; Venable et al., 2012). I explained the detailed description of DPs in design cycle 2 in Section 6.2 and design cycle 3 in Section 7.3.

The **fifth component** is focused on artifact mutability that discusses the changes of the artifact that is anticipated in theory. Table 8.5 shows a list of software artifacts designed and developed in this DSR project. In design cycle 1, I compared the effect of three VAF types with each other in an eye tracking pilot study. Two of them used off-line records of eye movement data, one with a proper example of attention allocation and the other one with an improper example of attention allocation from other users who did the same task. The third one shows the individualized VAF by tracking user's eye movement data in real-time and has the goal to increase user's self-awareness. Later in design cycle 2, I improved the design of individualized VAF and compared it with another software artifact that provides general feedback about attention in a text format in a large-scale eye tracking experiment. To design attentive information dashboards for the data exploration support with individualized VAF, I presented a system architecture with three subsystems. The details of this system architecture, the role of each subsystem, and the mapping to defined DPs are discussed in Section 6.3. Later in design cycle 3, I presented three software artifacts that use different highlighting methods (last point, heatmap, scanpath) as individualized VAF for resuming interrupted tasks, called gaze-based TRS. For designing individualized VAF as TRS, I presented a system architecture with four subsystems. The details of this system architecture, the role of each subsystem and the mapping to defined DPs are discussed in Section 7.4.

The sixth component describes the testable propositions. Study I was an exploratory study to investigate the role of limited attention and working memory on data exploration tasks. Study II was also an exploratory study to test the difference between two suggested solutions for VAF by integrating eye tracking technology (off-line and real-time use of eye movement data for VAF) in an eye tracking pilot study. Therefore, these two studies do not include any hypothesis to examine. Later in design cycle 2, I formulated three hypotheses for evaluating the effects of the DPs on information processing performance. The details of these hypotheses and the research model for this study are discussed in Section 6.4. The design cycle 3 was also an exploratory study to investigate the role of WMC on the effectiveness of different highlighting methods for gaze-based TRS. In this study, I evaluated the users' visual behavior from high and low WMC and summarized the findings. This study also does not include any hypothesis.

The aim is to develop attentive information dashboards with eye tracking technology to support users in processing information by managing limited attentional resources. Particularly, I focused on two scenarios that information dashboard users have difficulty to manage their limited attentional resources: Scope Conducting data exploration tasks. Resume an interrupted data exploration task. Purpose and 1 I proposed two theoretically grounded DPs for attentive information dashboards with individualized VAF that support data exploration, developed the software artifact and evaluated it in the lab experiment. • I proposed three theoretically grounded DPs for individualized VAF support task resumption, developed the artifact with three highlighting methods (heatmap, scanpath, latspoint) as gaze-based TRS and evaluated them in the lab experiment by including the role of WMC. Built on the attention-related theories as: Attention as a limited resource: Selective attention as Broadbent's Filter Model (Broadbent, 1958). Voluntary attention allocation: Goal-directed attention (Corbetta and Shulman, 2002). Attention as extrinsic behavior (eye movement): Overt attention (Posner, 1980). Eye fixations as attention allocation: Eye-mind assumption (Marcel Justificatory knowledge A. Just and Carpenter, 1980). Built on the memory-related theories as: • Different types of memory: Multi-store model of memory (Atkinson- $\mathbf{2}$ Shiffrin, 1968). Working Memory: Baddeley's model of working memory (Baddeley and Hitch, 1974), (Baddeley, 2000). Leverage HIP theory to explain different steps of processing information adapting Wickens et al. (2015). • Leverage the model of memory for goals (Altmann & Trafton, 2002) and stages of interruption and resumption (Trafton et al. 2003) to explain the role of individualized VAF for task resumption support (Study V). Leverage prescriptive knowledge from existing studies in the attentive UI and VAF discipline for data exploration and task resumption support.

Table 8.2.: Nascent design theory following Gregor and Jones (2007): purpose and justificatory knowledge.

		Design Cycle 1	 I considered three constructs in Study II using eye movement data: Attentional Resource Allocation: User performance in allocating limited attention on previously high and low-attended AOIs and ignore previously high-attended AOIs. Attentional Resource Management: The ability to distribute the attention properly among all stimulus on the screen. Resumption Success Rate: The percentage of users in each group (low and high WMC) that could select the last AOI before interruption as the first AOI after the resumption. I considered two constructs in Study III using eye movement data: Attentional Resource Allocation: User performance in allocating limited attention to previously low-attended AOIs and ignoring previously high-attended AOIs in the revisit phase. Attention Shift Rate: The centering of attention on limited set of AOIs on the dashboard.
	ucts	Design Cycle 2	 I considered three constructs in Study IV using eye movement data: Attentional Resource Allocation: User performance in allocating limited attention on previously low-attended AOIs and ignore previously high-attended AOIs. Attention Shift Rate: The centering of attention on limited set of AOIs on the dashboard. Attentional Resource Management: The ability to distribute the attention properly among all AOIs on the dashboard.
3	Constr	Design Cycle 3	 I considered four constructs in Study V for task resumption performance using eye movement data: Attentional Resource Allocation: User performance in allocating limited attention on previously low-attended AOIs and ignore previously high-attended AOIs after each gaze-based TRS. Attention Shift Rate: The centering of attention on limited set of AOIs before and after receiving each gaze-based TRS. Resumption Success Rate: The percentage of users in each group (Gaze-based TRS) that could select the last AOI before interruption as the first AOI after the resumption. Resumption Lag: The time-period between finishing the second task and resuming the primary task based on users' fixations. I considered one task performance construct using interaction data: Task Completion Time: How fast users could finish the data exploration task by checking their interaction data. I considered four constructs for gaze-based TRS relevance using survey: Perceived Usefulness: Users opinion about the usefulness level of the gaze-based TRS. Self-awareness: Users' opinion about the influence of gaze-based TRS on self-awareness. Behavioral Intention to Use: Users' opinion about the probability to use each gaze-based TRS in future.

Table 8.3.: Nascent design theory following Gregor and Jones (2007): constructs.

		Design Cycle 1	 Based on the results from an eye tracking study for diagnostic purposes and existing literature, I derived meta-requirements for designing attentive information dashboard. Later I tested two general approaches for providing VAF using eye movement data (real-time usage of eye movement vs. off-line records of eye movement)
4	form and function	Design Cycle 2	 Based on existing literature, I derived two theoretically grounded design principles (DPs) for individualized VAF that support the data exploration task. Also, I evaluated the proposed design quantitatively in a laboratory experiment. DP1: Provide the information dashboard with the functionality to monitor the users' eye movement in real-time to analyze the users' attention, given that the user visually focuses the information dashboard. DP2: Provide the information dashboard with the functionality to display individualized VAF based on the monitoring of the users' eye movement to support users in improving information processing performance. I suggested design features to support data exploration tasks on dashboards by providing dwall time on each chart as a time formation.
	Principle of	Design Cycle 3	 Based on existing literature, I derived three theoretically grounded design principles (DPs) for individualized VAF for supporting task resumption after an interruption. I also evaluated the proposed design and the role of WMC after a short-term interruption of a data exploration task, quantitatively in a laboratory experiment. DP1 (similar to DP1 in Cycle 2): Provide the information dashboard with the functionality to monitor the users' eye movement in real-time to analyze the users' attention, given that the user visually focuses the information dashboard. DP2: Provide the information dashboard with the ability to identify attention shifts to secondary task and resumption of primary task. DP3: Provide the information dashboard with gaze-based TRS in order to support users to resume their primary task. I suggested design features to support task resumption after an interruption by providing three gaze-based highlighting methods (heatmap, scanpath, last point).

Table 8.4.: Nascent design theory following Gregor and Jones (2007): principle of form and function.

СЛ		Design Cycle 1	 I discussed the mutability of information dashboards that provide VAF for data exploration, and the actual instantiation of the design in three software artifacts: Artifact 1: an information dashboard with individualized VAF. Artifact 2: an information dashboard with an example of proper attention allocation (off-line eye movement data). Artifact 3: an information dashboard with an example of improper attention allocation (off-line eye movement data).
	tifact mutability	Design Cycle 2	 I discussed the mutability of information dashboards that provide VAF for data exploration, and the actual instantiation of the design in two software artifacts: Artifact 1: an information dashboard with individualized VAF (real-time eye movement data). Artifact 2: an information dashboard with text-based explanations about ARA challenges as general VAF. I discuss the system architecture for attentive information dashboard that works with eye tracking technology and provide data exploration support.
	Ar	Design Cycle 3	 I discussed the mutability of the information dashboard that supports task resumption after an interruption, and the actual instantiation of the design in three software artifacts: Artifact 1: an information dashboard that provides heatmap highlighting method as gaze-based TRS. Artifact 2: an information dashboard that provides scanpath highlighting method as gaze-based TRS. Artifact 3: an information dashboard that provides last point highlighting methods as gaze-based TRS. I discuss the system architecture for attentive information dashboard that works with eye tracking technology and provide gaze-based TRS.
		Design Cycle 1	 Study I: Explorative eye tracking study to investigate the role of limited attention and working memory while exploring information dashboards. Study II: Explorative eye tracking study on the usage of off-line and real-time eye movement data as VAF.
6	Testable propositions	Design Cycle 2	 I formulated three hypotheses to test the effect of individualized VAF from three perspectives: H1: Providing individualized VAF results in better ARA performance in the revisit phase in comparison to providing generic VAF. H2: Providing individualized VAF results in lower ASR in the revisit phase in comparison to providing generic VAF. H3: Providing individualized VAF results in better ARM performance at the end of the task in comparison to providing generic VAF.
		Design Cycle 3	• Explorative eye tracking study on the role of WMC on the effectiveness of highlighting methods (heatmap, scanpath, last point) as gaze-based TRS.

Table 8.5.: Nascent design theory following Gregor and Jones (2007): artifact mutability and testable propositions.

According to the DSR knowledge contribution framework by Gregor and Hevner (2013), this thesis and the conducted DSR project is an improvement since I successfully developed new solutions for existing problems. There are two major problems addressed in this DSR project. First, the difficulty of users in managing limited attentional resources while exploring information dashboards. Investigating existing research and the results from the exploratory study as Study II highlight this problem. Furthermore, the results from the SLR performed in Study I highlight that there is a lack of solutions for this problem by integrating real-time tracking of eye movement data. The IS researchers have called for the usage of eye trackers to design innovative systems (Davis et al., 2014; Dimoka et al., 2012; Riedl and Léger, 2016; vom Brocke et al., 2013), and enhance users' cognitive limitations. However, no study has examined using eve movement data for designing VAF for IS applications (Lux et al., 2018). Therefore, as a solution, in design cycle 2, I designed and developed an attentive information dashboard with eve tracking technology. In this solution, I used the users' eve movement data to design individualized VAF and increase users' awareness of their previous visual behavior. The results from a large-scale laboratory experiment show that this solution could enhance users' ability to manage their limited attentional resources while exploring information dashboards.

The second problem focuses on the frequent interruptions of task execution in workplaces (Czerwinski et al., 2004; Mark et al., 2008), and the difficulty of employees to resume an interrupted task (Hemp, 2009; Hodgetts et al., 2015; Mark et al., 2005). The result from the Study II highlights this problem for information dashboard users. Leveraging eye movement data to detect an interruption and provide TRS by highlighting previously attended areas proved to support users in task resumption (Göbel and Kiefer, 2019; Jo et al., 2015; Kern et al., 2010; Mariakakis et al., 2015). However, investigating the papers in the Study I shows that existing gaze-based TRS studies did not examine the interruption on information dashboards and data exploration tasks so far. Furthermore, existing studies have centered on the last point approach as highlighting method for gaze-based TRS, and the heatmap and scanpath were not investigated. Besides, none of the studies investigated the role of WMC on the effectiveness of gaze-based TRS. As a solution, I designed and developed an attentive information dashboard that supports task resumption after short-term IT-mediated interruption by providing gaze-based TRS (last point, heatmap, scanpath) as an individualized VAF. I investigated this solution in design cycle 3 and evaluated it in a large-scale laboratory experiment. The results show that the need for gaze-based TRS is different for users with high and low WMC. Notably, the heatmap highlighting method is supportive for low WMC users, while users with high WMC may not need gaze-based TRS for short-term IT-mediated interruptions.

Furthermore, this DSR project contributes to the field of IS research that focuses on integrating neurophysiological tools for IS application as NeuroIS (Dimoka et al., 2012). The IS researchers have emphasized the need for integration of DSR and NeuroIS field of designing innovative IS applications (Riedl and Léger, 2016; vom Brocke et al., 2013). Based on the list of possible contributions in the field of NeuroIS provided by Riedl and Léger (2016), the designed attentive information dashboard related to the eighth contribution, using eye tracking as NeuroIS tool and delivering an IT artifact which tracks and adapts to the user's attentional state. Moreover, this DSR project contributes to the ninth contribution by providing individualized VAF as a live bio-feedback that assists users to control their limited attentional resources better.



Figure 8.1.: DSR knowledge contribution based on Gregor and Hevner (2013).

8.5. Limitations and Future Research

All five studies in this thesis were conducted, emphasizing rigor, and relevance. However, some limitations need to be considered and addressed in future research. In the following, I first discuss limitations and future work for the SLR study conducted as Study I in this thesis. Later I investigate the limitations and possible future research for the DSR project, including the Studies II, III, IV, and V from different perspectives.

Study I

Due to the time limitation considered in selecting eye-based IIS papers for the SLR study, research studies before 2009 were not included in the review. As eye tracking studies have a longer history than that, there can be some promising studies that have not been considered in this SLR study. Therefore, I suggest investigating the history of eye-based IIS without any time limitation. In this study, I first provided a general conceptual framework and then analyzed the collected papers based on that. This framework was improved during the analysis phase and in iterative ways. Furthermore, I included opinions from knowledgeable researchers in conducting SLR studies and eye tracking. However, it might miss some aspects of designing eye-based IIS or detailed categories for each dimension (for example, reasoning dimension can include several subdimensions). As future work, there is a need to evaluate this framework in detail. Researchers can also develop a framework based on

collected studies and compare it with the suggested one. Especially for some specific types of eye-based IIS, there may be more particular dimensions. Therefore, as future work, there is a possibility to focus on particular types of eye-based IIS (e.g., support collaboration, support interruption, summarization, etc.), provide a conceptual framework, and perform a SLR study for each perspective. As another limitation, the article collection and coding procedures were primarily done by only one researcher and might be biased. Therefore, no intercoder-reliability could be calculated, which can result in slightly different findings by other researchers. As future work, I suggest checking the intercoder-reliability by including more researchers in analyzing the papers. Furthermore, in this SLR study, I only focus on providing descriptive analysis about different dimensions of the eye-based IIS. Further studies can include relationship analysis between these dimensions by conducting a metaanalysis on collected data.

DSR Project

First, I presented the limitations and possible future works based on the apparatus used in this DSR project. The Tobii 4C eye tracker was the primary apparatus for all four studies (Studies II, III, IV and V) of the DSR project. I used this device for both design and evaluation purposes. Regarding the design purpose, this apparatus is known as a low-cost eye tracker and is mainly designed for eye-based interactive features. Also, the results from the Study I show that this apparatus is the most popular one for designing eve-based IIS. Such capability supports designing attentive information dashboards by tracking user's eye movement data in real-time and providing the individualized VAF that helps attention management. However, for the evaluation part, there are several devices in the market for evaluating user's visual behavior with higher accuracy than Tobii 4C eye tracker. By integrating devices with higher accuracy, there is a possibility of having access to high-resolution eye tracking data and promoting further aspects of user's visual behavior while working with individualized VAF. As an example, I needed to assign big AOIs while evaluating user's eye movement data. I considered each chart as an AOI and reported the analysis results based on that in all four studies. However, how the users allocated attention inside each chart with smaller AOIs can also include valuable information. Also, I did not use any eye movement analysis software to record and analysis eye movement data for the conducted studies. For Studies II and III, I used recorded dwell-time based on an integrated timer in the experimental software. For Study IV I used PyGaze (Dalmaijer et al., 2014) to extract fixation related values and for Study V, Tobii Pro SDK is used. All approaches represent the users' attention (Duchowski, 2017; Holmqvist et al., 2011) and the usage of techniques was stable within each study. In the future, to compare the users' visual behavior between studies, I suggest integrating one approach. Besides that, this device was limited to eye movement metrics and not accurate regarding pupil relevant data. By integrating an apparatus with higher accuracy, future research can extract pupil dilation to investigate user's cognitive load while using individualized VAF (Buettner et al., 2018; Fehrenbacher and Djamasbi, 2017; Perkhofer and Lehner, 2019). Also, using eye trackers for either designing or evaluation only includes recording overt attention. Duchowski (2017, p.13) has pointed out: "... in all eye tracking work ... we assume that attention is linked to foveal gaze direction, but we acknowledge that it may not always be so". To covert attention besides overt attention, future studies can use eye trackers beside biosignals like EEG (Léger et al., 2014) for evaluation purposes. Also, Roda and Thomas (2006) suggested to use eye trackers beside biosignals like heart rate, EEG, brain signals with fMRI, etc. for designing AUIs.

Second, I want to emphasize limitations and possible future works with regards to the information dashboard design. The information dashboards used in all four studies of the DSR project (Studies II, III, IV, and V) are not representing a real-world dashboard design. They all have the same design but with different content (check Appendices B, C, D, E). I selected this design for all studies to explore the users' goal-directed attention by controlling stimulus-driven attention. However, features to derive stimulus-driven attention (color, size, chart types, etc.) play an important role in the effectiveness of dashboards in realworld (Pauwels et al., 2009; Yigitbasioglu and Velcu, 2012). Besides the visual elements, I controlled for interactive features (e.g., filtering, zooming, etc.) in these dashboards as well, and all dashboards were static. Most dashboards in the market benefit from such features to support users in exploring information from different perspectives and affects its effectiveness (Pearson, 2013). For Study II, I decided to control these features since this is the fundamental study investigating the effects of limited attentional resources and WMC. However, these findings can be limited to the designed dashboard and difficult to generalize to all existing dashboards in the market. In future work, there is a need to investigate the role of the users' limited attention and WMC on the existing dashboard in the market. In studies III, IV and V, the same dashboard layout is used since I wanted to control the users' attention allocation in the revisit phase and identify the effects of VAF types. However, these effects can also be limited to this dashboard design, and there is a need to investigate such dashboard VAF types in with a real-world information dashboard.

Third, I want to mention some limitations with regards to the performed experiments:

• The participants in all four studies (Studies II, III, IV, V) of this DSR project are dominated by students. They were not daily users of information dashboards, which limit the generalizability of the results. Therefore, they have – similarly to novice users – little or no prior knowledge of the underlying experiment's process. Students can be considered an adequate and representative sample in the experimental setup (Burton-Jones and Meso, 2008). However, the users experience can play a role in managing limited attentional resources and their eye movements (Gegenfurtner et al., 2011). The users that already have experience with information dashboards may better allocate their attention to data exploration tasks. Therefore, the effectiveness of individualized VAF can be different for them. I suggest collecting eye movement data from users who already work with information dashboards and comparing the results as future work. Furthermore, long-term effects of attention management systems and their effects on human well-being are unknown (Anderson et al., 2018). Therefore, I suggest investigating the usage of attentive information dashboards and the suggested individual VAF types in long-term as field studies.

- The sample size of Studies II and III can be seen as a limitation in the case of a confirmatory study. Study I is considered an exploratory study to derive MRs for designing attentive information dashboards by exploring the users' visual behavior while working with dashboards. Study II is also considered as an eye tracking pilot study to compare using real-time or off-line records of eye movement data as VAF. Such studies with eye tracking focusing on problem identification require fewer participants than a study without eye tracking (Bojko, 2013, p. 163). Also, the review from Caine (2016) shows that the mean sample size for in-person studies using the eye tracking method is 21. Therefore, as all measures are derived from eye movement data in these two studies, I argue that the sample size in these two studies is synched with the first design cycle's purposes. However, as future work, there is an opportunity to conduct these studies with larger sample size for robust theorization and confirmatory studies.
- The results are limited to the data exploration task. In reality, the usage of an information dashboard is diverse, and users may use the information dashboards for visual search task besides data exploration. Furthermore, in the real-time BI&A systems, they may use it for monitoring or controlling (Lerch and Harter, 2001; Negash and Gray, 2008). Also the users may use information dashboards for planning to simulate various business scenarios (what if analysis) or in groups to ease the communications (Pauwels et al., 2009; Yigitbasioglu and Velcu, 2012). As the future work for Study II, I suggest investigating the role of limited attention and WMC of information dashboard users for other tasks. Also, VAF types for Study III, IV, and V can be tested for other tasks beyond data explorations.
- In Studies II and III, all users received VAF, and it was created following an automatic invocation approach. This can limit the generalization of the results since users may need different invocation styles for such feedback. Gregor and Benbasat (1999) mentioned invocation styles as user-invoked, automatic, and intelligent. Further research should be done on different invocation styles for individualized VAF. Also, the time to provide VAF was fixed for all the participants, and they received it during the data exploration task. Based on Morana et al. (2017), there are three types of timing for providing such feedback, including concurrently, prospectively, and retrospectively. As future work, there is an opportunity to investigate the effectiveness of individualized VAF for data exploration in these different timings.
- I all studies, I measured the quality of information processing by considering the users' overall management of the attentional resources, rather than by the quality of their task performance. Previous studies have shown that the pattern of attention can explain the task performance (Bera et al., 2019). Engaging users to the information on dashboards may allow users to extract and remember more detail information (Healey and Enns, 2012). I suggest expanding the experimental setup with further constructs to investigate the influence of individualized VAF on users' performance.

- In Study V, the results are limited to the effects of three gaze-based TRS highlighting methods under consideration of different WMC as user characteristics, short-term external interruption. This limits the ability to generalize the findings for other types of tasks, interruptions, context, and user characteristics. Therefore, I suggest testing the provided highlighting methods in different situations as future work. As an example, there is a need to investigate how providing such gaze-based TRS supports users in long-term interruptions. Furthermore, I compared the effects of gaze-based TRS types on users with low and high WMC separately. As future work, it is possible to compare the impact of gaze-based TRS between two groups. Also, I divided the groups into high and low WMC following the previous categorization of these individual characteristics in IS research (Lerch and Harter, 2001). As future work, it is possible to consider the user's WMC values for the analysis rather than the categorization of the users.
- In Study II and V, the role of visuospatial WMC is investigated as individual characteristics and presented the results based on this individual difference. As future work, there is a possibility to investigate other cognitive abilities of the users such as encoding strategy, digit working memory span, personality, etc. while exploring information dashboards.
- All the studies are conducted in a controlled lab environment, limiting the generalizing of the results for real-world setup. Therefore, I suggest integrating the individualized VAF for data exploration and task resumption in a real-world information dashboard and test them in laboratory and field studies.

Fourth, the results from Studies III, IV, and V are limited to the VAF design. The design that uses for Study III and Study IV to support the data exploration task includes the gaze duration on each chart to remind users about their previous visual behavior. There is an opportunity to investigate different gaze visualizations (e.g., heatmap, scanpath, etc.) as another highlighting method for individualized VAF that supports data exploration. For instance, we suggested designing a dashboard called "AttentionBoard" and presented the initial steps of designing it by providing MRs and DPs (Langner et al., 2020). Based on that, an individualized VAF is in the form of dashboards and presents dynamic heatmap and scanpath to show previous attention allocation of the information dashboard users from different views. This type of individualized VAF can be useful for both data exploration and resuming interrupted tasks.

Fifth, this DSR project is limited to provide individualized VAF for data exploration and resuming interrupted tasks. However, there are several other situations in which users of information dashboards may need attention support features. For instance, in global organizations, many meetings take place virtually using collaboration technologies, and a major requirement for successful collaboration is allocating joint attention. A lack of joint attention leads to misunderstandings or inefficient virtual teamwork. Therefore, I suggest designing attentive information dashboards to support remote collaboration by sharing the user's gaze position to overcome the lack of joint attention. For that, Toreini et al. (2018a) present a prototype and a pilot evaluation study to explore the effect of sharing gaze in a dyadic collaboration with an information dashboard. Also, findings from SLR study as Study I in this thesis shows different types of attention support features that can be used for information dashboard users.

Sixth, this DSR project is limited to integrate real-time eye movement data for information dashboard users. However, there is a possibility to design and test individualized VAF for other IS applications. For instance, Hummel et al. (2018) have suggested the usage of eye tracking technology to create attentive nudges with individualized VAF to support users not to miss information provided on nudges. Also, the example studies provided in the SLR study as the Study I in this thesis can be beneficial for existing IS applications.

Seventh, as the results from Study I show, an essential driver for adopting systems that integrate real-time eye movement data is the implementation of privacy features (Kunze et al., 2013; Steil et al., 2019). The developed attentive information dashboard and individualized VAF types in this study do not incorporate any mechanism for privacy aspects. As future work, there is a need to ensure that such systems respect the ethical and privacy aspects and considers the well-being of their users (Anderson et al., 2018).

9. Conclusion ¹

Human attention is known as a limited resource (Broadbent, 1958; Chun et al., 2011), and managing limited attentional resources is a challenge for users in the informationrich age (Gausby, 2015; Simon, 1971). However, having access to the enormous amount of data from various resources is beneficial for companies. With the use of big data technologies, companies collect and analyze data from various resources to assist users in making better decisions with BI&A systems (Chen et al., 2012). BI&A systems extract data from various resources, analyze it, derive insights, and present them to decisionmakers in the form of information dashboards (Pauwels et al., 2009; Yigitbasioglu and Velcu, 2012) which are known as one of the most effective BI&A tools (Negash and Gray, 2008). The challenge for organizations is not to collect more information, but to use the information in an effective way (Lerch and Harter, 2001). Information dashboards come with strong potentials to support better decision making by providing information from different perspectives, but they challenge their users' limited attentional resources since they include a huge amount of information (Alberts, 2017; Dilla et al., 2010; Lurie and Mason, 2007; Yigitbasioglu and Velcu, 2012). Existing research on BI&A systems is focused on their business significance and widespread use and provide solutions regarding users' challenges in managing limited attentional resources is a research gap (Browne and Parsons, 2012; Chen and Lee, 2003; Davern et al., 2012; Niu et al., 2013). The main research question of this thesis was focused on, "How to design attentive information dashboards for BI&A systems that enhance users' ability to manage attentional resources?". In order to answer this research question, five break-down research questions were formulated and answered step-by-step in five studies. Besides, as users' eyes are known as a proper resource to track their attention, this thesis focused on designing attentive information dashboards by integrating eye tracking technology. Eye tracking technology has matured considerably in recent years, primarily because of the availability of cheaper, faster, more accurate, and easier to use eye trackers (Duchowski, 2017). Researchers have suggested to integrate eye trackers technology beyond evaluation purposes and use it for designing supportive features for BI&A systems (Silva et al., 2019). Also, the usage of this technology to design innovative IS applications is suggested by IS researchers (Davis et al., 2014; Dimoka et al., 2012; vom Brocke et al., 2013). Furthermore, HCI researchers have considered it as the primary technology for designing AUIs (Bulling, 2016; Majaranta and Bulling, 2014).

In the Study I, I conducted a SLR study on previous research focusing on eye-based interactive intelligent systems. As part of this study, a conceptual framework was developed, state-of-the-art and future research directions were identified. Building on these results and following DSR methodology, three design cycles for designing attentive information dashboards were performed. The first design cycle includes two studies: Study II investigated attention problems of information dashboard users under consideration of their individual

¹This Chapter is based on the following studies which are published or in work: Toreini and Morana (2017), Toreini et al. (2018c), Toreini et al. (2018b), Toreini and Langner (2019), Toreini et al. (2020c), Toreini and Maedche (2020), Toreini et al. (2020a)

WMC. The results showed that users with high and low WMC have different difficulties in managing their attentional resources. Study III evaluated different potential solutions for supporting data exploration tasks using eye movement data. These solutions were named VAF types that aimed to increase users' awareness about their ARA performance and let them improve it. The results showed that providing individualized VAF integrating real-time eye movement data supports users in managing their attentional resources better than general VAF with integrated off-line recordings of eye movement data. In the second design cycle, I designed and evaluated an attentive information dashboard for the data exploration task. In Study IV, theoretically grounded DPs were articulated, instantiated as a software artifact, and evaluated in a large-scale laboratory experiment. The results from analyzing the users' eye movement data reveals that the suggested DPs have a positive effect on users' ability to manage limited attentional resources during data exploration tasks. In the third design cycle, I investigated attentive information dashboards to support resuming interrupted tasks. Study V instantiated three software artifacts using different gaze-based highlighting methods (last point, heatmap, scanpath) and evaluated them in a large-scale laboratory experiment by considering short-term IT-mediated interruptions and the role WMC. The results demonstrate the need for personalization of such support under consideration of users' WMC.

According to the DSR knowledge contribution framework by Gregor and Hevner (2013), this thesis and the conducted DSR project is an improvement since it successfully developed new solutions (attentive information dashboards) to existing problems (managing limited attentional resources). Furthermore, this thesis contributes to the intersection of IS and HCI fields by providing prescriptive knowledge in the form of nascent design theory for designing attentive information dashboards. This knowledge can support both researchers and practitioners. The proposed attentive information dashboards are the first BI&A systems presented in the IS field that use real-time eye movement data for designing advanced built-in attention support functions. Practitioners can leverage the findings from this thesis for integrating attentive IS applications that support users in managing limited attentional resources. I believe the findings of this thesis will serve as a reference for researchers and practitioners that focus on enhancing users' attention management capability by designing innovative IS applications with eye tracking technology.

Appendix

A. Study I

A.1. List of Papers for Literature Review

- 1. Fujii and Rekimoto (2019)
- 2. Kütt et al. (2019)
- 3. Bozkir et al. (2019)
- 4. Akkil et al. (2018)
- 5. Zhang et al. (2017)
- 6. D'Angelo and Begel (2017)
- 7. Newn et al. (2017)
- 8. Akkil and Isokoski (2016)
- 9. Kajan et al. (2016)
- 10. Toyama et al. (2015)
- 11. Wetzel et al. (2014)
- 12. Booth et al. (2013)
- 13. Dostal et al. (2013)
- 14. Ishii et al. (2013)
- 15. Pomarjanschi et al. (2012)
- 16. Buscher et al. (2012)
- 17. Alt et al. (2012)
- 18. Xu et al. (2008a)
- 19. Cheng et al. (2010)
- 20. Vrochidis et al. $\left(2011\right)$
- 21. D'Mello et al. (2012)
- 22. Cai and Lin (2012)
- 23. Kern et al. (2010)
- 24. Cheng et al. (2018)

- 25. Nguyen and Liu (2016)
- 26. Tremblay et al. (2018)
- 27. D'Mello et al. (2017)
- 28. Deza et al. (2017)
- 29. Mariakakis et al. (2015)
- 30. Hutt et al. (2019)
- 31. Taylor et al. (2015)
- 32. Jo et al. (2015)
- 33. Xu et al. (2008b)
- 34. D'Angelo and Gergle (2016)
- 35. Rozado et al. (2015)
- 36. Garrido et al. (2014)
- 37. Giannopoulos et al. (2012)
- 38. Umemoto et al. $\left(2012\right)$
- 39. Trösterer et al. (2015)
- 40. Siirtola et al. (2019)
- 41. Mills et al. (2019)
- 42. D'Angelo and Gergle (2018)
- 43. Qvarfordt et al. (2010)
- 44. Neider et al. (2010)
- 45. Higuch et al. (2016)
- 46. Schneider and Pea (2013)
- 47. Brennan et al. (2008)

A.2. Coding Tables for Literature Review

	User Context							Physical Context											Task										
	S	Bocia	al	U	ser]	Prof	ile	Type Digital Applications														r -	Гуре	е					
Authors	Individual Level	User Agent	Team Level	Age	Gender	Experience Level	Vision Status	Digital Env.	Real-World Env.	Driving Simulator	Websites	Exploration Tool	Multi-Display Software	AR/VR Application	Collaboration Tool	Gaming Software	Cellphone App	Training Platforms	Reading Tool	Gaming	Reading	Writing	Browsing / Search	Monitoring	Driving	Remote Physical Task	Learning	Programming	
(Fujii and Rekimoto, 2019)	Χ				Х			Х										Х									Х		
(Kütt et al., 2019)			Х	Х	х	Х		Х							Х							Х							
(Bozkir et al., 2019)	Χ			Х	Х	Х		Х		Х								Х							Х				
(Akkil et al., 2018)			Х	Х	Х	Х			Х						Х											Х			
(Zhang et al., 2017)			Х	Χ	Х			Х				Х			Х								Х						
(D'Angelo and Begel, 2017)			Х					Х							Х													Х	
(Newn et al., 2017)			Х	Х	Х	Х		Х								Х				Х									
(Akkil and Isokoski, 2016)			Х	Х	Х			Х		Х															Х				
(Kajan et al., 2016)	Χ			Х	Х	Х		Х			Х												Х						
(Toyama et al., 2015)	Χ							Х						Х									Х						
(Wetzel et al., 2014)	Χ			Х		Х		Х								Х				Х									
(Booth et al., 2013)	Χ			Х	Х	Х			Х														Х						
(Dostal et al., 2013)	Χ				Х			Х					Х										Х						
(Ishii et al., 2013)		Х						Х							Х								Х						
(Pomarjanschi et al., 2012)	Χ			Х	Х	Х		Х		Х															Х				
(Buscher et al., 2012)	Χ			Х	Х	Х		Х			Х								Х		Х								
(Alt et al., 2012)	Χ			Х				x			Х												Х						
(Xu et al., 2009)	Χ							Х											Х		Х								
(Cheng et al., 2010)	Χ			Х	Х	Х		x			Х												Х						
(Vrochidis et al., 2011)	Χ							x			Х												Х						
(D'Mello et al., 2012)	Х			Х	x			Х										Х									Х		
(Cai and Lin, 2012)	Χ			Х	Х	Х		Х		Х															Х				
(Kern et al., 2010)	Χ			Х	Х			Х		Х													Х						
(Cheng et al., 2018)	Χ			Х			Х	Х											Х		Х								
(Nguyen and Liu, 2016)	Χ			Х		Х		Х										Х									Х		
(Tremblay et al., 2018)	Χ						Х	Х					Х											Х					
(D'Mello et al., 2017)	Х							Х											Х		Х								
(Deza et al., 2017)	Χ							Х										Х					Х						
(Mariakakis et al., 2015)	Χ			Х	Х	Х	Х	Х									Х				Х								
(Hutt et al., 2019)	Χ			Х	Х			Х										Х									Х		
(Taylor, 2015)	Χ						Х	Х								Х								Х					
(Jo et al., 2015)	Χ			Х	Х		Х	Х											Х		Х								
(Xu et al., 2008)	Χ							Х			Х												Х						
(D'Angelo and Gergle, 2016)			Х	Х	Х	Х	Х	Х								Х							Х						
(Rozado et al., 2015)	Χ			Х				Х			Х												Х						
(Garrido et al., 2014)	Χ			x	Х	Х		Х	_				Х			_		_						Х					
(Giannopoulos et al., 2012)	Χ			Х	Х	Х		Х				Х											Х						
(Umemoto et al., 2012)	Х	_	_		Х	Х		Х	_		Х	_		_		_		_					Χ			_	_	_	
(Trösterer et al., 2015)			Х	Х	Х	Х		Х		Х					Х										Х				
(Siirtola et al., 2019)			Х	Х	Х		Х	Х							Х								Χ						
(Mills et al., 2019)				Х	Х	Х		Х											Х		Х								
(D'Angelo and Gergle, 2018)			Х	Х	Х	Х		Х				Х			Х								Х						
(Qvarfordt et al., 2010)	Χ			Х	Х		Х	Х				Х											Х						
(Neider et al., 2010)			Х				Х	Х				Х			Х								Х						
(Higuch et al., 2016)			Х	Х				Х							Х											Χ			
(Schneider and Pea, 2013)			Х	Х	Х	Х		Х				Х			Х			Х									Х		
(Brennan et al., 2008)			Χ					Х				Χ			Х								Х						

Table A.1.: Coding table of eye-based IIS literature review: context and task.

	Eye Tracking Technology												Experimental Setu							
	Ap	opar	atus	s Typ	e			Hardware Provider	Pr	ice	Met	hod	Study Design					Size		
Authors	Desktop-Mounted	Head-Mounted	Webcam	Virtual Reality	Smartphone Cam.	Tobii	Others	Name	High-Cost	Low-Cost	Laboratory	Field	Between-Subject	Mixed Factorial	Within-Subject	Single User/Group	Case Study	No. of Users		
(Fujii and Rekimoto, 2019)	х				2	X		Tobii 4C		Х	Х		Х					6		
(Kütt et al., 2019)	x					x		Tobii 4C		Х	Х		х					40		
(Bozkir et al., 2019)				Х		-	Х	HTC-Vive with Pupil-Labs	Х		X		X					16		
(Akkil et al., 2018)	x			_		x	_	Tobii T60	X		Х			_	Х			24		
(Zhang et al., 2017)	х				2	X		Tobii EyeX		Х	Х				Х			16		
(D'Angelo and Begel, 2017)	x					x		Tobii TX300 / Tobii EyeX	Х	Х	Х				Х			24		
(Newn et al., 2017)	х				2	X		Tobii EyeX		Х	Х				Х			27		
(Akkil and Isokoski, 2016)	х				1	X		Tobii T60	Х		Х				Х			12		
(Kajan et al., 2016)	Х				2	X		Tobii TX300	Х		Х				Х			15		
(Toyama et al., 2015)		Х					Х	SMI Glasses/Brother AirScouter	Х		Х					Х		12		
(Wetzel et al., 2014)		х					Х	EyeLink II	Х		Х				Х			31		
(Booth et al., 2013)		Х					Х	SMI Glasses	Х		Х		Х					20		
(Dostal et al., 2013)			Х				Х	Webcam (self gaze detector)		Х		Х					Х	1		
(Ishii et al., 2013)	Х				2	X		Tobii X120	Х		Х					Х		10		
(Pomarjanschi et al., 2012)	Х						Х	SMI RED250	Х		Х		Χ					30		
(Buscher et al., 2012)	Х				2	X		Tobii 1750	Х		Х		Х					32		
(Alt et al., 2012)	Х				2	X		Tobii X120	Х		Х				Х			12		
(Xu et al., 2009)			Х				Х	Webcam (Opengazer)		Х	Х					Х		12		
(Cheng et al., 2010)	Х	Х					Х	ASL 6000 / ASL RS H6	Х		Х				Х			9		
(Vrochidis et al., 2011)	Х						Х	faceLAB 5		Х	Х					Х		8		
(D'Mello et al., 2012)	х				2	X		Tobii T60	Х		Х				Х			48		
(Cai and Lin, 2012)	Х				2	X		Tobii X50	Х		Х				Х			20		
(Kern et al., 2010)	Х				2	X		Tobii X120	Х		Х				Х			16		
(Cheng et al., 2018)		Х					Х	MultiGaze (self dev. device)		Х	Х					Х		11		
(Nguyen and Liu, 2016)	Х				2	X		Tobii EyeX		Х	Х		Χ					16		
(Tremblay et al., 2018)	Х	_	_	_			Х	SmartEye Pro	Х		Х		Х					99		
(D'Mello et al., 2017)	Х				2	X		Tobii TX300	Х		Х		Х					104		
(Deza et al., 2017)	Х	_				_	Χ	EyeLink 1000	Х		Χ				Χ			18		
(Mariakakis et al., 2015)					X		Х	Phone Cam. (self gaze detector)		Х	Х				Х			17		
(Hutt et al., 2019)	X	_		_	_		Χ	EyeTribe		Χ		Χ				Χ	Х	135		
(Taylor, 2015)	X						Х	GazePoint GP3		Х	X		Х					44		
(Jo et al., 2015)	X	_		_	-12	X		Tobii X60	X		X	_			X			13		
(Xu et al., 2008)			Х		-		X	Webcam (Opengazer)		X	X					Х		5		
(D'Angelo and Gergle, 2016)	X					r	X	Eyetribe		X	X		37		Х	_		36		
$\frac{(\text{Rozado et al., 2015})}{(\text{Rozado et al., 2014})}$	X		v		- 2	X	v			X	X		X		37	_		20		
$\frac{(\text{Garrido et al., 2014})}{(\text{Garrido et al., 2014})}$		37	X	_	-		X	Webcam (self gaze detector)	v	X	X	_			X	_		12		
(Giannopoulos et al., 2012)	os et al., 2012) X X Ergoneers Dikablis		Ergoneers Dikablis	X		X			_	X	v	_	40							
(Umemoto et al., 2012)	., 2012) X X Tobii T60		Consert First Data	A v		A v				v	Λ		8							
(110sterer et al., 2010)	2019) X X SmartEye Pro		A V	v	A V			v	Λ	_		20								
(Mills at al 2019)	x v					x. V	Λ	Tobii T X 300	X V	Λ	X V			X V				60		
(D'Angelo and Cergle 2018)	x x					x		Tobii TX300 Tobii 4C		x	X			X				96		
(Overfordt et al 2010)	x x					x		Tobii X190	x	Λ	x			Λ	x			8		
(Neider et al. 2010)	Λ	X				x	x	EveLink II	X		X		x		Λ			32		
(Higuch et al., 2016)	iguch et al. 2016) X X Tobij EyeX				Tobii EveX	A	x	X		X					8					
(Schneider and Pea. 2013)	X					~ X		Tobii X1		X	X		X					42		
(Brennan et al., 2008)		X			Ť	-	х	EveLink II	X		X		X					40		

Table A.2.: Coding table of eye-based IIS literature review: eye tracking technology and experimental setup.
									S	\mathbf{ens}	e										l	Rea	son	ing			Fo	cu
			Gaz	ze-ba	ased			F	ixat	ion-	base	ed	Sa	icca	de	Pu	ıpil	Otł	ners			Т]ype	,			Ту	pe
Authors	Gaze Position	Gaze Duration	Gaze Transition	Mutual Gaze	Visit Counts	Gaze Depth	Gaze Presence	Fixation Position	Fixation Duration	Number of Fixation	First Fixated	Last Fixated	Number of Saccades	Saccade Direction	Saccade Length	Pupil Dilation	Dilation Speed	Scanpath	Global Eye Feature	Engagement	Relevance/Interesting	Attention Switching	Uncertainty	Confidence	Mind Wandering	Intention	User Adaptation	System Adaptation
(Fujii and Rekimoto, 2019)										X			X								x							x
(Kütt et al., 2019)				X		-	_			77	_		77	_	_						X		_		_		X	11
(Bozkir et al., 2019)	X															X					X							X
(Akkil et al., 2018)	X																				X						X	
(Zhang et al., 2017)				Х						Х											X						X	
(D'Angelo and Begel, 2017)	X																				X				_		X	
(Newn et al., 2017)								х		Х								x			X				_		X	
(Akkil and Isokoski, 2016)	X																				X						Х	
(Kajan et al., 2016)		Х			Х					Х		Х									X							X
(Toyama et al., 2015)						X														X			_		_			X
(Wetzel et al., 2014)			Х						Х	Х			х					X			Х		Х	Х	_			X
(Booth et al., 2013)										Х											X							X
(Dostal et al., 2013)	X															Х					X	Х					Х	
(Ishii et al., 2013)		Х	Х	Х											Х	Х				Х								Χ
(Pomarjanschi et al., 2012)	Χ																				Х							Χ
(Buscher et al., 2012)									Х					Х	Х						Х							Х
(Alt et al., 2012)		Х							Х	Х										Х	Х							Х
(Xu et al., 2009)										Х											Х							Χ
(Cheng et al., 2010)									Х	Х	Х					Х					Х							Х
(Vrochidis et al., 2011)									Х	Х						Х	Х				Х							Х
(D'Mello et al., 2012)	Χ																			Х	Х							Х
(Cai and Lin, 2012)								Х												Х	Х							Х
(Kern et al., 2010)												Х									Х	Х					Х	
(Cheng et al., 2018)												Х									Х	Х					Х	
(Nguyen and Liu, 2016)		Х					Х	Х													Х	Х					Х	Х
(Tremblay et al., 2018)								Х	Х												Х	Х						Х
(D'Mello et al., 2017)																			Х						Х			Х
(Deza et al., 2017)								Х		Х			Х								Х							Χ
(Mariakakis et al., 2015)							Х					Х		Х							Х	Х				Х	Х	
(Hutt et al., 2019)																			Х						х			Х
(Taylor, 2015)																Х					Х	Х					Х	
(Jo et al., 2015)								Х				Х									Х						Х	
(Xu et al., 2008)								Х													Х							Х
(D'Angelo and Gergle, 2016)	Х																				Х						Х	
(Rozado et al., 2015)								Х													Х							Х
(Garrido et al., 2014)	Χ																					Х						Χ
(Giannopoulos et al., 2012)								Х	Х												Х						Х	
(Umemoto et al., 2012)	Χ																				Х					Х		Х
(Trösterer et al., 2015)	Χ																				Х						Х	
(Siirtola et al., 2019)	Χ																				Х						Х	
(Mills et al., 2019)																			Х						Х			Х
(D'Angelo and Gergle, 2018)				Х				Х	Х	Х								Х			Х						Х	
(Qvarfordt et al., 2010)		Х						Х	Х												Х						Х	
(Neider et al., 2010)								Х													Х						Х	
(Higuch et al., 2016)								Х													Х						Х	
(Schneider and Pea, 2013)	Х																				Х						Х	
	77																				77						-	

Table A.3.: Coding table of eye-based IIS literature review: sense-reasoning and focus.

			Perception Outcome							Behavior Outcome																	
	Sta	tus					ľ	Meas	sure	s					Sta	tus					Me	easu	res				
Authors	Yes	No	Distraction	Frustration	Challenge	Fun	Nervousness	Ease of Use	Satisfaction	Correctness	Usefulness	Mental Workload	Confidence	Collab. Quality	Yes	No	Engagement	Attention Allocation	Search Strategy	Change in Attitude	Reading Speed	Attention Shift	Gaze Overlap	Objective Workload	User's Interactions	Conversation Analysis	Global Bias
(Fujii and Rekimoto, 2019)	X										Х					х											
(Kütt et al., 2019)	X										Х			Х		Х											
(Bozkir et al., 2019)		Х													Х									Х			
(Akkil et al., 2018)	Χ													Х	Х											Х	
(Zhang et al., 2017)	Х		Х					Х			Х	Х				Х											
(D'Angelo and Begel, 2017)	X										Х				Х		Х						х				
(Newn et al., 2017)	Χ							Х			Х					Х											
(Akkil and Isokoski, 2016)	Х										Х		Х			Х											
(Kajan et al., 2016)	Χ															Х											
(Toyama et al., 2015)	Х										Х					Х											
(Wetzel et al., 2014)	Χ			Х	Х	Х	Х								Х			Х									
(Booth et al., 2013)		Х													Х			Х									
(Dostal et al., 2013)	Χ										Х				Х							Х			Х		
(Ishii et al., 2013)	Х													Х	Х		Х										
(Pomarjanschi et al., 2012)		Х													Х							Х			Х		
(Buscher et al., 2012)		Х														Х											
(Alt et al., 2012)		Х													Х		Х	Х		Х							
(Xu et al., 2009)		Х														Х											
(Cheng et al., 2010)	Х							Х	Х						Х									Х			
(Vrochidis et al., 2011)		Х														Х											
(D'Mello et al., 2012)	Х								Х						Х		Х					Х					
(Cai and Lin, 2012)	Х		Х						Х	Х	Х					Х											
(Kern et al., 2010)	Х								Х		Х					Х											
(Cheng et al., 2018)	Х								Х		Х				Х							Х					
(Nguyen and Liu, 2016)		Х													Х										Х		
(Tremblay et al., 2018)	Х											Х			Х			Х									
(D'Mello et al., 2017)	Х															Х											
(Deza et al., 2017)		Х													Х			Х									
(Mariakakis et al., 2015)	Х											Х			Х						Х	Х					
(Hutt et al., 2019)		Х														Х											
(Taylor, 2015)		Х													Х			Х						Х			Х
(Jo et al., 2015)	Χ										Х			_	Х			Х									
(Xu et al., 2008)		Х														Х											
(D'Angelo and Gergle, 2016)		Х		_	_	_	_	_	_				_		Х			_						_		Χ	
(Rozado et al., 2015)		Х														Х											_
(Garrido et al., 2014)		Х		_	_	_	_	_	_		_		_			Х		_					_	_			
(Giannopoulos et al., 2012)	X			_					Х		Х				Х										Х		
(Umemoto et al., 2012)		Х		_		_	_	_	_				_	_		Χ	_		_	_	_	_	_	_		_	
(Trösterer et al., 2015)	X		X	_		_				Χ	Χ	Χ			X			Χ						_		_	
(Siirtola et al., 2019)	X										X					X											_
(Mills et al., 2019)		Х														Х											
(D'Angelo and Gergle, 2018)	X								X			37			X			X.	X							Χ	
(Qvarfordt et al., 2010)	X	7.5							Х			Х			X			X	Х				11				
(Neider et al., 2010)	37	Х									v			Xr.	X			X					Х				
(Figuen et al., 2016)	X		v								Х			X	X			X					v	v			v
(Schneider and Pea, 2013)	X	37	X											Χ	X			X	v	_			Χ	Χ		v	X
(Drennan et al., 2008)		A													A			Χ	Χ							Λ	

Table A.4.: Coding table of eye-based IIS literature review: perception and behavior.

			User Performance								System Performance					
	Sta	itus			Μ	leasur	es			Sta	Status Measures					
Authors	Yes	No	Reaction Time	Completion Time	Driving	Self Report	Comprehension	Effectiveness	Accuracy	Y_{es}	No	Accuracy	Running Time	Network Traffic		
(Fujii and Rekimoto, 2019)	Х								Х		Х					
(Kütt et al., 2019)	X			Х					X		X					
(Bozkir et al., 2019)	Х				Х						Х					
(Akkil et al., 2018)	Х			Χ							Х					
(Zhang et al., 2017)	Х			Х					Х		Х					
(D'Angelo and Begel, 2017)	Х			Х		Х			Х	Х			Х			
(Newn et al., 2017)	Х								Х		Х					
(Akkil and Isokoski, 2016)	Х				Х					Х		Х				
(Kajan et al., 2016)	Х			Χ						Х		Х				
(Toyama et al., 2015)		Х								x		Х				
(Wetzel et al., 2014)		Х									Х					
(Booth et al., 2013)	Х			Х					Х		Х					
(Dostal et al., 2013)		Х									Х					
(Ishii et al., 2013)	Х								Х	Х		Х				
(Pomarjanschi et al., 2012)	Х		Х						Х		Х					
(Buscher et al., 2012)		Х								Х		Х				
(Alt et al., 2012)		Х									Χ					
(Xu et al., 2009)		Х								Χ		Х				
(Cheng et al., 2010)	Χ			Χ						Χ		Χ				
(Vrochidis et al., 2011)		Х								Х		Χ				
(D'Mello et al., 2012)	Х					Χ			Х		Χ					
(Cai and Lin, 2012)	Χ								Х		Χ					
(Kern et al., 2010)	Х			Χ							Χ					
(Cheng et al., 2018)	Χ								Х		Χ					
(Nguyen and Liu, 2016)	Х			Х					Х		Х					
(Tremblay et al., 2018)		Χ									Χ		_			
(D'Mello et al., 2017)	Χ						Х			Χ		Χ				
(Deza et al., 2017)	Χ						Χ		Χ		Χ					
(Mariakakis et al., 2015)	Х						Х			Х		Х				
(Hutt et al., 2019)		X								X		X	_			
(Taylor, 2015)	X		X						X		X					
(Jo et al., 2015)	X								Χ		X		_			
(Xu et al., 2008)		Х								Х		X				
(D'Angelo and Gergle, 2016)	Х			Х					Х		Х					
(Rozado et al., 2015)		Х								Х			Х	Х		
(Garrido et al., 2014)	Х								Х		Χ					
(Giannopoulos et al., 2012)	Х			Х				Х			Х					
(Umemoto et al., 2012)		Х								Х		Χ				
(Trösterer et al., 2015)	Х				Х						Χ					
(Siirtola et al., 2019)	Х		Х						Х		Χ					
(Mills et al., 2019)	Х			Χ		Χ	Х				Χ					
(D'Angelo and Gergle, 2018)	Х		Х	Х							Х					
(Qvarfordt et al., 2010)	Х								Х		Х					
(Neider et al., 2010)	X		Х						X		X					
(Higuch et al., 2016)	Х			Х					Х		Х					
(Schneider and Pea, 2013)	Х						Х				Х					
(Brennan et al., 2008)	Х		Х								Х					

Table A.5.: Coding table of eye-based IIS literature review: performance.

B. Study II

B.1. Experimental Software



Figure B.1.: The information dashboard layout used in Study II.



Figure B.2.: The break page as an interruption used in Study II.

B.2. Further Analysis

	First Visit Phase	Revisit Phase	End				
High WMC							
Low WMC							

Figure B.3.: The heatmaps of users with high and low WMC in three phases of the experiment in Study II.

C. Study III

C.1. Experimental Software



Figure C.1.: The information dashboard layout used in the first round of Study III.



Figure C.2.: The information dashboard layout used in the second round in Study III.



Figure C.3.: Example of VAF design used in the second round in Study III.



C.2. Further Analysis

Figure C.4.: The heatmaps of users in first round of the experiment in Study III.



Figure C.5.: The heatmaps of users in second round of experiment in Study III.

D. Study IV

D.1. Experimental Software



Figure D.1.: Simplified information dashboard used as the learning phase in Study IV.



Figure D.2.: The information dashboard layout used in Study IV.

173



Figure D.3.: An example of individualized VAF used in Study IV.



Figure D.4.: The general VAF as a text-based explanation used in Study IV.

D.2. Survey Items

Controls	Items	Measure
Age	• Please specify your age	-
Gender	• Please specify your gender	-
Experience Level	 Using the BI&A dashboard is a new experience for me. Using the BI&A dashboard is not similar to anything that I've done before. Using the BI&A dashboard is different from other experiences I have had. 	Moore, G.C. (1989) Likert Scale 1 to 7

Table D.1.: Control variables and items used for Study IV.

D.3. Further Analysis

Control Variable	Conditions	Mdn	Mean	SD	W	Р	R		
	Individualized VAF	22.5	22.77	2.75	051 5	0.10	0.167		
Age	General VAF	24	24.06	3.8	851.5	0.10	-0.167		
	Individualized VAF	5	5	1.38		0.11	0.169		
Experience Level	General VAF 5.67 5.42 1.43 850.5				0.11	-0.163			
WMC – Corsi	Individualized VAF	6	6.01	0.84	1100 5	0 50	0.000		
Span	Span General VAF 5.5 5.94 1.09 1139.5 0.50 -0.063								
$\mathbf{WMC}-\mathbf{Digit}$	Individualized VAF	7	7.32	1.27	10745	0.00	0.015		
span	General VAF	7	7.25	1.33	1074.5	0.88	-0.015		
Note: * p<0.05, **p<0.01									

Table D.2.: Comparing the control variables – wilcoxon signed-rank test.



Figure D.5.: Distribution of fixation duration values before and after VAF types.



Figure D.6.: Distribution of number of fixations before and after VAF types.



Figure D.7.: ARM of the users in the first and end of the data exploration tasks based on fixation duration values.



Figure D.8.: ARM of the users in the first and end of the data exploration tasks based on number of fixation values.

E. Study V



E.1. Experimental Software

Figure E.1.: First information dashboard used in Study V.



Figure E.2.: Second information dashboard used in Study V.



Figure E.3.: Third information dashboard used in Study V.



Figure E.4.: Fourth information dashboard used in Study V.

You received an email. Please read and answer that	\bowtie
Email Title: Welcome!	
This is Larry, I am the executive assistant of the COO. We will meet each other soon in the next board meeting, but I just thought to write you an email before that to welco	me you to DOPA.
We are so excited to have you on board. Other employees are looking forward to helping you find your way in our company and learn more about what we do here. We consider the some of your amazing skills. We believe that you will be a great asset to our company.	an't wait to see your work and show off
Also, let me know if our team can be of any assistance. Please know that many of us moved here in the past few years and we know how hard it can be to grow accustomed help you out and give you recommendations on neighborhoods, schools, restaurants, gyms, and everything else you might want to know.	l to a new environment. We are here to
As the first step, I and some collogues are going to the gym on Tuesdays and Thursdays from 6:00 – 7:00 PM. Are you interested in sports? Do you like to join us for the ne email about that	xt week? Just drop me an answer to this
We are like a big family in our company and we look forward to meeting you and your family once you settle in.	
Looking forward to working with you,	
Best, Larry	
1	
	Sand
	send

Figure E.5.: First e-mail used as an IT-mediated Interruption in Study V.

You received an email. Please read and answer that	\searrow
Email Title: What do you like to eat?	
This is Ginni, office assistant of the CEO and his team. This is just a quick note to tell you that our whole team is excited about your decision to accept our offer of employm you to the team.	ent. We couldn't be happier to welcome
Besides the meeting that you need to prepare using the BIA dashboard, for the rest of your first day, I will provide you with additional information about our team. Our go In your new role. We have put together a schedule for your first week. You will also meet your mentor, Paul Smith. He'll help you get to know the company. Also, you will a benefits and complete the new employee aparework.	al is that you get immediately productive ttend an HR orientation workshop about
Your new team anticipates taking you out to lunch to get to know you and to make sure that you meet everyone with whom you will be working. There are several good re just tell us what type of food you like? We need to reserve a desk for that	estaurants close to the office. Can you
-Pizza	
-Burger	
-Sushi	
Again, welcome to the team. If you have questions, please call me at any time, or send me an email, if that is more convenient.	
Best, Ginni	
	Send

Figure E.6.: Second e-mail used as an IT-mediated Interruption in Study V.

You received an email. Please read and	d answer that
Email Title: Please send us your information soon	
This is John from the IT department of DOPA. We are all really excited to welcome you to our team! We care about giving our employees everything they need to perform their best. We have prepared your workplace with computer, software, and accounts. Your new online account includes the company email and calendar, modern commun monitors, a phone, a mouse, a keyboard, and a headset at your workplace. One thing which is left is granting you access to all conference rooms! We could not manage it because of some technica you access to enter the conference room before your important meeting.	ih all the necessary IT equipment. Our team is committed to helping you set up your nication and collaboration tools, office software. Besides that, you should have a laptop, two al issues in the last days sorry about that! The problem is already solved and we can give
For that, could you please send me the following information:	
2- Year of birth (only year)	
Thanks	
Best, John	
1	
	Send

Figure E.7.: Third e-mail used as an IT-mediated Interruption in Study V.

You received an email. Please read and answer that	\searrow
Email Title: Which workshop do you want to join?	
This is Alice from the internal education department of DOPA. Welcome to the team! We are thrilled to have you at DOPA.	
For your first week, we have planned a few training sessions in the field of business intelligence analytics (BIA). We run several BIA tool introduction workshops for the ne upcoming workshop is focused on "PowerBI Desktop". As you might know, "Power BI Desktop" is a free application you can install on the local computer that lets you con data. With Power BI Desktop, you can connect to multiple different sources of data, and combine them (offen called modeling) into a data model that lets you build report people inside your organization. Most users who work on BIA projects use Power BI Desktop to create reports, and then use the Power BI service to share their reports wit	w arrivals in the next weeks. The next nect to, transform, and visualize your ts. You can share reports with other th others.
We offer three different workshops for users based on their previous knowledge about BIA systems. Could you tell us which group should we assign you?	
A) Entry-level – I do not have an idea about BIA systems	
B) Intermediate level – I know basics about BIA systems	
C) Advanced level – I worked with different BIA systems before, but not PowerBI Desktop	
Best, Alice	
	Send

Figure E.8.: Fourth e-mail used as an IT-mediated Interruption in Study V.



Figure E.9.: The fixation cross position before first visit in Study V.



Figure E.10.: The fixation cross position after gaze-based TRS in Study V.



Figure E.11.: The control condition in Study V in which users do not receive any gaze-based TRS.

E.2. Survey Items

Controls	Items	Measure
Perceived Usefulness	 "Using this feedback would improve my performance to resume the exploration task after answering the email." "Using this feedback would increase my productivity on resuming the interrupted data exploration task." "Using this feedback would make it easier to resume my interrupted data exploration task." "I would find this feedback useful in resuming the interrupted exploration task." 	Davis (1989) Likert Scale 1 to 7
Easy to Use	 "Learning to operate this attention feedback would be easy for me." "I would find this attention feedback easy to use." "It would be easy to become skillful at using this attention feedback." 	Davis (1989) Likert Scale 1 to 7
Self- Awareness	 "Due to this attention feedback, I'm conscious of my data exploration strategy." "Due to this attention feedback, I'm reflective of my data exploration strategy." "Due to this attention feedback, I'm aware of my data exploration strategy." 	Twenge et al. (2007) Likert Scale 1 to 7
Behavioral Intention	 "I intend to use this feedback in the future to support me in resuming my interrupted task." "I predict I would use this feedback in the future to support me in resuming my interrupted task." "I plan to use this feedback in the future to support me in resuming my interrupted task." 	Venkatesh et al. (2003) Likert Scale 1 to 7

Table E.1.: Items used in the final survey of Study V.

	First Visit Phase (30 Sec.)	Mandatory Revisit Phase (30 Sec.)
None		
Heatmap		
Scanpath		
Last Point		

E.3. Further Analysis

Figure E.12.: The heatmaps of users with low WMC in first visit and mandatory revisit phases of the experiment in Study V.



Figure E.13.: The heatmaps of users with high WMC in first visit and mandatory revisit phases of the experiment in Study V.

References

- Adamczyk, P. D. and Bailey, B. P. (2004). If Not Now, when?: The Effects of Interruption at Different Moments Within Task Execution. In *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*, CHI '04, pages 271–278. ACM Press.
- Addas, S. (2010). A call for engaging context in HCI/MIS research with examples from the area of technology interruptions. AIS Transactions on Human-Computer Interaction, 2(4):178–196.
- Addas, S. and Pinsonneault, A. (2015). The many faces of information technology interruptions: A taxonomy and preliminary investigation of their performance effects. *Information Systems Journal*, 25(3):231–273.
- Ahn, J.-H., Bae, Y.-S., Ju, J., and Oh, W. (2018). Attention Adjustment, Renewal, and Equilibrium Seeking in Online Search: An Eye-Tracking Approach. *Journal of Management Information Systems*, 35(4):1218–1250.
- Akkil, D. and Isokoski, P. (2016). Gaze Augmentation in Egocentric Video Improves Awareness of Intention. In Proceedings of the 2016 CHI Conference on Human Factors in Computing Systems, CHI '16, pages 1573–1584. ACM Press.
- Akkil, D., Thankachan, B., and Isokoski, P. (2018). I see what you see: Gaze Awareness in Mobile Video Collaboration. In *Proceedings of the 2018 ACM Symposium on Eye Tracking Research & Applications*, ETRA '18, pages 1–9. ACM Press.
- Al-Samarraie, H., Sarsam, S. M., Alzahrani, A. I., and Alalwan, N. (2018). Personality and individual differences: the potential of using preferences for visual stimuli to predict the Big Five traits. *Cognition, Technology & Work*, 20(3):337–349.
- Alberts, A. (2017). Eye-tracking study: 5 key learnings for data designers everywhere.
- Alt, F., Shirazi, A. S., Schmidt, A., and Mennenöh, J. (2012). Increasing the user's attention on the web. In *Proceedings of the 7th Nordic Conference on Human-Computer Interaction Making Sense Through Design*, NordiCHI '12, page 544. ACM Press.
- Altmann, E. M. and Trafton, J. G. (2002). Memory for goals: an activation-based model. Cognitive Science, 26(1):39–83.
- Altmann, E. M. and Trafton, J. G. (2004). Task interruption: Resumption lag and the role of cues. In Proceedings of the 26th Annual Conference of the Cognitive Science Society.
- Anderson, C., Hübener, I., Seipp, A.-K., Ohly, S., David, K., and Pejovic, V. (2018). A Survey of Attention Management Systems in Ubiquitous Computing Environments.

Proceedings of the ACM on Interactive, Mobile, Wearable and Ubiquitous Technologies, 2(2):1–27.

- Atkinson, R. C. and Shiffrin, R. M. (1968). Human Memory: A Proposed System and its Control Processes. Psychology of Learning and Motivation, 2(4):89–195.
- Bačić, D. and Fadlalla, A. (2016). Business information visualization intellectual contributions: An integrative framework of visualization capabilities and dimensions of visual intelligence. *Decision Support Systems*, 89:77–86.
- Baddeley, A. (1992). Working Memory. Science, 255(5044):556-559.
- Baddeley, A. (2000). The episodic buffer: a new component of working memory? *Trends* in Cognitive Sciences, 4(11):417–423.
- Baddeley, A. (2012). Working Memory: Theories, Models, and Controversies. Annual Review of Psychology, 63(1):1–29.
- Baddeley, A. and Hitch, G. (1974). Working Memory. In Psychology of Learning and Motivation, volume 8, pages 47–89.
- Bailey, B. P. and Iqbal, S. T. (2008). Understanding changes in mental workload during execution of goal-directed tasks and its application for interruption management. ACM Transactions on Computer-Human Interaction, 14(4):1–28.
- Bailey, B. P. and Konstan, J. A. (2006). On the need for attention-aware systems: Measuring effects of interruption on task performance, error rate, and affective state. *Computers in Human Behavior*, 22(4):685–708.
- Baker, J., Jones, D. R., and Burkman, J. (2009). Using Visual Representations of Data to Enhance Sensemaking in Data Exploration Tasks. *Journal of the Association for Information Systems*, 10(7):533–559.
- Balzer, W. K., Doherty, M. E., and O'Connor, R. (1989). Effects of cognitive feedback on performance. *Psychological Bulletin*, 106(3):410–433.
- Baskett, L., LeRouge, C., and Tremblay, M. C. (2008). Using the dashboard technology properly. *Health Progress*, 89(5):16–23.
- Bednarik, R. and Tukiainen, M. (2006). An eye-tracking methodology for characterizing program comprehension processes. In *Proceedings of the 2006 Symposium on Eye Tracking Research & Applications*, ETRA '06, pages 125–132. ACM Press.
- Behrisch, M., Streeb, D., Stoffel, F., Seebacher, D., Matejek, B., Weber, S. H., Mittelstaedt, S., Pfister, H., and Keim, D. (2018). Commercial Visual Analytics Systems-Advances in the Big Data Analytics Field. *IEEE Transactions on Visualization and Computer Graphics*, 25(10):3011–3031.
- Benyon, D. (2013). Designing interactive systems: A comprehensive guide to HCI, UX and interaction design. Pearson Higher.
- Bera, P. (2014). Do distracting dashboards matter? Evidence from an eye tracking study. Information Systems: Education, Applications, Research. SIGSAND/PLAIS 2014. Lecture Notes in Business Information Processing, 193:65–74.
- Bera, P. (2016). How colors in business dashboards affect users' decision making. Communications of the ACM, 59(4):50–57.
- Bera, P., Soffer, P., and Parsons, J. (2019). Using Eye Tracking to Expose Cognitive Processes in Understanding Conceptual Models. *MIS Quarterly*, 43(4):1105–1126.

- Berkovsky, S., Taib, R., Koprinska, I., Wang, E., Zeng, Y., Li, J., and Kleitman, S. (2019). Detecting Personality Traits Using Eye-Tracking Data. In *Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems*, CHI '19, pages 1–12. ACM Press.
- Bixler, R. and D'Mello, S. K. (2016). Automatic gaze-based user-independent detection of mind wandering during computerized reading. User Modeling and User-Adapted Interaction, 26(1):33–68.
- Blascheck, T., Kurzhals, K., Raschke, M., Burch, M., Weiskopf, D., and Ertl, T. (2014). State-of-the-Art of Visualization for Eye Tracking Data. In *Eurographics Conference* on Visualization, pages 1–20.
- Bojko, A. (2013). Eye Tracking the User Experience: A Practical Guide to Research. Rosenfeld Media.
- Booth, T., Sridharan, S., McNamara, A., Grimm, C., and Bailey, R. (2013). Guiding attention in controlled real-world environments. In ACM Symposium on Applied Perception 2013, SAP '13, page 75. ACM Press.
- Borkin, M. A., Bylinskii, Z., Kim, N. W., Bainbridge, C. M., Yeh, C. S., Borkin, D., Pfister, H., and Oliva, A. (2016). Beyond Memorability: Visualization Recognition and Recall. *IEEE Transactions on Visualization and Computer Graphics*, 22(1):519– 528.
- Borkin, M. A., Vo, A. A., Bylinskii, Z., Isola, P., Sunkavalli, S., Oliva, A., and Pfister, H. (2013). What makes a visualization memorable? *IEEE Transactions on Visualiza*tion and Computer Graphics, 19(12):2306–2315.
- Borst, J. P., Taatgen, N. A., and van Rijn, H. (2015). What Makes Interruptions Disruptive? A Process-Model Account of the Effects of the Problem State Bottleneck on Task Interruption and Resumption. In *Proceedings of the 33rd Annual ACM Conference on Human Factors in Computing Systems*, volume 1 of *CHI '15*, pages 2971–2980. ACM Press.
- Bozkir, E., Geisler, D., and Kasneci, E. (2019). Assessment of Driver Attention during a Safety Critical Situation in VR to Generate VR-based Training. In ACM Symposium on Applied Perception 2019, SAP '19, pages 1–5. ACM.
- Brennan, S. E., Chen, X., Dickinson, C. A., Neider, M. B., and Zelinsky, G. J. (2008). Coordinating cognition: The costs and benefits of shared gaze during collaborative search. *Cognition*, 106(3):1465–1477.
- Broadbent, D. E. (1958). Perception and communication. Pergamon Press, Elmsford, NY.
- Browne, G. and Parsons, J. (2012). More Enduring Questions in Cognitive IS Research. Journal of the Association for Information Systems, 13(12):1000–1011.
- Buettner, R., Sauer, S., Maier, C., and Eckhardt, A. (2018). Real-time Prediction of User Performance based on Pupillary Assessment via Eye-Tracking. AIS Transactions on Human-Computer Interaction, 10(1):26–60.
- Bulling, A. (2016). Pervasive Attentive User Interfaces. Computer, 49(1):94–98.
- Bulling, A. and Gellersen, H. (2010). Toward mobile eye-based human-computer interaction. *IEEE Pervasive Computing*, 9(4):8–12.
- Bulling, A., Roggen, D., and Tröster, G. (2011). What's in the Eyes for Context-

Awareness? IEEE Pervasive Computing, 10(2):48–57.

- Burch, M., Heinrich, J., Konevtsova, N., Höferlin, M., and Weiskopf, D. (2011). Evaluation of traditional, orthogonal, and radial tree diagrams by an eye tracking study. *IEEE Transactions on Visualization and Computer Graphics*, 17(12):2440–2448.
- Burton, L., Albert, W., and Flynn, M. (2014). A Comparison of the Performance of Webcam vs. Infrared Eye Tracking Technology. Proceedings of the Human Factors and Ergonomics Society Annual Meeting, 58(1):1437–1441.
- Burton-Jones, A. and Meso, P. N. (2008). The effects of decomposition quality and multiple forms of information on novices' understanding of a domain from a conceptual model. *Journal of the Association for Information Systems*, 9(12):748–802.
- Buscher, G., Dengel, A., Biedert, R., and Elst, L. V. (2012). Attentive documents: Eye tracking as implicit feedback for information retrieval and beyond. ACM Transactions on Interactive Intelligent Systems, 1(2):1–30.
- Cai, H. and Lin, Y. (2012). Coordinating Cognitive Assistance With Cognitive Engagement Control Approaches in Human–Machine Collaboration. *IEEE Transactions on Systems, Man, and Cybernetics - Part A: Systems and Humans*, 42(2):286–294.
- Caine, K. (2016). Local Standards for Sample Size at CHI. In Proceedings of the 2016 CHI Conference on Human Factors in Computing Systems, CHI '16, pages 981–992. ACM Press.
- Cane, J. E., Cauchard, F., and Weger, U. W. (2012). The time-course of recovery from interruption during reading: Eye movement evidence for the role of interruption lag and spatial memory. *Quarterly Journal of Experimental Psychology*, 65(7):1397– 1413.
- Card, S. K. (1983). The Human Information Processor. Crc Press.
- Carrasco, M. (2011). Visual attention: The past 25 years. Vision Research, 51(13):1484–1525.
- Chaudhuri, S. and Dayal, U. (1997). An overview of data warehousing and OLAP technology. ACM SIGMOD Record, 26(1):65–74.
- Chen, G. and Kotz, D. (2000). A Survey of Context-Aware Mobile Computing Research. Technical report, Department of Computer Science, Dartmouth College.
- Chen, H., Chiang, R. H. L., and Storey, V. C. (2012). Business Intelligence and Analytics: From Big Data to Big Impact. *MIS Quarterly*, 36(4):1165.
- Chen, J. Q. and Lee, S. M. (2003). An exploratory cognitive DSS for strategic decision making. *Decision Support Systems*, 36(2):147–160.
- Chen, S., Epps, J., and Chen, F. (2013). Automatic and continuous user task analysis via eye activity. In *Proceedings of the 2013 international conference on Intelligent user* interfaces - IUI '13, page 57, New York, New York, USA. ACM Press.
- Cheng, S., Fan, J., and Dey, A. K. (2018). Smooth Gaze: a framework for recovering tasks across devices using eye tracking. *Personal and Ubiquitous Computing*, 22(3):489– 501.
- Cheng, S., Liu, X., Yan, P., Zhou, J., and Sun, S. (2010). Adaptive user interface of product recommendation based on eye-tracking. In *Proceedings of the 2010 workshop on Eye* gaze in intelligent human machine interaction, EGIHMI '10, pages 94–101. ACM

Press.

- Cheung, M. Y. M., Hong, W., and Thong, J. Y. (2017). Effects of animation on attentional resources of online consumers. *Journal of the Association for Information Systems*, 18(8):605–632.
- Choe, E. K., Lee, N. B., Lee, B., Pratt, W., and Kientz, J. A. (2014). Understanding quantified-selfers' practices in collecting and exploring personal data. In *Proceedings* of the SIGCHI Conference on Human Factors in Computing Systems, CHI'14, pages 1143–1152. ACM Press.
- Chuang, L., Duchowski, A. T., Qvarfordt, P., and Weiskopf, D. (2019). Ubiquitous Gaze Sensing and Interaction (Dagstuhl Seminar 18252). In Blascheck, T., editor, *Dagstuhl Reports*, volume 8, pages 77–148.
- Chun, M. M., Golomb, J. D., and Turk-Browne, N. B. (2011). A Taxonomy of External and Internal Attention. Annual Review of Psychology, 62(1):73–101.
- Cindi, H., James, R., Rita, S., and Austin, K. (2019). Magic Quadrant for Business Intelligence Platforms. Technical report, Gartner Inc.
- Cöltekin, A., Fabrikant, S. I., and Lacayo, M. (2010). Exploring the efficiency of users' visual analytics strategies based on sequence analysis of eye movement recordings. *International Journal of Geographical Information Science*, 24(10):1559–1575.
- Conati, C., Lallé, S., Rahman, M. A., and Toker, D. (2017). Further Results on Predicting Cognitive Abilities for Adaptive Visualizations. In *Proceedings of the Twenty-Sixth International Joint Conference on Artificial Intelligence*, IJCAI-17, pages 1568–1574.
- Conway, A. R. A., Kane, M. J., Bunting, M. F., Hambrick, D. Z., Wilhelm, O., and Engle, R. W. (2005). Working memory span tasks: A methodological review and user's guide. *Psychonomic Bulletin & Review*, 12(5):769–786.
- Conway, A. R. A., Kane, M. J., and Engle, R. W. (2003). Working memory capacity and its relation to general intelligence. *Trends in Cognitive Sciences*, 7(12):547–552.
- Corbetta, M. and Shulman, G. L. (2002). Control of Goal-Directed and Stimulus-Driven Attention in the Brain. *Nature Reviews Neuroscience*, 3(3):215–229.
- Cowan, N. (2010). The magical mystery four: How is working memory capacity limited, and why? *Current Directions in Psychological Science*, 19(1):51–57.
- Czerwinski, M., Cutrell, E., and Horvitz, E. (2000). Instant messaging and interruption: Influence of task type on performance. In *Proceedings of the OZCHI 2000*, pages 356–361.
- Czerwinski, M., Horvitz, E., and Wilhite, S. (2004). A diary study of task switching and interruptions. In Proceedings of the SIGCHI Conference on Human Factors in Computing Systems, CHI '04, pages 175–182. ACM Press.
- Dabbish, L., Mark, G., and González, V. M. (2011). Why do i keep interrupting myself?: environment, habit and self-interruption. In *Proceedings of the 2011 Annual Con*ference on Human Factors in Computing Systems, CHI '11, pages 3127–3130. ACM Press.
- Dalmaijer, E. (2014). Is the low-cost EyeTribe eye tracker any good for research? Technical report, PeerJ PrePrints.
- Dalmaijer, E. S., Mathôt, S., and Van der Stigchel, S. (2014). PyGaze: An open-source,

cross-platform toolbox for minimal-effort programming of eyetracking experiments. Behavior Research Methods, 46(4):913–921.

- D'Angelo, S. and Begel, A. (2017). Improving Communication Between Pair Programmers Using Shared Gaze Awareness. In Proceedings of the 2017 CHI Conference on Human Factors in Computing Systems, CHI '17, pages 6245–6290. ACM Press.
- D'Angelo, S. and Gergle, D. (2016). Gazed and Confused:Understanding and Designing Shared Gaze for Remote Collaboration. In *Proceedings of the 2016 CHI Conference* on Human Factors in Computing Systems, CHI '16, pages 2492–2496. ACM Press.
- D'Angelo, S. and Gergle, D. (2018). An Eye For Design: Gaze Visualizations for Remote Collaborative Work. In Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems, volume 2018-April of CHI '18, pages 1–12. ACM Press.
- Davenport, T. H. and Beck, J. C. (2001). The Attention Economy. Ubiquity, (May):1.
- Davern, M., Shaft, T., and Te'eni, D. (2012). Cognition Matters: Enduring Questions in Cognitive IS Research. Journal of the Association for Information Systems, 13(4):273–314.
- Davis, F. D. (1989). Perceived usefulness, perceived ease of use, and user acceptance of information technology. MIS Quarterly, 13(3):319–339.
- Davis, F. D., Riedl, R., and Hevner, A. R. (2014). Towards a NeuroIS Research Methodology: Intensifying the Discussion on Methods, Tools, and Measurement. *Journal of* the Association for Information Systems, 15(10):I–XXXV.
- Delen, D. and Ram, S. (2018). Research challenges and opportunities in business analytics. Journal of Business Analytics, 1(1):2–12.
- Deng, X. and Chi, L. (2012). Understanding Postadoptive Behaviors in Information Systems Use: A Longitudinal Analysis of System Use Problems in the Business Intelligence Context. Journal of Management Information Systems, 29(3):291–326.
- Desimone, R. and Duncan, J. (1995). Neural Mechanisms of Selective Visual Attention. Annual Review of Neuroscience, 18(1):193–222.
- Deza, A., Peters, J. R., Taylor, G. S., Surana, A., and Eckstein, M. P. (2017). Attention Allocation Aid for Visual Search. In *Proceedings of the 2017 CHI Conference on Human Factors in Computing Systems*, CHI '17, pages 220–231. ACM Press.
- Dilla, W., Janvrin, D. J., and Raschke, R. (2010). Interactive Data Visualization: New Directions for Accounting Information Systems Research. *Journal of Information* Systems, 24(2):1–37.
- Dimoka, A., Davis, F. D., Pavlou, P. A., and Dennis, A. R. (2012). On the Use of Neurophysiological Tools in IS Research: Developing a Research Agenda for NeuroIS. *MIS Quarterly*, 36(3):679 – 702.
- Dinter, B., Schieder, C., and Gluchowski, P. (2015). A stakeholder lens on metadata management in business intelligence and big data Results of an empirical investigation.
 In Twenty-first Americas Conference on Information Systems, AMCIS 2015.
- D'Mello, S., Olney, A., Williams, C., and Hays, P. (2012). Gaze tutor: A gaze-reactive intelligent tutoring system. International Journal of Human-Computer Studies, 70(5):377–398.
- Doshi, A. and Trivedi, M. M. (2009). On the roles of eye gaze and head dynamics in

predicting driver's intent to change lanes. *IEEE Transactions on Intelligent Trans*portation Systems, 10(3):453–462.

- Dostal, J., Kristensson, P. O., and Quigley, A. (2013). Subtle gaze-dependent techniques for visualising display changes in multi-display environments. In *Proceedings of the* 2013 International Conference on Intelligent User Interfaces, IUI '13, pages 137–147. ACM Press.
- Dragunov, A. N., Dietterich, T. G., Johnsrude, K., McLaughlin, M., Li, L., and Herlocker, J. L. (2005). TaskTracer: A Desktop Environment to Support Multi-tasking Knowledge Workers. In *Proceedings of the 10th international conference on Intelligent user interfaces*, IUI '05, pages 75–82. ACM Press.
- Drewes, H., De Luca, A., and Schmidt, A. (2007). Eye-gaze interaction for mobile phones. In Proceedings of the 4th international conference on mobile technology, applications, and systems and the 1st international symposium on Computer human interaction in mobile technology, Mobility '07, pages 364–371. ACM Press.
- Driver, J. (2001). A selective review of selective attention research from the past century. British Journal of Psychology, 92(1):53–78.
- Duchowski, A. T. (2002). A breadth-first survey of eye-tracking applications. Behavior Research Methods, Instruments, & Computers, 34(4):455–470.
- Duchowski, A. T. (2017). *Eye Tracking Methodology: Theory and Practice*. Springer International Publishing, Cham, 3 edition.
- Duchowski, A. T., Cournia, N., and Murphy, H. (2004). Gaze-Contingent Displays: A Review. *CyberPsychology & Behavior*, 7(6):621–634.
- Duchowski, A. T., Krejtz, K., Krejtz, I., Biele, C., Niedzielska, A., Kiefer, P., Raubal, M., and Giannopoulos, I. (2018). The Index of Pupillary Activity. In *Proceedings of* the 2018 CHI Conference on Human Factors in Computing Systems, CHI '18, pages 1–13. ACM Press.
- Dybdal, M. L., Agustin, J. S., and Hansen, J. P. (2012). Gaze input for mobile devices by dwell and gestures. In *Proceedings of the Symposium on Eye Tracking Research and Applications*, ETRA'12, pages 225–228. ACM Press.
- D'Mello, S. K., Mills, C., Bixler, R., and Bosch, N. (2017). Zone out no more: Mitigating mind wandering during computerized reading. In *Proceedings of the 10th International Conference on Educational Data Mining (EDM)*, pages 8–15.
- Einhorn, H. J. and Hogarth, R. M. (1981). Behavioral Decision Theory: Processes of Judgement and Choice. Annual Review of Psychology, 32(1):53–88.
- Elbashir, M. Z., Collier, P. A., and Davern, M. (2008). Measuring the effects of business intelligence systems: The relationship between business process and organizational performance. *International Journal of Accounting Information Systems*, 9(3):135– 153.
- Engle, R. W. (2002). Working Memory Capacity as Executive Attention. Current Directions in Psychological Science, 11(1):19–23.
- Engle, R. W., Kane, M. J., and Tuholski, S. W. (1999). Individual Differences in Working Memory Capacity and What They Tell Us About Controlled Attention, General Fluid Intelligence, and Functions of the Prefrontal Cortex. In Miyake, A. and Shah,

P., editors, Models of working memory: Mechanisms of active maintenance and executive control, pages 102–134. Cambridge University Press.

- Eriksen, C. W. and Yeh, Y.-Y. (1985). Allocation of attention in the visual field. *Journal* of Experimental Psychology: Human Perception and Performance, 11(5):583–597.
- Eskilsson, H. (2019). Don't Blink: Eye Tracking is Coming. https://blog.tobii.com/dontblink-eye-tracking-is-coming-f07b855995be.
- Farnsworth, B. (2019). Eye Tracker Prices An Overview of 20+ Eye Trackers.
- Fayyad, U., Piatetsky-Shapiro, G., and Smyth, P. (1996). Knowledge Discovery and Data Mining: Towards a Unifying Framework. In Proceedings of the Second International Conference on Knowledge Discovery and Data Mining, KDD'96, page 82–88. AAAI Press.
- Fehrenbacher, D. D. and Djamasbi, S. (2017). Information systems and task demand: An exploratory pupillometry study of computerized decision making. *Decision Support* Systems, 97:1–11.
- Few, S. (2006). Information Dashboard Design, The Effective Visual Communication of Data. O'Reilly Media, 1 edition.
- Figl, K. and Laue, R. (2011). Cognitive Complexity in Business Process Modeling. In Mouratidis, H. and Rolland, C., editors, Advanced Information Systems Engineering. CAiSE 2011. Lecture Notes in Computer Science, volume 6741, pages 452–466. Springer, Berlin.
- Fink, L., Yogev, N., and Even, A. (2017). Business intelligence and organizational learning: An empirical investigation of value creation processes. *Information & Management*, 54(1):38–56.
- Foroughi, C. K., Werner, N. E., McKendrick, R., Cades, D. M., and Boehm-Davis, D. A. (2016). Individual differences in working-memory capacity and task resumption following interruptions. *Journal of Experimental Psychology: Learning, Memory, and Cognition*, 42(9):1480–1488.
- Fujii, K. and Rekimoto, J. (2019). SubMe: An Interactive Subtitle System with English Skill Estimation Using Eye Tracking. In *Proceedings of the 10th Augmented Human International Conference*, AH2019, pages 1–9. ACM Press.
- Galer, S. (2019). SAP arbeitet mit Start-up zusammen an Eye-Tracking. https://news.sap.com/germany/2019/03/eye-tracking-start-up/.
- Galluch, P. S., Grover, V., and Thatcher, J. B. (2015). Interrupting the workplace: Examining stressors in an information technology context. *Journal of the Association* for Information Systems, 16(1):1–47.
- Garrido, J. E., Penichet, V. M. R., Lozano, M. D., Quigley, A., and Kristensson, P. O. (2014). AwToolkit: Attention-Aware User Interface Widgets. In Proceedings of the 2014 International Working Conference on Advanced Visual Interfaces, AVI '14, pages 9–16. ACM Press.
- Gausby, A. (2015). Attention Spans. Technical report, Consumer Insights Lead, Microsoft Canada.
- Gegenfurtner, A., Lehtinen, E., and Säljö, R. (2011). Expertise Differences in the Comprehension of Visualizations: A Meta-Analysis of Eye-Tracking Research in Professional

Domains. Educational Psychology Review, 23(4):523–552.

- Giannopoulos, I., Kiefer, P., and Raubal, M. (2012). GeoGazemarks. Proceedings of the 14th ACM international conference on Multimodal interaction - ICMI '12, (20120036433):165.
- Gillie, T. and Broadbent, D. (1989). What makes interruptions disruptive? A study of length, similarity, and complexity. *Psychological research*, 50(4):243–250.
- Göbel, F. and Kiefer, P. (2019). POITrack: improving map-based planning with implicit POI tracking. In *Proceedings of the 11th ACM Symposium on Eye Tracking Research* & Applications, ETRA '19, pages 1–9. ACM Press.
- Goldhaber, M. H. (1997). The attention economy and the Net. First Monday, 2(4).
- Gregor, S. (2006). The Nature of Theory in Information Systems. *MIS Quarterly*, 30(3):611–642.
- Gregor, S. and Benbasat, I. (1999). Explanations From Intelligent Systems: Theoretical Foundations and Implications for Practice. MIS Quarterly, 23(4):497–530.
- Gregor, S. and Hevner, A. R. (2013). Positioning and Presenting Design Science Research for Maximum Impact. *MIS Quarterly*, 37(2):337–355.
- Gregor, S. and Jones, D. (2007). The Anatomy of a Design Theory. Journal of the Association for Information Systems, 8(2):312–335.
- Günther, W. A., Rezazade Mehrizi, M. H., Huysman, M., and Feldberg, F. (2017). Debating big data: A literature review on realizing value from big data. *Journal of Strategic Information Systems*, 26(3):191–209.
- Gupta, A., Li, H., and Sharda, R. (2013). Should I send this message? Understanding the impact of interruptions, social hierarchy and perceived task complexity on user performance and perceived workload. *Decision Support Systems*, 55(1):135–145.
- Haroz, S. and Whitney, D. (2012). How capacity limits of attention influence information visualization effectiveness. *IEEE Transactions on Visualization and Computer Graphics*, 18(12):2402–2410.
- Haugtvedt, C. P. and Wegener, D. T. (1994). Message Order Effects in Persuasion: An Attitude Strength Perspective. Journal of Consumer Research, 21(1):205–2018.
- Hayes, T. R. and Henderson, J. M. (2018). Scan patterns during scene viewing predict individual differences in clinical traits in a normative sample. *PLOS ONE*, 13(5):e0196654.
- Hayhoe, M. and Ballard, D. (2005). Eye movements in natural behavior. Trends in Cognitive Sciences, 9(4):188–194.
- Healey, C. and Enns, J. (2012). Attention and visual memory in visualization and computer graphics. *IEEE Transactions on Visualization and Computer Graphics*, 18(7):1170– 1188.
- Hemp, P. (2009). Death by information overload. *Harvard business review*, 87(9):82—9, 121.
- Henderson, J. M., Shinkareva, S. V., Wang, J., Luke, S. G., and Olejarczyk, J. (2013). Predicting Cognitive State from Eye Movements. *PLoS ONE*, 8(5):e64937.
- Hevner, A. R., March, S. T., Park, J., and Ram, S. (2004). Design Science in Information Systems Research. MIS Quarterly, 28(1):75–105.

- Higuch, K., Yonetani, R., and Sato, Y. (2016). Can Eye Help You?: Effects of Visualizing Eye Fixations on Remote Collaboration Scenarios for Physical Tasks. In Proceedings of the 2016 CHI Conference on Human Factors in Computing Systems, CHI '16, pages 5180–5190. ACM Press.
- Hodgetts, H. M., Tremblay, S., Vallières, B. R., and Vachon, F. (2015). Decision support and vulnerability to interruption in a dynamic multitasking environment. *International Journal of Human Computer Studies*, 79:106–117.
- Holmqvist, K., Nyström, M., Andersson, R., Dewhurst, R., Jarodzka, H., and Van De Weijer, J. (2011). Eye Tracking: A comprehensive guide to methods and measures. Oxford University Press, Oxford.
- Holzinger, A. (2013). Human-Computer Interaction and Knowledge Discovery (HCI-KDD): What is the benefit of bringing those two fields to work together? In Cuzzocrea, A., Kittl, C., Simos, D., Weippl, E., and Xu, L., editors, Availability, Reliability, and Security in Information Systems and HCI. CD-ARES 2013. Lecture Notes in Computer Science, volume 8127, pages 319–328. Springer, Berlin, Heidelberg.
- Hong, W., Thong, J. Y., and Tam, K. Y. (2004a). Does animation attract online users' attention? The effects of flash on information search performance and perceptions. *Information Systems Research*, 15(1):60–86.
- Hong, W., Thong, J. Y., and Tam, K. Y. (2004b). The effects of information format and shopping task on consumers' online shopping behavior: A cognitive fit perspective. *Journal of Management Information Systems*, 21(3):149–184.
- Hoppe, S., Loetscher, T., Morey, S. A., and Bulling, A. (2018). Eye Movements During Everyday Behavior Predict Personality Traits. *Frontiers in Human Neuroscience*, 12:1–8.
- Hummel, D., Toreini, P., and Maedche, A. (2018). Improving Digital Nudging Using Attentive User Interfaces: Theory Development and Experiment Design Using Eyetracking. In Research in Progress Proceedings of the 13th International Conference on Design Science Research in Information Systems and Technology (DESRIST), pages 1–8, Chennai, India.
- Hutchinson, T., White, K., Martin, W., Reichert, K., and Frey, L. (1989). Humancomputer interaction using eye-gaze input. *IEEE Transactions on Systems, Man,* and Cybernetics, 19(6):1527–1534.
- Hutt, S., Krasich, K., Mills, C., Bosch, N., White, S., Brockmole, J. R., and D'Mello, S. K. (2019). Automated gaze-based mind wandering detection during computerized learning in classrooms. User Modeling and User-Adapted Interaction, 29(4):821–867.
- Iqbal, S. T. and Horvitz, E. (2007). Disruption and recovery of computing tasks. In Proceedings of the SIGCHI Conference on Human Factors in Computing Systems, CHI '07, pages 677–686. ACM Press.
- Ishii, R., Nakano, Y. I., and Nishida, T. (2013). Gaze awareness in conversational agents: Estimating a user's conversational engagement from eye gaze. ACM Transactions on Interactive Intelligent Systems, 3(2):1–25.
- Itti, L. and Koch, C. (2001). Computational modelling of visual attention. Nature Reviews

Neuroscience, 2(3):194–203.

- Jacob, R. J. K. (1991). The use of eye movements in human-computer interaction techniques: what you look at is what you get. ACM Transactions on Information Systems, 9(3):152–169.
- Jacob, R. J. K. and Karn, K. S. (2003). Eye Tracking in Human-Computer Interaction and Usability Research. In *The Mind's Eye*, pages 573–605. Elsevier.
- Jameson, A. and Riedl, J. (2011). Introduction to the Transactions on Interactive Intelligent Systems. ACM Transactions on Interactive Intelligent Systems, 1(1):1–6.
- Jo, J., Kim, B., and Seo, J. (2015). EyeBookmark: Assisting Recovery from Interruption during Reading. In Proceedings of the 33rd Annual ACM Conference on Human Factors in Computing Systems, CHI '15, pages 2963–2966. ACM Press.
- Jung, J. H., Schneider, C., and Valacich, J. (2010). Enhancing the Motivational Affordance of Information Systems: The Effects of Real-Time Performance Feedback and Goal Setting in Group Collaboration Environments. *Management Science*, 56(4):724–742.
- Just, M. A. and Carpenter, P. A. (1980). A theory of reading: From eye fixations to comprehension. *Psychological Review*, 87(4):329–354.
- Kahneman, D. (1973). Attention and Effort. Prentice-Hall, Inc., Englewood Cliffs, New Jersey.
- Kajan, R., Herout, A., Bednarik, R., and Povolný, F. (2016). PeepList: Adapting ex-post interaction with pervasive display content using eye tracking. *Pervasive and Mobile Computing*, 30:71–83.
- Kane, M. J., Bleckley, M. K., Conway, A. R. A., and Engle, R. W. (2001). A controlledattention view of working-memory capacity. *Journal of Experimental Psychology: General*, 130(2):169–183.
- Kane, M. J. and Engle, R. W. (2003). Working-memory capacity and the control of attention: The contributions of goal neglect, response competition, and task set to Stroop interference. *Journal of Experimental Psychology: General*, 132(1):47–70.
- Keim, D., Andrienko, G., Fekete, J.-d., Görg, C., Kohlhammer, J., and Melançon, G. (2008). Visual Analytics: Definition, Process, and Challenges. In Kerren, A., Stasko, J., Fekete, J., and North, C., editors, *Information Visualization. Lecture Notes in Computer Science*, pages 154–175. Springer, Berlin, Heidelberg, vol 4950 edition.
- Kelton, A. S., Pennington, R. R., and Tuttle, B. M. (2010). The Effects of Information Presentation Format on Judgment and Decision Making: A Review of the Information Systems Research. Journal of Information Systems, 24(2):79–105.
- Kern, D., Marshall, P., and Schmidt, A. (2010). Gazemarks Gaze-Based Visual Placeholders to Ease Attention Switching. In *Proceedings of the 28th International Conference* on Human Factors in Computing Systems, CHI '10, pages 2093–2102. ACM Press.
- Kessels, R. P. C., van Zandvoort, M. J. E., Postma, A., Kappelle, L. J., and de Haan, E. H. F. (2000). The Corsi Block-Tapping Task: Standardization and Normative Data. *Applied Neuropsychology*, 7(4):252–258.
- Kim, J., Seo, J., and Laine, T. H. (2018). Detecting boredom from eye gaze and EEG. Biomedical Signal Processing and Control, 46:302–313.
- Kitchenham, B. and Charters, S. (2007). Performing systematic literature reviews in

software engineering. Technical report, Keele University and Durham University Joint Report.

Kothari, J. (2019). Glass Enterprise Edition 2: faster and more helpful. https://www.blog.google/products/hardware/glass-enterprise-edition-2/.

Kowler, E. (2011). Eye movements: The past 25 years. Vision Research, 51(13):1457-1483.

- Kuechler, B. and Vaishnavi, V. (2008). Theory development in design science research: anatomy of a research project. *European Journal of Information Systems*, 17(5):489– 504.
- Kuechler, W. and Vaishnavi, V. (2012). A Framework for Theory Development in Design Science Research : Multiple Perspectives. Journal of the Association for Information Systems, 13(6):395–423.
- Kunze, K., Utsumi, Y., Shiga, Y., Kise, K., and Bulling, A. (2013). I know what you are reading: recognition of document types using mobile eye tracking. In *Proceedings of* the 17th annual international symposium on International symposium on wearable computers - ISWC '13, volume 145, page 113, New York, New York, USA. ACM Press.
- Kurzhals, K., Fisher, B., Burch, M., and Weiskopf, D. (2016). Eye tracking evaluation of visual analytics. *Information Visualization*, 15(4):340–358.
- Kütt, G. H., Lee, K., Hardacre, E., and Papoutsaki, A. (2019). Eye-Write: Gaze Sharing for Collaborative Writing. In *Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems (CHI '19)*, CHI '19, pages 1–12. ACM Press.
- Lai, M. L., Tsai, M. J., Yang, F. Y., Hsu, C. Y., Liu, T. C., Lee, S. W. Y., Lee, M. H., Chiou, G. L., Liang, J. C., and Tsai, C. C. (2013). A review of using eye-tracking technology in exploring learning from 2000 to 2012. *Educational Research Review*, 10(88):90–115.
- Lallé, S., Toker, D., Conati, C., and Carenini, G. (2015). Prediction of Users' Learning Curves for Adaptation while Using an Information Visualization. In *Proceedings* of the 20th International Conference on Intelligent User Interfaces, IUI '15, pages 357–368. ACM Press.
- Langner, M., Toreini, P., and Maedche, A. (2020). AttentionBoard: A Quantified-Self Dashboard for Enhancing Attention Management with Eye-Tracking (in press). In Davis, F., Riedl, R., vom Brocke, J., Léger, P., Randolph, A., and Fischer, T., editors, Information Systems and Neuroscience (NeuroIS Retreat 2020). Virtual Conference.
- Léger, P.-M., Sénécal, S., Courtemanche, F., Guinea, A., Titah, R., Fredette, M., and Labonte-LeMoyne, (2014). Precision is in the Eye of the Beholder: Application of Eye Fixation-Related Potentials to Information Systems Research. Journal of the Association for Information Systems, 15(10):651–678.
- Lerch, F. J. and Harter, D. E. (2001). Cognitive Support for Real-Time Dynamic Decision Making. Information Systems Research, 12(1):63–82.
- Lim, K. H., O'Connor, M. J., and Remus, W. E. (2005). The impact of presentation media on decision making: Does multimedia improve the effectiveness of feedback? *Information and Management*, 42(2):305–316.
- Liversedge, S. P. and Findlay, J. M. (2000). Saccadic eye movements and cognition. Trends

in Cognitive Sciences, 4(1):6-14.

- Lorigo, L., Haridasan, M., Brynjarsdóttir, H., Xia, L., Joachims, T., Gay, G., Granka, L., Pellacini, F., and Pan, B. (2008). Eye tracking and online search: Lessons learned and challenges ahead. Journal of the American Society for Information Science & Technology, 59(7):1041–1052.
- Lupton, D. (2016). The diverse domains of quantified selves: self-tracking modes and dataveillance. *Economy and Society*, 45(1):101–122.
- Lurie, N. H. and Mason, C. H. (2007). Visual Representation: Implications for Decision Making. Journal of Marketing, 71(1):160–177.
- Lux, E., Adam, M. T. P., Dorner, V., Helming, S., Knierim, M. T., and Weinhardt, C. (2018). Live Biofeedback as a User Interface Design Element: A Review of the Literature. Communications of the Association for Information Systems, 43:257–296.
- Mack, A. and Rock, I. (1998). Inattentional blindness. MIT Press.
- Maglio, P. P., Barrett, R., Campbell, C. S., and Selker, T. (2000). SUITOR: an attentive information system. In *Proceedings of the 5th international conference on Intelligent* user interfaces, IUI '00, pages 169–176. ACM Press.
- Majaranta, P. and Bulling, A. (2014). Eye Tracking and Eye-Based Human-Computer Interaction. In Fairclough, S. H.-.-.-p. and Gilleade, K., editors, Advances in Physiological Computing. Human-Computer Interaction Series, pages 39–65. Springer, London.
- Majaranta, P. and Räihä, K.-J. (2002). Twenty years of eye typing: systems and design issues. In Proceedings of the symposium on Eye tracking research & applications, ETRA '02, page 15. ACM Press.
- March, J. G. and Shapira, Z. (1987). Managerial Perspectives on Risk and Risk Taking. Management Science, 33(11):1404–1418.
- Mariakakis, A., Goel, M., Aumi, M. T. I., Patel, S. N., and Wobbrock, J. O. (2015). SwitchBack: Using Focus and Saccade Tracking to Guide Users' Attention for Mobile Task Resumption. In Proceedings of the 33rd Annual ACM Conference on Human Factors in Computing Systems, CHI '15, pages 2953–2962. ACM Press.
- Mark, G., González, V. M., and Harris, J. (2005). No Task Left Behind?: Examining the Nature of Fragmented Work. In Proceedings of the SIGCHI Conference on Human Factors in Computing Systems, CHI '05, pages 321–330. ACM Press.
- Mark, G., Gudith, D., and Klocke, U. (2008). The Cost of Interrupted Work: More Speed and Stress. In Proceedings of the SIGCHI Conference on Human Factors in Computing Systems, CHI '08, pages 107–110. ACM Press.
- McFarlane, D. and Latorella, K. (2002). The Scope and Importance of Human Interruption in Human-Computer Interaction Design. *Human-Computer Interaction*, 17(1):1–61.
 Microsoft (2019). Get started with eye control in Windows 10.
- Miller, G. A. (1956). The magical number seven, plus or minus two: some limits on our capacity for processing information. *Psychological Review*, 63(2):81–97.
- Mills, C., Gregg, J., Bixler, R., and D'Mello, S. K. (2019). Eye-Mind reader: an intelligent reading interface that promotes long-term comprehension by detecting and responding to mind wandering. *Human–Computer Interaction*, pages 1–27.

- Monk, C. A., Boehm-Davis, D. A., Mason, G., and Trafton, J. G. (2004). Recovering From Interruptions: Implications for Driver Distraction Research. *Human Factors: The Journal of the Human Factors and Ergonomics Society*, 46(4):650–663.
- Monk, C. A., Trafton, J. G., and Boehm-Davis, D. A. (2008). The Effect of Interruption Duration and Demand on Resuming Suspended Goals. *Journal of Experimental Psychology: Applied*, 14(4):299–313.
- Moore, G. C. (1989). An examination of the implementation of information technology for end users: a diffusion of innovations perspective. PhD thesis, University of British Columbia.
- Morana, S., Schacht, S., Scherp, A., and Maedche, A. (2017). A review of the nature and effects of guidance design features. *Decision Support Systems*, 97:31–42.
- Morimoto, C. H. and Mimica, M. R. (2005). Eye gaze tracking techniques for interactive applications. Computer Vision and Image Understanding, 98(1):4–24.
- Mueller, S. T. and Piper, B. J. (2014). The Psychology Experiment Building Language (PEBL) and PEBL Test Battery. *Journal of Neuroscience Methods*, 222:250–259.
- Nadj, M., Maedche, A., and Schieder, C. (2020). The effect of interactive analytical dashboard features on situation awareness and task performance. *Decision Support* Systems, 135.
- Nah, F. and Benbasat, I. (2004). Knowledge-based Support in a Group Decision Making Context: An Expert-Novice Comparison. Journal of the Association for Information Systems, 5(3):125–150.
- Nakano, Y. I., Bednarik, R., Huang, H.-H., and Jokinen, K. (2016). Introduction to the Special Issue on New Directions in Eye Gaze for Interactive Intelligent Systems. ACM Transactions on Interactive Intelligent Systems, 6(1):1–3.
- Negash, S. and Gray, P. (2008). Business Intelligence. In Handbook on Decision Support Systems 2, pages 175–193. Springer, Berlin, Heidelberg.
- Neider, M. B., Chen, X., Dickinson, C. A., Brennan, S. E., and Zelinsky, G. J. (2010). Coordinating spatial referencing using shared gaze. *Psychonomic Bulletin and Review*, 17(5):718–724.
- Networks and PWC (2016). The Battle for Attention. https://www.newsworks.org.uk/resources/the-battle-for-attention.
- Newn, J., Velloso, E., Allison, F., Abdelrahman, Y., and Vetere, F. (2017). Evaluating Real-Time Gaze Representations to Infer Intentions in Competitive Turn-Based Strategy Games. In *Proceedings of the Annual Symposium on Computer-Human Interaction in Play*, CHI PLAY '17, pages 541–552. ACM Press.
- Nguyen, C. and Liu, F. (2016). Gaze-based Notetaking for Learning from Lecture Videos. In Proceedings of the 2016 CHI Conference on Human Factors in Computing Systems, CHI '16, pages 2093–2097. ACM Press.
- Nguyen, T. V., Zhao, Q., and Yan, S. (2018). Attentive Systems: A Survey. International Journal of Computer Vision, 126(1):86–110.
- Nielsen, J. (1993). Noncommand user interfaces. Communications of the ACM, 36(4):83– 99.
- Nielsen, J. (2006). F-Shaped Pattern For Reading Web Content.

199

https://www.nngroup.com/articles/f-shaped-pattern-reading-web-content-discovered/.

- Niu, L., Lu, J., Zhang, G., and Wu, D. (2013). FACETS: A cognitive business intelligence system. *Information Systems*, 38(6):835–862.
- O'Conaill, B. and Frohlich, D. (1995). Timespace in the workplace: dealing with interruptions. In Conference Companion on Human Factors in Computing Systems (CHI '95), pages 262–263. ACM Press.
- O'Donnell, E. and David, J. S. (2000). How information systems influence user decisions: A research framework and literature review. International Journal of Accounting Information Systems, 1(3):178–203.
- Okan, Y., Galesic, M., and Garcia-Retamero, R. (2016). How People with Low and High Graph Literacy Process Health Graphs: Evidence from Eye-tracking. *Journal of Behavioral Decision Making*, 29(2-3):271–294.
- Okoe, M., Alam, S. S., and Jianu, R. (2014). A gaze-enabled graph visualization to improve graph reading tasks. *Computer Graphics Forum*, 33(3):251–260.
- Orquin, J. L. and Mueller Loose, S. (2013). Attention and choice: A review on eye movements in decision making. Acta Psychologica, 144(1):190–206.
- Otto, K., Castner, N., Geisler, D., and Kasneci, E. (2018). Development and evaluation of a gaze feedback system integrated into eyetrace. In *Proceedings of the 2018 ACM* Symposium on Eye Tracking Research & Applications, ETRA '18, pages 1–5. ACM Press.
- Ou, C. X. and Davison, R. M. (2011). Interactive or Interruptive? Instant Messaging at Work. Decision Support Systems, 52(1):61–72.
- Patterson, R. E., Blaha, L. M., Grinstein, G. G., Liggett, K. K., Kaveney, D. E., Sheldon, K. C., Havig, P. R., and Moore, J. A. (2014). A human cognition framework for information visualization. *Computers & Graphics*, 42:42–58.
- Pauwels, K., Ambler, T., Clark, B. H., LaPointe, P., Reibstein, D., Skiera, B., Wierenga, B., and Wiesel, T. (2009). Dashboards as a Service. *Journal of Service Research*, 12(2):175–189.
- Pearson, M. (2013). Insight to action: The eyes of dynamic operations. Logistics Management, 52(2):22–23.
- Peffers, K., Tuunanen, T., Rothenberger, M. A., and Chatterjee, S. (2007). A Design Science Research Methodology for Information Systems Research. *Journal of Man*agement Information Systems, 24(3):45–77.
- Perkhofer, L. and Lehner, O. (2019). Using Gaze Behavior to Measure Cognitive Load. In Davis, F. D., Riedl, R., vom Brocke, J., Léger, P.-M., and Randolph, A. B., editors, *Information Systems and Neuroscience (NeuroIS). Lecture Notes in Information Systems and Organisation*, volume 29, pages 73–83. Springer, Cham.
- Peters, M. D., Wieder, B., Sutton, S. G., and Wakefield, J. (2016). Business intelligence systems use in performance measurement capabilities: Implications for enhanced competitive advantage. *International Journal of Accounting Information Systems*, 21:1–17.
- Phillips-Wren, G., Iyer, L. S., Kulkarni, U., and Ariyachandra, T. (2015). Business ana-

200
lytics in the context of big data: A roadmap for research. Communications of the Association for Information Systems, 37:448–472.

- Pomarjanschi, L., Dorr, M., and Barth, E. (2012). Gaze guidance reduces the number of collisions with pedestrians in a driving simulator. ACM Transactions on Interactive Intelligent Systems, 1(2):1–14.
- Ponsoda, V., Scott, D., and Findlay, J. M. (1995). A probability vector and transition matrix analysis of eye movements during visual search. Acta Psychologica, 88(2):167– 185.
- Posner, M. I. (1980). Orienting of attention. *Quarterly Journal of Experimental Psychology*, 32(1):3–25.
- Power, D. J. (2007). A Brief History of Decision Support Systems. Technical report, DSSResources.COM, http://dssresources.com/history/dsshistory.html.
- Preece, J., Sharp, H., and Rogers, Y. (2015). Interaction Design: Beyond Human-Computer Interaction. John Wiley & Sons, 4 edition.
- Pries-Heje, J., Baskerville, R., and Venable, J. R. (2008). Strategies for Design Science Research Evaluation. In Proceedings of European Conference on Information Systems, volume 87.
- Proctor, R. W. and Vu, K.-p. L. (2006). The Cognitive Revolution at Age 50: Has the Promise of the Human Information-Processing Approach Been Fulfilled? International Journal of Human-Computer Interaction, 21(3):253–284.
- Purchase, H. C., Andrienko, N., Jankun-Kelly, T. J., and Ward, M. O. (2008). Theoretical Foundations of Information Visualization. In Kerren, A., Stasko, J., Fekete, J., and C., N., editors, *Information Visualization. Lecture Notes in Computer Science*, pages 46–64. Springer, Berlin, Heidelberg.
- Qvarfordt, P., Biehl, J. T., Golovchinsky, G., and Dunningan, T. (2010). Understanding the benefits of gaze enhanced visual search. In *Proceedings of the 2010 Symposium* on Eye-Tracking Research & Applications, ETRA '10, pages 283–290. ACM Press.
- Raptis, G. E., Fidas, C. A., and Avouris, N. M. (2016). Using Eye Tracking to Identify Cognitive Differences: A Brief Literature Review. In *Proceedings of the 20th Pan-Hellenic Conference on Informatics*, PCI '16, pages 1–6. ACM Press.
- Ratwani, R. M. and Trafton, J. G. (2008). Spatial memory guides task resumption. Visual Cognition, 16(8):1001–1010.
- Ratwani, R. M. and Trafton, J. G. (2010). An eye movement analysis of the effect of interruption modality on primary task resumption. *Human Factors*, 52(3):370–380.
- Rayner, K. (1998). Eye movements in reading and information processing: 20 years of research. Psychological Bulletin, 124(3):372–422.
- Richardson, D. C. and Spivey, M. J. (2004). Eye tracking: research areas and applications. In Wnek, G. E. and Bowlin, G. L., editors, *Encyclopedia of Biomaterials and Biomedical Engineering*, pages 573–582.
- Riedl, R., Fischer, T., and Léger, P.-M. (2017). A Decade of NeuroIS Research: Status Quo, Challenges, and Future Directions. In *Proceedings of the 38th International Conference on Information Systems*, pages 1–28, South Korea.
- Riedl, R. and Léger, P.-M. (2016). Fundamentals of NeuroIS: Information systems and

the brain. Studies in Neuroscience, Psychology and Behavioral Economics. Springer, Berlin, Heidelberg.

- Rivera-Pelayo, V., Fessl, A., Müller, L., and Pammer, V. (2017). Introducing Mood Self-Tracking at Work: Empirical Insights from Call Centers. ACM Transactions on Computer-Human Interaction, 24(1):1–28.
- Roda, C. (2011). Human attention and its implications for human-computer interaction. In Human Attention in Digital Environments, pages 11–62. Cambridge University Press.
- Roda, C. and Thomas, J. (2006). Attention aware systems: Theories, applications, and research agenda. *Computers in Human Behavior*, 22(4):557–587.
- Rosch, J. L. and Vogel-Walcutt, J. J. (2013). A review of eye-tracking applications as tools for training. *Cognition, Technology & Work*, 15(3):313–327.
- Rozado, D., El Shoghri, A., and Jurdak, R. (2015). Gaze dependant prefetching of web content to increase speed and comfort of web browsing. *International Journal of Human Computer Studies*, 78:31–42.
- Sallam, R. L., Tapadinhas, J., Parenteau, J., Yuen, D., and Hostmann, B. (2017). Magic Quadrant for Business Intelligence and Analytics Platforms. Technical report, Gartner Inc.
- Salvucci, D. D. (2010). On reconstruction of task context after interruption. In Proceedings of the SIGCHI Conference on Human Factors in Computing Systems, CHI '10, pages 89–92. ACM Press.
- Sarter, N. B. (2000). The Need for Multisensory Interfaces in Support of Effective Attention Allocation in Highly Dynamic Event-Driven Domains: The Case of Cockpit Automation. The International Journal of Aviation Psychology, 10(3):231–245.
- Schilit, B., Adams, N., and Want, R. (1994). Context-Aware Computing Applications. In First Workshop on Mobile Computing Systems and Applications, pages 85–90. IEEE.
- Schneider, B. and Pea, R. (2013). Real-time mutual gaze perception enhances collaborative learning and collaboration quality. *International Journal of Computer-Supported Collaborative Learning*, 8(4):375–397.
- Schwarz, C., Schwarz, A., and Black, W. C. (2014). Examining the Impact of Multicollinearity in Discovering Higher-Order Factor Models. *Communications of the* Association for Information Systems, 34:1247–1268.
- Sedig, K. and Pasob, P. (2013). Interaction Design for Complex Cognitive Activities with Visual Representations: A Pattern-Based Approach. Transaction on Human Computer-Interaction, 5:84–133.
- Sengupta, K. and Te'eni, D. (1993). Cognitive Feedback in GDSS: Improving Control and Convergence. MIS Quarterly, 17(1):87–113.
- Shao, L., Silva, N., Eggeling, E., and Schreck, T. (2017). Visual Exploration of Large Scatter Plot Matrices by Pattern Recommendation based on Eye Tracking. In Proceedings of the 2017 ACM Workshop on Exploratory Search and Interactive Data Analytics, ESIDA '17, pages 9–16. ACM Press.
- Sharafi, Z., Soh, Z., and Guéhéneuc, Y.-G. (2015). A systematic literature review on the usage of eye-tracking in software engineering. *Information and Software Technology*,

67:79-107.

- Sharma, K., Alavi, H. S., Jermann, P., and Dillenbourg, P. (2016). A gaze-based learning analytics model: In-Video Visual Feedback to Improve Learner's Attention in MOOCs. In Proceedings of the Sixth International Conference on Learning Analytics & Knowledge, LAK '16, pages 417–421. ACM Press.
- Shen, W., Hu, Y. J., and Rees Ulmer, J. (2015). Competing for Attention: An Empirical Study of Online Reviewers' Strategic Behavior. MIS Quarterly, 39(3):683–696.
- Siirtola, H., Špakov, O., Istance, H., and Räihä, K.-J. (2019). Shared Gaze in Collaborative Visual Search. International Journal of Human–Computer Interaction, 35(18):1693– 1705.
- Silva, N., Blascheck, T., Jianu, R., Rodrigues, N., Weiskopf, D., Raubal, M., and Schreck, T. (2019). Eye tracking support for visual analytics systems. In *Proceedings of the* 11th ACM Symposium on Eye Tracking Research & Applications, ETRA '19, pages 1–10. ACM Press.
- Silva, N., Schreck, T., Veas, E., Sabol, V., Eggeling, E., and Fellner, D. W. (2018). Leveraging eye-gaze and time-series features to predict user interests and build a recommendation model for visual analysis. In *Proceedings of the 2018 ACM Symposium* on Eye Tracking Research & Applications, ETRA '18, pages 1–9, New York, NY, USA. ACM Press.
- Simon, H. A. (1971). Designing organizations for an in formation-rich world. In Greenberger, M., editor, Computers, Communications and the Public Interest, pages 37–51. The Johns Hopkins Press, Baltimore.
- Simons, D. J. and Rensink, R. A. (2005). Change blindness: Past, present, and future. Trends in Cognitive Sciences, 9(1):16–20.
- Singh, D. T. (1998). Incorporating cognitive aids into decision support systems: The case of the strategy execution process. *Decision Support Systems*, 24(2):145–163.
- Smerecnik, C. M., Mesters, I., Kessels, L. T., Ruiter, R. A., De Vries, N. K., and De Vries, H. (2010). Understanding the positive effects of graphical risk information on comprehension: Measuring attention directed to written, tabular, and graphical risk information. *Risk Analysis*, 30(9):1387–1398.
- Smith, J., Legg, P., Matovic, M., and Kinsey, K. (2018). Predicting User Confidence During Visual Decision Making. ACM Transactions on Interactive Intelligent Systems, 8(2):1–30.
- Somervell, J., McCrickard, D., North, C., and Shukla, M. (2002). An evaluation of information visualization in attention-limited environments. In Ebert, D., Brunet, P., and Navazo, I., editors, *Eurographics / IEEE VGTC Symposium on Visualization*, pages 211–216. The Eurographics Association.
- Speier, C., Valacich, J. S., and Vessey, I. (1999). The Influence of Task Interruption on Individual Decision Making: An Information Overload Perspective. *Decision Sciences*, 30(2):337–360.
- Sridharan, S., McNamara, A., and Grimm, C. (2012). Subtle gaze manipulation for improved mammography training. In *Proceedings of the Symposium on Eye Tracking Research and Applications*, ETRA'12, pages 75–82. ACM Press.

- Statistics Brain (2015). Attention Span Statistics. https://www.statisticbrain.com/attention-span-statistics/.
- Steichen, B., Carenini, G., and Conati, C. (2013). User-Adaptive Information Visualization - Using Eye Gaze Data to Infer Visualization Tasks and User Cognitive Abilities. In Proceedings of the 2013 International Conference on Intelligent User Interfaces, IUI '13, pages 317–328. ACM Press.
- Steichen, B., Conati, C., and Carenini, G. (2014). Inferring visualization task properties, user performance, and user cognitive abilities from eye gaze data. ACM Transactions on Interactive Intelligent Systems, 4(2):1–29.
- Steil, J., Hagestedt, I., Huang, M. X., and Bulling, A. (2019). Privacy-aware eye tracking using differential privacy. In *Proceedings of the 11th ACM Symposium on Eye Track*ing Research & Applications, ETRA '19, pages 1–9, New York, NY, USA. ACM.
- Stothart, C., Mitchum, A., and Yehnert, C. (2015). The attentional cost of receiving a cell phone notification. Journal of Experimental Psychology: Human Perception and Performance, 41(4):893–897.
- Tam, K. Y. and Ho, S. Y. (2006). Understanding the Impact of Web Personalization on User Information Processing and Decision Outcomes. *MIS Quarterly*, 30(4):865.
- Taylor, P., Bilgrien, N., He, Z., and Siegelmann, H. T. (2015). EyeFrame: real-time memory aid improves human multitasking via domain-general eye tracking procedures. *Frontiers in ICT*, 2:1–13.
- Tegarden, D. P. (1999). Business Information Visualization. Communications of the Association for Information Systems, 1:1–37.
- Tien, T., Pucher, P. H., Sodergren, M. H., Sriskandarajah, K., Yang, G.-Z., and Darzi, A. (2014). Eye tracking for skills assessment and training: a systematic review. *Journal of Surgical Research*, 191(1):169–178.
- Toet, A. (2006). Gaze directed displays as an enabling technology for attention aware systems. Computers in Human Behavior, 22(4):615–647.
- Toker, D., Conati, C., Steichen, B., and Carenini, G. (2013). Individual user characteristics and information visualization. In *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*, CHI '13, pages 295–304. ACM Press.
- Toreini, P., Benke, I., Langner, M., Schaumann, S., Schwarzenbach, J., Bamberger, C., and Maedche, A. (2018a). Enhancing Joint Attention in Collaborative Information Dashboards with Shared Gaze Awareness. In Proceedings of the SIGHCI 2018 Proceedings-HCI/MIS Workshop 2018-The 17th Annual Pre-ICIS Workshop on HCI Research in MIS, San Francisco, California.
- Toreini, P. and Langner, M. (2019). Desigining User-Adaptive Information Dashboards: Considering Limited Attention and Working Memory. In Proceedings of the 27th European Conference on Information Systems (ECIS2019), Stockholm-Uppsala, Sweden.
- Toreini, P., Langner, M., and Maedche, A. (2018b). Designing attention-aware business intelligence and analytics dashboards to support task resumption. In Proceedings of the European Conference on Information Systems (ECIS2018), pages 11–29, Portsmouth, UK.

- Toreini, P., Langner, M., and Maedche, A. (2018c). Use of attentive information dashboards to support task resumption in working environments. In *Proceedings of the* 2018 ACM Symposium on Eye Tracking Research & Applications, ETRA '18, pages 1-3. ACM Press.
- Toreini, P., Langner, M., and Maedche, A. (2020a). Gaze-based Task Resumption Support – The Role of Working Memory Capacity (Working Paper).
- Toreini, P., Langner, M., and Maedche, A. (2020b). Using Eye-Tracking for Visual Attention Feedback. In Davis, F., Riedl, R., vom Brocke, J., Léger, P., Randolph, A., and Fischer, T., editors, *Information Systems and Neuroscience (NeuroIS Retreat 2019)*. Lecture Notes in Information Systems and Organisation, volume 32, pages 261–270. Springer, Vienna, Austria.
- Toreini, P. and Maedche, A. (2020). Eye-based Interactive Intelligent Systems: State-ofthe-Art and Future Direction (Working Paper).
- Toreini, P., Maedche, A., Langner, M., Morana, S., and Vogel, T. (2020c). Designing Attentive Information Dashboards (Working Paper).
- Toreini, P. and Morana, S. (2017). Designing Attention-aware Business Intelligence and Analytics Dashboards. In Research in Progress Proceedings of the 12th International Conference on Design Science Research in Information Systems and Technology (DESRIST), pages 64–72, Karlsruhe, Germany.
- Torres, R., Sidorova, A., and Jones, M. C. (2018). Enabling firm performance through business intelligence and analytics: A dynamic capabilities perspective. *Information & Management*, 55(7):822–839.
- Toyama, T., Sonntag, D., Orlosky, J., and Kiyokawa, K. (2015). Attention Engagement and Cognitive State Analysis for Augmented Reality Text Display Functions. In *Proceedings of the 20th International Conference on Intelligent User Interfaces*, IUI '15, pages 322–332. ACM Press.
- Trafton, J. G., Altmann, E. M., Brock, D. P., and Mintz, F. E. (2003). Preparing to resume an interrupted task: Effects of prospective goal encoding and retrospective rehearsal. *International Journal of Human Computer Studies*, 58(5):583–603.
- Treisman, A. M. and Gelade, G. (1980). A feature-integration theory of attention. Cognitive Psychology, 12(1):97–136.
- Tremblay, S., Lafond, D., Chamberland, C., Hodgetts, H. M., and Vachon, F. (2018). Gaze-Aware Cognitive Assistant for Multiscreen Surveillance. In Karwowski, W. and Ahram, T., editors, *Intelligent Human Systems Integration. IHSI 2018. Advances in Intelligent Systems and Computing*, Advances in Intelligent Systems and Computing, pages 230–236. Springer, Cham.
- Trieu, V. H. (2017). Getting value from Business Intelligence systems: A review and research agenda. Decision Support Systems, 93:111–124.
- Trösterer, S., Wuchse, M., Döttlinger, C., Meschtscherjakov, A., and Tscheligi, M. (2015). Light my way. In Proceedings of the 7th International Conference on Automotive User Interfaces and Interactive Vehicular Applications, AutomotiveUI '15, pages 196–203. ACM Press.
- Twenge, J. M., Baumeister, R. F., DeWall, C. N., Ciarocco, N. J., and Bartels, J. M.

205

(2007). Social exclusion decreases prosocial behavior. *Journal of Personality and Social Psychology*, 92(1):56–66.

- Umemoto, K., Yamamoto, T., Nakamura, S., and Tanaka, K. (2012). Search intent estimation from user's eye movements for supporting information seeking. In *Proceedings* of the International Working Conference on Advanced Visual Interfaces, AVI '12, pages 349–356. ACM Press.
- van Gog, T., Jarodzka, H., Scheiter, K., Gerjets, P., and Paas, F. (2009). Attention guidance during example study via the model's eye movements. *Computers in Human Behavior*, 25(3):785–791.
- Vandenbosch, B. and Huff, S. L. (1997). Searching and Scanning: How Executives Obtain Information from Executive Information Systems. *MIS Quarterly*, 21(1):81–107.
- Vasseur, A., Léger, P.-M., and Sénécal, S. (2019). Eye-tracking for IS Research: A Literature Review. In Proceedings of the Eighteenth Annual Pre-ICIS Workshop on HCI Research in MIS, Munich, Germany, December 15, 2019.
- Velloso, E. and Carter, M. (2016). The Emergence of EyePlay: A Survey of Eye Interaction in Games. In Proceedings of the 2016 Annual Symposium on Computer-Human Interaction in Play, CHI PLAY '16, pages 171–185.
- Venable, J. R., Pries-Heje, J., and Baskerville, R. (2012). A Comprehensive Framework for Evaluation in Design Science Research. In Peffers, K., Rothenberger, M., and Kuechler, B., editors, *Design Science Research in Information Systems. Advances in Theory and Practice. DESRIST 2012. Lecture Notes in Computer Science*, volume 7286, pages 423–438. Springer, Berlin, Heidelberg.
- Venkatesh, V., Morris, M. G., Davis, G. B., and Davis, F. D. (2003). User Acceptance of Information Technology: Toward a Unified View. *MIS Quarterly*, 27(3):425–478.
- Vertegaal, R. (2003). Attentive User Interfaces. Communications of the ACM, 46(3):30-33.
- Vila, J. and Gomez, Y. (2016). Extracting business information from graphs: An eye tracking experiment. Journal of Business Research, 69(5):1741–1746.
- vom Brocke, J., Riedl, R., and Léger, P.-M. (2013). Application Strategies for Neuroscience in Information Systems Design Science Research. *Journal of Computer Information* Systems, 53(3):1–13.
- Vrochidis, S., Patras, I., and Kompatsiaris, I. (2011). An eye-tracking-based approach to facilitate interactive video search. Proceedings of the 1st ACM International Conference on Multimedia Retrieval - ICMR '11, pages 1–8.
- Wang, R. Y. and Strong, D. M. (1996). Beyond Accuracy: What Data Quality Means to Data Consumers. Journal of Management Information Systems, 12(4):5–33.
- Ward, M. O., Grinstein, G., and Keim, D. (2010). Interactive Data Visualization: Foundations, Techniques, and Applications. CRC Press, Boca Raton.
- Ware, C. (2012). Information Visualization: Perception for Design. Morgan Kaufmann Publishers is an imprint of Elsevier, San Francisco, 2 edition.
- Watson, H. J. (2009). Tutorial: Business intelligence Past, present, and future. Communications of the Association for Information Systems, 25(1):487–510.
- Wedel, M. and Pieters, R. (2008). A Review of Eye-Tracking Research in Marketing. In Naresh, K. M., editor, *Review of marketing research*, volume 4, pages 123–147.

Emerald Group Publishing Limited, Bingley.

- Werner, N. E., Cades, D. M., Boehm-Davis, D. A., Chang, J., Khan, H., and Thi, G. (2011). What makes us resilient to interruptions? Understanding the role of individual differences in resumption. *Proceedings of the Human Factors and Ergonomics Society*, pages 296–300.
- Wetzel, S., Spiel, K., and Bertel, S. (2014). Dynamically adapting an AI game engine based on players' eye movements and strategies. In *Proceedings of the 2014 ACM* SIGCHI symposium on Engineering interactive computing systems, EICS '14, pages 3–12. ACM Press.
- Wickens, C. D., Hollands, J. G., Banbury, S., and Parasuraman, R. (2016). Engineering Psychology and Human Performance. Routledge, Taylor & Francis Group, 4 edition.
- Wilhelm, O., Hildebrandt, A., and Oberauer, K. (2013). What is working memory capacity, and how can we measure it? Frontiers in Psychology, 4:1–22.
- Wixom, B. H., Watson, H. J., and Werner, T. (2011). Developing an enterprise business intelligence capability: The Norfolk Southern journey. MIS Quarterly Executive, 10(2):61–71.
- Wolfe, J. M. and Horowitz, T. S. (2004). What attributes guide the deployment of visual attention and how do they do it? *Nature Reviews Neuroscience*, 5(6):495–501.
- Xu, S., Jiang, H., and Lau, F. C. (2008a). Personalized online document, image and video recommendation via commodity eye-tracking. In *Proceedings of the 2008 ACM* conference on Recommender systems, RecSys '08, pages 83–90. ACM Press.
- Xu, S., Jiang, H., and Lau, F. C. (2008b). User-oriented document summarization through vision-based eye-tracking. Proceedingsc of the 13th international conference on Intelligent user interfaces - IUI '09, page 7.
- Yarbus, A. L. (1967). Eye Movements and Vision. Plenum Press, New York.
- Yi, J. S., Kang, Y. A., Stasko, J., and Jacko, J. (2007). Toward a deeper understanding of the role of interaction in information visualization. *IEEE transactions on* visualization and computer graphics, 13(6):1224–31.
- Yigitbasioglu, O. M. and Velcu, O. (2012). A review of dashboards in performance management: Implications for design and research. International Journal of Accounting Information Systems, 13(1):41–59.
- Zhai, S. (2003). What's in the eyes for attentive input. Communications of the ACM, 46(3):34-39.
- Zhang, Y., Pfeuffer, K., Chong, M. K., Alexander, J., Bulling, A., and Gellersen, H. (2017). Look together: using gaze for assisting co-located collaborative search. *Personal and Ubiquitous Computing*, 21(1):173–186.
- Zhicheng Liu and Stasko, J. T. (2010). Mental Models, Visual Reasoning and Interaction in Information Visualization: A Top-down Perspective. *IEEE Transactions on Visualization and Computer Graphics*, 16(6):999–1008.
- Zugal, S. and Pinggera, J. (2014). Low-Cost Eye-Trackers: Useful for Information Systems Research? In Iliadis, L., Papazoglou, M., and Pohl, K., editors, Advanced Information Systems Engineering Workshops. CAiSE 2014. Lecture Notes in Business Information Processing, volume 178, pages 159–170. Springer, Cham.

Züger, M. and Fritz, T. (2015). Interruptibility of Software Developers and its Prediction Using Psycho-Physiological Sensors. In Proceedings of the 33rd Annual ACM Conference on Human Factors in Computing Systems, CHI '15, pages 2981–2990. ACM Press.

List of Publications

Publications Related to this Thesis

- Toreini, P., and Morana, S. (2017). Designing Attention-aware Business Intelligence and Analytics Dashboards. In Research in Progress Proceedings of the 12th International Conference on Design Science Research in Information Systems and Technology (DESRIST) (pp. 64–72). Karlsruhe, Germany.
- Toreini, P., Langner, M., and Maedche, A. (2018a). Designing attention-aware business intelligence and analytics dashboards to support task resumption. In *Proceedings* of the European Conference on Information Systems (ECIS2018) (pp. 11–29). Portsmouth, United Kingdom.
- Toreini, P., Langner, M., and Maedche, A. (2018b). Use of attentive information dashboards to support task resumption in working environments. In *Proceedings* of the 2018 ACM Symposium on Eye Tracking Research & Applications (ETRA '18) (pp. 1–3). Warsaw, Poland: ACM Press.
- Toreini, P., Benke, I., Langner, M., Schaumann, S., Schwarzenbach, J., Bamberger, C., and Maedche, A. (2018c). Enhancing Joint Attention in Collaborative Information Dashboards with Shared Gaze Awareness. In *Proceedings of the SIGHCI 2018 Proceedings-HCI/MIS Workshop 2018-The 17th Annual Pre-ICIS Workshop on HCI Research in MIS.* San Francisco, California.
- Hummel, D., Toreini, P., and Maedche, A. (2018d). Improving Digital Nudging Using Attentive User Interfaces: Theory Development and Experiment Design Using Eye-tracking. In Research in Progress Proceedings of the 13th International Conference on Design Science Research in Information Systems and Technology (DESRIST) (pp. 1–8). Chennai, India.
- Toreini, P., and Langner, M. (2019). Desigining User-Adaptive Information Dashboards: Considering Limited Attention and Working Memory. In Proceedings of the 27th European Conference on Information Systems (ECIS2019). Stockholm-Uppsala, Sweden.
- Toreini, P., Langner, M., and Maedche, A. (2020). Using Eye-Tracking for Visual Attention Feedback. In Davis F., Riedl R., vom Brocke J., Léger PM., Randolph A., Fischer T. (eds) Information Systems and Neuroscience (NeuroIS Retreat 2019). Lecture Notes in Information Systems and Organisation (Vol. 32, pp. 261– 270). Vienna, Austria: Springer.
- Langner, M., Toreini, P., and Maedche, A. (in press). AttentionBoard: A Quantified-Self Dashboard for Enhancing Attention Management with Eye-Tracking. In Davis F., Riedl R., vom Brocke J., Léger PM., Randolph A., Fischer T. (eds) Information Systems and Neuroscience (NeuroIS Retreat 2020). Virtual Conference.

Publications Under Review or Working Papers

- Toreini, P., Maedche, A., Langner, M., Morana, S., and Vogel, T. (2020a). Designing Attentive Information Dashboards. Under review in the Journal of the Association for Information Systems (JAIS).
- Toreini, P., and Maedche, A. (2020). Eye-based Interactive Intelligent Systems: Stateof-the-Art and Future Direction. *Working Paper*.
- Toreini, P., Langner, M., and Maedche, A. (2020c). Gaze-based Task Resumption Support - The Role of Working Memory Capacity. *Working Paper*.

Eidesstattliche Versicherung

gemäß § 6 Abs. 1 Ziff. 4 der Promotionsordnung des Karlsruher Instituts für Technologie für die Fakultät für Wirtschaftswissenschaften

1. Bei der eingereichten Dissertation zu dem Thema "Designing Attentive Information Dashboards with Eye Tracking Technology" handelt es sich um meine eigenständig erbrachte Leistung.

2. Ich habe nur die angegebenen Quellen und Hilfsmittel benutzt und mich keiner unzulässigen Hilfe Dritter bedient. Insbesondere habe ich wörtlich oder sinngemäß aus anderen Werken übernommene Inhalte als solche kenntlich gemacht.

3. Die Arbeit oder Teile davon habe ich bislang nicht an einer Hochschule des In- oder Auslands als Bestandteil einer Prüfungs- oder Qualifikationsleistung vorgelegt.

4. Die Richtigkeit der vorstehenden Erklärungen bestätige ich.

5. Die Bedeutung der eidesstattlichen Versicherung und die strafrechtlichen Folgen einer unrichtigen oder unvollständigen eidesstattlichen Versicherung sind mir bekannt. Ich versichere an Eides statt, dass ich nach bestem Wissen die reine Wahrheit erklärt und nichts verschwiegen habe.

Karlsruhe, den

Peyman Toreini